BID DOCUMENTS



COUNTY OF GLENN

NORTH COUNTY ADMIN ROOF UPGRADE

821 E South St, Orland, CA 95963



BIDS DUE DECEMBER 27, 2023 NO LATER THAN 10:00 A.M. AT 453 E. COUNTY ROAD 49 ½ WILLOWS, CA 95988

Owner: Glenn County Board of Supervisors

Owner's Representative:

Scott H. De Moss, County Administrative Officer

COUNTY OF GLENN

525 W. SYCAMORE ST., STE. B1

WILLOWS, CA 95988

Mandatory Pre-Bid Conference is scheduled for Thursday, December 14, 2023 beginning at 10:00 a.m. at 821 E. South St. Orland, CA 95963

North County Admin Roof Upgrade Issue for Bid – **Dec 2023**

SECTION 00 TABLE OF CONTENTS

	TABLE OF CONTENTS	
00 01 30	Project Directory	Page 1
00 10 0	Invitation to Bid	Page 2
00 10 0	Project Site Location	Page 5
00 21 0	Instructions to Bidders	Page 6
00 41 0	Bid Form	Page 14
00 52 0	Contractor Agreement	Page 21
00 60 0	Project Forms/Bonds	Page 25
00 72 0	General Conditions of the Contract for Construction	Page 28
00 73 0	Supplementary Conditions for Trade Contracts	Page 89
Attachr	nent:	
1.	Staging Area	Page 94
2. Photos		Page 95

SECTION 00 01 30 PROJECT DIRECTORY

Owner:

Glenn County Board of Supervisors 525 W. Sycamore Street, Suite B1

Willows, CA 95988

Contact: Scott H. De Moss, County Administrative Officer

Email: gcboard@countyofglenn.net

Phone: (530) 934-6400

Owner Representative:

Glenn County Administrative Office 525 W. Sycamore Street, Suite B1

Willows, CA 95988

Contact: Scott H. De Moss, County Administrative Officer

Email: gcboard@countyofglenn.net

Phone: (530) 934-6400

Construction Manager:

Glenn County General Services 453 E. County Road 49 ½ Willows, CA 95988

Contact: Samantha Grundy, Accounting and General Services Contact: Ricardo Valdez, Facilities Operations Manager

Email: facilities2@countyofglenn.net

Phone: (530) 934-6545

INVITATION TO BID – NOTICE TO BIDDERS Glenn County

North County Admin Roof Upgrade

BID SUBMISSION. Glenn County will receive sealed Bids no later than 10:00 a.m. on December 27th, 2023 at 453 E. County Road 49 ½ Willows, CA 95988. All copies of the Bid, the bid security, if any, and any other documents required to be submitted with the Bid shall be enclosed in a sealed opaque envelope. The envelope shall be addressed to the party receiving the Bids and shall be identified with the Project name, the Bidder's name, the Bidder's Department of Industrial Relations number, address and, if applicable, the designated portion of the Work for which the Bid is submitted. If the Bid is sent by mail, the sealed envelope shall be enclosed in a separate mailing envelope with the notation "SEALED BID ENCLOSED" on the face thereof. Bids shall be properly and completely executed on bid forms included in the Specifications.

DESCRIPTION OF WORK. The project location is 821 E. South St. Orland, CA 95963. Glenn County intends to award a construction contract which includes the furnishing of all labor, materials, equipment, transportation and services necessary for the completion of the project. A summary of the work is included below:

- 1. Work of the Contract can be summarized as follows: Provide all labor, equipment and materials to remove and install a new roof system over a properly prepared substrate at 821 E. South St. Orland, CA 95963. Contractor will tear off existing roof, remove heating, ventilation, and air conditioning (HVAC) units and the structures they are currently installed on, locate and repair/replace all areas of damaged substrate, install coverboard, 2-ply modified bitumen roofing system, installation of HPR Aqua Shield underlayment, installation of Composition Shingles, use Title 24 compliant roof coating and re-install HVAC units on current structures. Specific details included in the bid package shall further define the description of work.
- 2. The alternate scope of work can be summarized as follows: Provide all labor, equipment and materials to remove and install a new roof system over a properly prepared substrate at 821 E. South St. Orland, CA 95963. Contractor will tear off existing roof, remove heating, ventilation, and air conditioning (HVAC) units and the structures they are currently installed on, locate and repair/replace all areas of damaged substrate, install coverboard, Duro-Last PVC thermoplastic membrane with fleece, use Title 24 compliant membrane and re-install HVAC units on current structures. Specific details included in the bid package shall further define the description of work.

PUBLIC OPENING. Bids received prior the stated deadline shall be opened and publicly read on December 27th, 2023 10:00 a.m. at 453 E. County Road 49 ½ Willows, CA 95988. All interested citizens are invited to attend and should any citizens require special provisions, such as handicapped modifications or non-English translation personnel, the County of Glenn will provide such provisions as long as the request is made by December 18th, 2023 at 5:00 p.m.

AWARD. Award will be made to the lowest, responsive, responsible bidder. The low, responsive, responsible bidder must not be debarred, suspended, or otherwise be excluded from or ineligible for participation in federally assisted programs under Executive Order 12549. The Owner may make such investigations as deemed necessary to determine the ability of the Bidder to perform the work and the Bidder shall furnish to the Owner all such information and data for this purpose

as the Owner may request. The Owner reserves the right to reject any bid if the evidence submitted by, or investigation of, such Bidder fails to satisfy the Owner that such Bidder is properly qualified to carry out the obligations of the Agreement and to complete the work contemplated therein.

The Owner reserves the right to reject any bid, or all bids, or to accept any bid or bids, or to

make such combination of bids as may seem desirable, and to waive any and all informalities in bidding. Any bid may be withdrawn prior to the above scheduled time for the opening of bids or authorized postponement thereof.

BID DOCUMENTS. Contract Documents may be attained at the Glenn County General Services Department, 453 E. County Road 49 ½ Willows, CA 95988, between the hours of 8:00 a.m. – 5:00 p.m., Monday through Friday for a fee. The project documents are also available free of charge on the County's website at: https://www.countyofglenn.net/govt/bids

CONTACT INFORMATION.

Glenn County General Services 453 E. County Road 49 1/2 Willows, CA 95988

Contact: Samantha Grundy, Accounting and General Services Contact: Ricardo Valdez, Facilities Operations Manager

Email: facilities2@countyofglenn.net

Phone: (530) 934-6545

MANDATORY PRE-BID SITE VISIT. A mandatory project walk-through will be conducted on **December 14th, 2023 at 10:00 a.m.** at the North County Admin Building, Orland, CA 95963.

BONDS. Each bid shall be accompanied by a certified check or acceptable bidder's bond made payable to the Owner, in a sum of not less than ten percent (10%) of the total amount of the highest aggregate bid, which check or bond will be held by the Owner as evidence that the bidder will, if awarded the contract, enter into the same with the Owner upon notification from him to do so within ten (10) days of said notification. Approved performance and payment bonds guaranteeing faithful and proper performance of the work and materials, to be executed by an acceptable surety company, will be required of the Contractor at the time of contract execution. The bonds will be in the amount of 100% of the Contract Price and must be in full force and effect throughout the term of the Construction Contract plus a period of twelve (12) months from the date of substantial completion.

PREVAILING WAGE / LABOR COMPLIANCE. This project will be under prevailing wage requirements. Refer to Senate Bill (SB) 854 for recent changes to the laws governing how the Department of Industrial Relations (DIR) monitors compliance with prevailing wage requirements on public works projects. All contractors and subcontractors are required to following these new laws and include all cost as part of the base bid. No contractor or subcontractor may be listed on a bid proposal for public work on a public works project (submitted on or after March 1, 2015) unless registered with the Department of Industrial Relations (DIR) pursuant to Labor Code section 1725.5 [with limited exceptions from this requirement for bid purposes only under Labor Code section 1771.1(a)].

No contractor or subcontractor may be awarded a contract for public work on a public works project (awarded on or after April 1, 2015) unless registered with the Department of Industrial Relations (DIR) pursuant to Labor Code section 1725.5.

This project is subject to compliance monitoring and enforcement by the Department of Industrial Relations (DIR). The General Contractor must post job site notices prescribed by regulation. [See 8 Calif. Code Reg. §16451(d) for the notice that previously was required for projects monitored by the Compliance Monitoring Unit (CMU).] All contractors and subcontractors must furnish electronic certified payroll records directly to the Labor Commissioner (aka Division of Labor Standards Enforcement).

Per Senate Bill 854, the County of Glenn will notify the Department of Industrial Relations that we have awarded a construction contract using the online form PWC-100.

This contract requires compliance with the Davis-Bacon and Related Acts and adherence to the current U.S. Department of Labor Wage Decision. The Contractor must comply with the minimum rates for wages for laborers and mechanics as determined by the Secretary of Labor in accordance with the provisions of the Davis-Bacon and Related Acts. Attention is called to the fact that not less than the minimum salaries and wages set forth in the Contract Documents must be paid on this project. This is a Public Works Project subject to the higher rate of Davis Bacon wages and the prevailing wages as established by the California Department of Industrial Relations. Bidders are notified that the higher of either the Davis-Bacon or the State prevailing wage rate shall apply.

STATE AND FEDERAL REQUIREMENTS: Bidders on this work shall be required to comply with the provisions of the President's Executive Order No. 11246, as amended. The Bidders shall also comply with the requirements of 41 CFR Part 60 - 4 entitled Construction Contractors - Affirmative Action Requirements. A copy of 41 CFR Part 60 - 4 may be found in the Supplemental General Conditions of the Contract Documents and Specifications. The Bidders attention is also called to the "Minority/Women Business Participation" requirements contained in the Project Specifications. The California Department of Housing and Community Development encourages grantees to contract with MBE/WBE businesses when possible. The Contractor must meet guidelines and practices established by the Department of Housing and Community Development and appropriate federal regulations including: 1) Executive Order 11246, 2) Section 3 of the Housing and Community Development Act of 1968, as amended, 3) Certification of Non-Segregated Facilities, 4) OMB Circular A-102, 5) Title VI of the Civil Rights Act of 1964, 6) Section 504, Rehabilitation Act of 1973, 7) Age Discrimination Act of

1975, 8) Executive Order 12138, 9) Conflict of Interest Clause, 10) Retention and Custodial Requirements for Records Clause, 11) Contractors and Subcontractors Certifications, and others that may be appropriate or necessary. Contract procurement is subject to the federal regulations contained in 2 CFR 200. Any contract(s) awarded under this Advertisement for Bids are expected to be funded in part by a grant from the Department of Housing and Urban Development, as administered by the California Department of Housing and Community Development. Neither the United States nor any of its departments, agencies or employees is or will be a party to this Advertisement for Bids or any resulting contract.

PROJECT SITE LOCATION MAP

North County Admin Roof Upgrade 821 E South St, Orland, CA 95963





ASESSOR PARCEL NUMBER: 046-240-001-000

__

SECTION 00 21 00 BID PACKAGE: PRIME CONTRACTOR

Owner:

Glenn County Board of Supervisors 525 W. Sycamore Street, Suite B1 Willows, CA 95988

Contact: Scott H. De Moss, County Administrative Officer

Email: gcboard@countyofglenn.net

Phone: (530) 934-6400

Owner Representative:

Glenn County Administrative Office 525 W. Sycamore Street, Suite B1

Willows, CA 95988

Contact: Scott H. De Moss, County Administrative Officer

Email: gcboard@countyofglenn.net

Phone: (530) 934-6400

Construction Manager:

Glenn County General Services 453 E. County Road 49 1/2 Willows, CA 95988

Contact: Samantha Grundy, Accounting and General Services Contact: Ricardo Valdez, Facilities Operations Manager

Email: facilities2@countyofglenn.net

Phone: (530) 934-6545

TABLE OF ARTICLES

- 1 DEFINITIONS
- 2 BIDDER'S REPRESENTATIONS
- 3 BIDDING DOCUMENTS
- 4 BIDDING PROCEDURES
- 5 CONSIDERATION OF BIDS
- 6 POST-BID INFORMATION
- 7 PERFORMANCE BOND AND PAYMENT BOND
- 8 FORM OF AGREEMENT BETWEEN OWNER AND CONTRACTOR

ARTICLE 1 DEFINITIONS

- § 1.1 Bidding Documents include the Bidding Requirements and the proposed Contract Documents, The Bidding Requirements consist of the Advertisement or Invitation to Bid, Instructions to Bidders, Supplementary Instructions to Bidders, the bid form, and other sample bidding and contract forms. The proposed Contract Documents consist of the form of Agreement between the Owner and Contractor, Conditions of the Contract (General, Supplementary and other Conditions), Drawings, Specifications and all Addenda issued prior to execution of the Contract.
- § 1.2 Definitions set forth in the Contract Documents are applicable to the Bidding Documents.
- § 1.3 Addenda are written or graphic instruments issued by the Construction Manager prior to the execution of the Contract which modify or interpret the Bidding Documents by additions, deletions, clarifications or corrections.
- § 1.4 A Bid is a complete and properly executed proposal to do the Work for the sums stipulated therein, submitted in accordance with the Bidding Documents.
- § 1.5 The Base Bid is the sum stated in the Bid for which the Bidder offers to perform the Work described in the Bidding Documents as the base, to which Work may be added or from which Work may be deleted for sums stated in Alternate Bids.
- § 1.6 An Alternate Bid (or Alternate) is an amount stated in the Bid to be added to or deducted from the amount of the Base Bid if the corresponding change in the Work, as described in the Bidding Documents, is accepted.
- § 1.7 A Unit Price is an amount stated in the Bid as a price per unit of measurement for materials, equipment or services or a portion of the Work as described in the Bidding Documents.
- § 1.8 A Bidder is a person or entity who submits a Bid and who meets the requirements set forth in the Bidding Documents.
- **§ 1.9** A Sub-bidder is a person or entity who submits a bid to a Bidder for materials, equipment or labor for a portion of the Work.

ARTICLE 2 BIDDER'S REPRESENTATIONS

- **§ 2.1** The Bidder by making a Bid represents that:
- § 2.1.1 The Bidder has read and understands the Bidding Documents or Contract Documents, to the extent that such documentation relates to the Work for which the Bid is submitted, and for other portions of the Project, if any, being bid concurrently or presently under construction.
- § 2.1.2 The Bid is made in compliance with the Bidding Documents.
- § 2.1.3 The Bidder has visited the site, become familiar with local conditions under which the Work is to be performed and has correlated the Bidder's personal observations with the requirements of the proposed Contract Documents.

§ 2.1.4 The Bid is based upon the materials, equipment and systems required by the Bidding Documents without exception.

ARTICLE 3 BIDDING DOCUMENTS

§ 3.1 COPIES

- § 3.1.1 Bidders may obtain complete sets of the Bidding Documents from the issuing office designated in the Advertisement or Invitation to Bid in the number and for the deposit sum, if any, stated therein.
- § 3.1.2 Bidding Documents will not be issued directly to Sub-bidders unless specifically offered in the Advertisement or Invitation to Bid, or in supplementary instructions to bidders.
- § 3.1.3 Bidders shall use complete sets of Bidding Documents in preparing Bids; neither the Owner, Construction Manager nor Engineer assumes responsibility for errors or misinterpretations resulting from the use of incomplete sets of Bidding Documents.
- § 3.1.4 The Owner and Construction Manager may make copies of the Bidding Documents available on the above terms for the purpose of obtaining Bids on the Work. No license or grant of use is conferred by issuance of copies of the Bidding Documents.

§ 3.2 INTERPRETATION OR CORRECTION OF BIDDING DOCUMENTS

- § 3.2.1 The Bidder shall carefully study and compare the Bidding Documents with each other, and with other work being bid concurrently or presently under construction to the extent that it relates to the Work for which the Bid is submitted, shall examine the site and local conditions, and shall at once report to the Construction Manager errors, inconsistencies or ambiguities discovered.
- § 3.2.2 Bidders and Sub-bidders requiring clarification or interpretation of the Bidding Documents shall make a written request which shall reach the Construction Manager by the date and time indicated in the Supplementary Instructions to bidders.
- § 3.2.3 Interpretations, corrections and changes of the Bidding Documents will be made by Addendum. Interpretations, corrections and changes of the Bidding Documents made in any other manner will not be binding, and Bidders shall not rely upon them.

§ 3.3 SUBSTITUTIONS

- § 3.3.1 The materials, products and equipment described in the Bidding Documents establish a standard of required function, dimension, appearance and quality to be met by any proposed substitution.
- § 3.3.2 No substitution will be considered unless prior to receipt of Bids a written request for approval has been received by the Construction Manager for Engineer and Owner review at least ten (10) days prior to the date for receipt of Bids. Such requests shall include the name of the material or equipment for which it is to be substituted and a complete description of the proposed substitution including drawings, performance and test data, and other information necessary for an evaluation. A statement setting forth changes in other materials, equipment or other portions of the

North County Admin Roof Upgrade Issue for Bid – **Dec 2023**

Work, including changes in the work of other contracts that incorporation of the proposed substitution would require, shall be included. The burden of proof of the merit of the proposed substitution is upon the proposer. The Engineer's and Construction Manager's decision of approval or disapproval of a proposed substitution shall be final.

- § 3.3.3 If the Engineer and Construction Manager approve a proposed substitution prior to receipt of Bids, such approval will be set forth in an Addendum. Bidders shall not rely upon approvals made in any other manner.
- § 3.3.4 No substitutions will be considered after the Contract award unless specifically provided for in the Contract Documents.

§ 3.4 ADDENDA

- § 3.4.1 Addenda will be transmitted to all who are known by the issuing office to have received a complete set of Bidding Documents.
- § 3.4.2 Copies of Addenda will be made available for inspection wherever Bidding Documents are on file for that purpose.
- § 3.4.3 Addenda will be issued no later than four (4) days prior to the date for receipt of Bids except an Addendum withdrawing the request for Bids or one which includes postponement of the date for receipt of Bids.
- § 3.4.4 Each Bidder shall ascertain prior to submitting a Bid that the Bidder has received all Addenda issued, and the Bidder shall acknowledge their receipt in the Bid.

ARTICLE 4 BIDDING PROCEDURES

§ 4.1 PREPARATION OF BIDS

- § 4.1.1 Bids shall be submitted on the forms included with the Bidding Documents.
- § 4.1.2 All blanks on the bid form shall be legibly executed in a non-erasable medium.
- **§ 4.1.3** Sums shall be expressed in both words and figures. In case of discrepancy, the amount written in words shall govern.
- § 4.1.4 Interlineations, alterations and erasures must be initialed by the signer of the Bid.
- § 4.1.5 All requested Alternates shall be bid. If no change in the Base Bid is required, enter "No Change."
- § 4.1.6 Where two or more Bids for designated portions of the Work have been requested, the Bidder may, without forfeiture of the bid security, state the Bidder's refusal to accept award of less than the combination of Bids stipulated by the Bidder. The Bidder shall make no additional stipulations on the bid form nor qualify the Bid in any other manner.

§ 4.1.7 Each copy of the Bid shall state the legal name of the Bidder and the nature of legal form of the Bidder. The Bidder shall provide evidence of legal authority to perform within the jurisdiction of the Work. Each copy shall be signed by the person or persons legally authorized to bind the Bidder to a contract. A Bid by a corporation shall further give the state of incorporation and have the corporate seal affixed. A Bid submitted by an agent shall have a current power of attorney attached certifying the agent's authority to bind the Bidder.

§ 4.2 BID SECURITY

- § 4.2.1 Each Bid shall be accompanied by a bid security in the form and amount required if so stipulated in the Instructions to Bidders. The Bidder pledges to enter into a Contract with the Owner on the terms stated in the Bid and will, if required, furnish bonds covering the faithful performance of the Contract and payment of all obligations arising thereunder. Should the Bidder refuse to enter into such Contract or fail to furnish such bonds if required, the amount of the bid security shall be forfeited to the Owner as liquidated damages, not as a penalty. The amount of the bid security shall not be forfeited to the Owner in the event the Owner fails to comply with Section 6.2.
- § 4.2.2 If a surety bond is required, it shall be as provided in the Bidding Documents.
- § 4.2.3 The Owner will have the right to retain the bid security of Bidders to whom an award is being considered until either (a) the Contract has been executed and bonds, if required, have been furnished, or (b) the specified time has elapsed so that Bids may be withdrawn or (c) all Bids have been rejected.

§ 4.3 SUBMISSION OF BIDS

- **§ 4.3.1** All copies of the Bid, the bid security, if any, and any other documents required to be submitted with the Bid shall be enclosed in a sealed opaque envelope. The envelope shall be addressed to the party receiving the Bids and shall be identified with the Project name, the Bidder's name, the Bidder's Department of Industrial Relations number, and address and, if applicable, the designated portion of the Work for which the Bid is submitted. If the Bid is sent by mail, the sealed envelope shall be enclosed in a separate mailing envelope with the notation "SEALED BID ENCLOSED" on the face thereof.
- § 4.3.2 Bids shall be deposited at 453 E. County Road 49 ½ Willows, CA 95988 prior to 10:00 a.m. on December 27, 2023. Bids received after the time and date for receipt of Bids may be rejected.
- **§ 4.3.3** The Bidder shall assume full responsibility for timely delivery at the location designated for receipt of Bids.
- § 4.3.4 Oral, telephonic, telegraphic, facsimile or other electronically transmitted bids will not be considered.

§ 4.4 MODIFICATION OR WITHDRAWAL OF BID

- § 4.4.1 A Bid may not be modified, withdrawn or canceled by the Bidder during the stipulated time period following the time and date designated for the receipt of Bids, and each Bidder so agrees in submitting a Bid.
- § 4.4.2 Prior to the time and date designated for receipt of Bids, a Bid submitted may be modified or withdrawn by notice to the party receiving Bids at the place designated for receipt of Bids. Such notice shall be in writing over the signature of the Bidder. Written confirmation over the signature of the Bidder shall be received, and date- and time-stamped by the receiving party on or before the date and time set for receipt of Bids. A change shall be so worded as not to reveal the amount of the original Bid.
- § 4.4.3 Withdrawn Bids may be resubmitted up to the date and time designated for the receipt of Bids provided that they are then fully in conformance with these Instructions to Bidders.
- § 4.4.4 Bid security, if required, shall be in an amount sufficient for the Bid as resubmitted.

ARTICLE 5 CONSIDERATION OF BIDS § 5.1 OPENING OF BIDS

Bids will be unsealed and opened in a Public Bid Opening, held at 453 E. County Road 49 $\frac{1}{2}$ Willows, CA 95988 at 10:01 a.m. on December 27th, 2023..

§ 5.2 REJECTION OF BIDS

The Owner shall have the right to reject any or all Bids. A Bid not accompanied by a required bid security or by other data required by the Bidding Documents, or a Bid which is in any way incomplete or irregular is subject to rejection.

§ 5.3 ACCEPTANCE OF BID (AWARD)

- § 5.3.1 It is the intent of the Owner to award a Contract to the lowest qualified Bidder provided the Bid has been submitted in accordance with the requirements of the Bidding Documents and does not exceed the funds available. The Owner shall have the right to waive informalities and irregularities in a Bid received and to accept the Bid which, in the Owner's judgment, is in the Owner's own best interests.
- § 5.3.2 The Owner shall have the right to accept Alternates in any order or combination, unless otherwise specifically provided in the Bidding Documents, and to determine the low Bidder on the basis of the sum of the Base Bid and Alternates accepted.

ARTICLE 6 POST-BID INFORMATION

§ 6.1 CONTRACTOR'S QUALIFICATION STATEMENT

Bidders to whom award of a Contract is under consideration shall have completed and submitted all required information from the Bid Documents.

§ 6.2 OWNER'S FINANCIAL CAPABILITY

The Owner shall, at the request of the Bidder to whom award of a Contract is under consideration and no later than seven days prior to the expiration of the time for withdrawal of Bids, furnish to

the Bidder reasonable evidence that financial arrangements have been made to fulfill the Owner's obligations under the Contract. Unless such reasonable evidence is furnished, the Bidder will not be required to execute the Agreement between the Owner and Contractor.

§ 6.3 SUBMITTALS

- § 6.3.1 The Bidder shall, as soon as practicable or as stipulated in the Bidding Documents, after notification of selection for the award of a Contract, furnish to the Owner through the Construction Manager in writing:
 - 1. A designation of the Work to be performed with the Bidder's own forces;
 - 2. Names of the manufacturers, products, and the suppliers of principal items or systems of materials and equipment proposed for the Work; and
 - 3. Names of persons or entities (including those who are to furnish materials or equipment fabricated to a special design) proposed for the principal portions of the Work.
- § 6.3.2 The Bidder will be required to establish to the satisfaction of the Construction Manager and Owner the reliability and responsibility of the persons or entities proposed to furnish and perform the Work described in the Bidding Documents.
- § 6.3.3 Prior to the execution of the Contract, the Construction Manager will notify the Bidder in writing if either the Owner or Construction Manager, after due investigation, has reasonable objection to a person or entity proposed by the Bidder. If the Owner or Construction Manager has reasonable objection to a proposed person or entity, the Bidder may, at the Bidder's option, (1) withdraw the Bid or (2) submit an acceptable substitute person or entity with an adjustment in the Base Bid or Alternate Bid to cover the difference in cost occasioned by such substitution. The Owner may accept the adjusted bid price or disqualify the Bidder. In the event of either withdrawal or disqualification, bid security will not be forfeited.
- § 6.3.4 Persons and entities proposed by the Bidder and to whom the Owner and Construction Manager have made no reasonable objection must be used on the Work for which they were proposed and shall not be changed except with the written consent of the Owner and Construction Manager.

ARTICLE 7 BID, PERFORMANCE BOND AND PAYMENT BOND § 7.1 BOND REQUIREMENTS

- § 7.1.1 As stipulated in Section 00 73 00, Supplemental Conditions, of the Bidding Documents, the Bidder shall furnish bonds covering the bid, faithful performance of the Contract, and payment of all obligations arising thereunder. Bonds may be secured through the Bidder's usual sources.
- § 7.1.2 As the furnishing of such bonds is stipulated in the Bidding Documents, the cost shall be included in the Bid. If the furnishing of such bonds is required after receipt of bids and before execution of the Contract, the cost of such bonds shall be included in the Bid.
- § 7.1.3 If the Owner requires that bonds be secured from other than the Bidder's usual sources, changes in cost will be adjusted as provided in the Contract Documents.

§ 7.2 TIME OF DELIVERY AND FORM OF BONDS

- § 7.2.1 The Bidder shall deliver the required bonds to the Owner not later than ten days following the date of execution of the Contract. If the Work is to be commenced prior thereto in response to a letter of intent, the Bidder shall, prior to commencement of the Work, submit evidence satisfactory to the Owner that such bonds will be furnished and delivered in accordance with this Section 7.2.1.
- § 7.2.2 Unless otherwise provided, the bonds shall be written on the Owner supplied Performance Bond and Payment Bond forms. Both bonds shall be written in the amount of the Contract Sum.
- § 7.2.3 The bonds shall be dated on or after the date of the Contract.
- § 7.2.4 The Bidder shall require the attorney-in-fact who executes the required bonds on behalf of the surety to affix thereto a certified and current copy of the power of attorney.

ARTICLE 8 FORM OF AGREEMENT BETWEEN OWNER AND CONTRACTOR § 8.1 The Agreement for the Work will be written on the sample agreement provided in the Bidding Documents.

END OF SECTION

SECTION 00 41 00

BID FORM

FOR:	Glenn County	
	North County Admin Roof Upgrade 821 E. South Street, Orland, 95963	
BID TO:	Glenn County General Services 453 E. County Road 49 ½ Willows, CA 95988 Glenn County, State of California	
BID FROM:		
Firm Name: _		Telephone: ()
Address:		
Contractor's L	icense Number:	<u></u>
License Classi	fication:	_ License Expiration Date:
Contractor's F	ederal Tax I.D. Number:	
California Dep	partment of Industrial Relations Numb	er:
SAMS Unique	e Identifier Number:	
[] [] I HEREBY S	CORPORATION: STATE OF INCO PARTNERSHIP JOINT VENTURE PRIVATE INDIVIDUAL INDIVIDUAL DOING BUSINESS U WEAR AND CERTIFY UNDER PE	RPORATION: UNDER A FIRM NAME NALTY OF PERJURY THAT THE
	TEMENTS ARE TRUE.	
	cation submitted by:	
Print Name:	Authorized Representative	
	Authorized Representative	
Title:		

1.0 BIDDER'S REPRESENTATIONS

Bidder, represents that:

- A) It has the appropriate active Contractor's license required by the State of California;
- B) It shall complete the attached Statement of Experience;
- C) It has carefully read and examined the Bidding Documents for the proposed Work on this Project;
- D) It has examined the site of the proposed Work and all Information Available to Bidders;
- E) It has become familiar with all the conditions related to the proposed Work, including the availability of labor, materials, and equipment.
- F) It shall comply with all Labor compliance and regulations in accordance with the Department of Industrial Relations.

Bidder hereby offers to furnish all labor, materials, equipment, tools, transportation, and services necessary to complete the proposed Work on this Project in accordance with the Contract Documents for the sums quoted. Bidder further agrees that it will not withdraw its Bid within Forty-Five {45} days after the Bid Deadline, and that, if it is selected as the apparent lowest responsive and responsible Bidder, that it will, within fourteen {14} days after receipt of notice of selection, sign and deliver to Glenn County the Agreement and furnish to Glenn County all items required by the Bidding Documents. If awarded the contract, Bidder agrees to schedule and execute the Work in accordance with the Construction Documents and agrees to fully complete the Work within the Contract Time.

2.0 ADDENDA

Bidder acknowledges receipt of the following addenda and has included all work in its Lump Sum Bid amount.

Addendum #		<u>Date</u>
1	_	
2	_	
3	_	
4	_	
5	-	
6	-	
	=	

3.0 BID ITEM LIST

Pursuant to your published Notice to Bidders for the above referenced project, and in accordance with the approved Plans and Specifications for that project, the following bid for said project is submitted by the firm indicated on the first page of this Bid Form. Bids can be provided for all some or none of items 1. thru 5. below. Each option below is to bid as an individual project.

				Total bid item price, dollars
Item	Description	Quantity	Unit	
1.	MOBILIZATION	1	LS	
2.	DEMOLITION	1	LS	
3.	HVAC REMOVAL	7	LS	
4.	2 PLY MODIFIED BITUMEN INSTALLATION	1	LS	
5.	COMPOSITION SHINGLE INSTALLATION	1	LS	
6.	TOTAL BID			

3.0 BID ITEM LIST-DURO-LAST ALTERNATE SCOPE OF WORK

Pursuant to your published Notice to Bidders for the above referenced project, and in accordance with the approved Plans and Specifications for that project, the following bid for said project is submitted by the firm indicated on the first page of this Bid Form. Bids can be provided for all some or none of items 1. thru 4. below. Each option below is to bid as an individual project.

				Total bid item price, dollars
Item	Description	Quantity	Unit	
1.	MOBILIZATION	1	LS	
2.	DEMOLITION	1	LS	
3.	HVAC REMOVAL	7	LS	
4.	PVC THERMOPLASTIC MEMBRANE	1	LS	
5.	TOTAL BID			

4.0 SELECTION OF APPARENT LOW BIDDER

- 1. The County shall determine the lowest responsible, responsive bidder based on the lowest bid per item.
- 2. If this proposal shall be accepted and the undersigned shall fail to enter into the contract and furnish the two bonds in the sums to be determined as aforesaid with surety satisfactory to the County of Glenn, within 10 days, not including Saturdays, Sundays, and legal holidays, after the bidder has received notice from the County that the contract has been awarded, the County of Glenn may, at its option, determine that the bidder has abandoned the contract, and thereupon this proposal and the acceptance thereof shall be null and void and the forfeiture of such security accompanying this proposal shall operate and the same shall be the property of the County of Glenn.
- 3. In the event that the product of a unit price and an estimated quantity does not equal the extended amount stated, the unit price will govern and the correct product of the unit price and the estimated quantity shall be deemed to be the amount bid.
- 4. The County, if it chooses to award, shall award the contract to the lowest responsible, responsive bidder, however, the County may at its discretion, award the base bid along with any combination of the bid alternates it chooses.

5.0 <u>BID GUARANTY</u>

Bid security must be a bidders bond, a certified check or cashiers check payable to the County of Glenn, or cash. Bids secured by personal checks or personal guarantees will be rejected.

6.0 <u>AFFIDAVIT OF NONCOLLUSION</u>

In accordance with Title 23 United States Code Section 112 and Public Contract Code 7106 the bidder swears, deposes and says that he or she, as the party making the foregoing bid, declares that the bid is not made in the interest of, or on behalf of, any undisclosed person, partnership, company, association, organization, or corporation; that the bid is genuine and not collusive or sham; that the bidder has not directly or indirectly induced or solicited any other bidder to put in a false or sham bid, and has not directly or indirectly colluded, conspired, connived, or agreed with any bidder or anyone else to put in a sham bid, or that anyone shall refrain from bidding; that the bidder has not in any manner, directly or indirectly, sought by agreement, communication, or conference with anyone to fix the bid price of the bidder or any other bidder, or to fix any overhead, profit, or cost element of the bid price, or that of any other bidder, or to secure any advantage against the public body awarding the contract of anyone interested in the proposed contract; that all statements contained in the bid are true; and, further, that the bidder has not, directly or indirectly, submitted his or her bid price or any breakdown thereof, or the contents thereof, or divulged information or data relative thereto, or paid, and will not pay, any fee to any corporation, partnership, company association, organization, bid depository, or to any member or agent thereof to effectuate a collusive or sham bid.

North County Admin Roof Upgrade Issue for Bid – **December 2023**

Note: The above Noncollusion Affidavit is part of the Proposal. Signing this Proposal on the signature portion thereof shall also constitute signature of this Noncollusion Affidavit. Bidders are cautioned that making a false certification may subject the certifier to criminal prosecution. Noncollusion Affidavit (Exhibit 12-E Attachment D, Title 23 United States Code Section 112 and Public Contract Code Section 7106)

7.0

STATEMENT OF EXPERIENCE

The bidder has been engaged in the contracting business, under the present business name for years. Experience in work of a nature similar to that covered in the proposal extends over a period of years.
The bidder, as a Contractor, has never failed to satisfactorily complete a Contract awarded to him, except as follows:
The following contracts have been satisfactorily completed in the last three (3) years for the persons firm or authority indicated and to whom reference is made:

Persons, firm or authority indicated, and to whom reference is made:

Year Type of Work Contract Amount Owner/Agency for Work was Perf

Year	Type of Work	Contract Amount	Owner/Agency for Whom Work was Performed

8.0 SUBCONTRACTOR LISTING

In accordance with the California Public Contract Code, Division 2, Part 1, Chapter 4, Section 4100, and following, the subcontractors listed on the Bid Form attachment will perform the indicated work of improvement on the project.

The list shall specify the name and the location of the place of business of each subcontractor who will perform work or labor or render service to the contractor in or about the construction of the work or improvement, or a subcontractor licensed by the state of California who, under subcontract to the contractor, specifically fabricates and installs a portion of the work or improvement according to detailed drawings contained in the plans and specifications, (b) the portion of the

North County Admin Roof Upgrade Issue for Bid – **December 2023**

work which will be done by each subcontractor. The contractor shall list only one subcontractor for each such portion as is defined by the contractor in its bid. Per 00 20 00 Instructions to Bidders.

The following are the names and locations of places of business of all subcontractors who will perform work or labor or render service to the bidder in or about the work or improvement according to detailed drawings contained in the plans and specifications, in an amount in excess of one-half of one percent (0.5%) of the total bid or in the case of bids for the construction of streets and highways, including bridges, in an amount in excess of one-half of one percent (0.5%) of the total bid or ten thousand dollars (\$10,000) whichever is greater.

Portion of Work	Subcontractor Name / Contractor's License #	Place of Business	DIR Registration#

North County Admin Roof Upgrade



COUNTY OF GLENN

Department of General Services- Facilities 453 E. County Road 49 1/2 Willows, CA 95988 Phone: 530-934-6545

Facilities2@countyofglenn.net www.countyofglenn.net

SCOTT H. DEMOSS

County Administrative Officer 525 W. Sycamore St. Suite B1 Willows, CA 95988 Phone: 530-934-6400 Fax: 530-934-6419 gcboard@countyofglenn.net

CHANGE ORDER No. XX

Project: Project Title		Date: date				
To: Contractor Name address address	Attn: Contra	ctor Contact				
Contractor: You are hereby dir do the following described work described shall be done in acco modified by this contract chang	not included in the plans rdance with the applicab	and specifications o	f this contract. All n	ew work herein		
Description of Change:						
Method of Payment:						
		Estimated	or Total) Cost: \$	XX.XX		
Contract Time Adjustment:						
Authorization to Proceed with equipment, materials, overhead change. Other costs are non-co	, profit, any and all indire					
Recommended By:	Date:	Approved by:	Date:			
Ricardo Valdez Facilities Manager		Scott H. De Moss County Administrativ	ve Officer			
Contractor's Acceptance:						
	Printed Name		Date			

USDA, Director, Office of Civil Rights, 1400 Independence Avenue, S.W., Washington, D.C. 20250-9410 or call (800) 795-3272 (voice) or (202) 720-6382 (TDD).

PUBLIC WORKS AGREEMENT

THIS AGREEMENT made and entered into this day of date, between the County of Glenn, a political subdivision of the State of California, hereinafter referred to as and County and Contractor, hereinafter referred to Contractor;

WHEREAS, the Governing Board of the County caused plans and specifications for the work hereinafter mentioned to be prepared, and therefore did approve and adopt said plans and specifications; and

WHEREAS, the Governing Board of the County did cause to be published for the time and in the manner required by law, a Notice inviting sealed bids for the performance of said work; and

WHEREAS, the Contractor, in response to such Notice, submitted to the Governing Board of said within the time specified in said Notice, and in the manner provided for therein, a sealed bid for the performance of the work specified in said plans and specifications, which said bid and proposal, and the other bids and proposals submitted in response to said Notice, the Governing Board of public opened and canvassed in the manner provided by law; and

WHEREAS, the Contractor was the lowest responsible bidder for the performance of said work, and said Governing Board of the County, as a result of the canvass of said bids, did determine and declare Contractor to be the lowest responsible bidder for said work and award to a contract therefor.

NOW, THEREFORE, in consideration of the promises herein, it is mutually agreed between the parties hereto as follows:

I. CONTRACT DOCUMENTS

The following documents are by this reference incorporated in and made a part of this Agreement: The Standard Construction Specifications adopted by the Glenn County Board of Supervisors on date; the Special Provisions; the contract drawings, all addenda; the Notice to Contractors; the Proposal; all required bonds; and all supplemental Agreements covering alterations, amendments, or extensions to the contract. The documents which describe the work to be performed are sometimes collectively referred to herein as the Plans and Specifications.

II. SCOPE OF WORK

That the Contractor will furnish all labor, materials, services, transportation, appliances, and mechanical workmanship, as provided for and set forth in Exhibit "B" said plans and specifications, or in either of them, which said plans and specifications are hereby referred to and by such reference incorporated herein and made a part of this Agreement.

All of the said work done under this Agreement shall be under the supervision of and performed to the satisfaction of County who shall have the right to reject any and all materials and supplies furnished by the Contractor which do not comply with said plans and specifications, together with the right to require the Contractor to replace any and all work furnished by the Contractor which shall not either in workmanship or material be in strict accordance with said plans and specifications.

Said work shall be completed and ready for acceptance.

IV. PAYMENT

Attached hereto as Exhibit "A" and by reference made a part hereof, is the bid and proposal of Contractor. Said bid and proposal containing, as required by the terms of said specifications, the full and complete schedule of the different items with the lump sums or unit prices as so specified. The County agrees, in consideration of the work to be performed herein and subject to the terms and conditions thereof, to pay Contractor all sums of money which may become due to Contractor in accordance with the terms of the abovementioned bid and proposal, and this Agreement, to wit: (\$XX,XXX.XX). Said sum shall be paid in accordance with Section 8 of the Standard Specifications. With respect to that portion of the above sum as is based upon the estimated quantities specified for the general scope of the work to be performed herein, actual payment will be based upon the quantities as measured upon completion. No payment made under this Agreement shall be construed to be an acceptance of defective work or improper materials.

V. PREVAILING WAGES

Pursuant to the provisions of Articles 1 and 2 of Chapter 1, Part 7, Division II, of the Labor Code of the State of California, not less than the general prevailing rate of per diem wages, and not less than the general prevailing rate of per diem wages for holidays and overtime work, for each craft, classification or type of worker needed to execute the work contemplated under this Agreement shall be paid to all workers, laborers and mechanics employed in the execution of said work by Contractor, or by any subcontractor doing or contracting to do any part of said work. The appropriate determination of the Director of the California Department of Industrial Relations is filed with, and available for inspection at, the office of the Clerk of the Governing Board.

Contractor shall post, at each job site, a copy of such prevailing rate of per diem wages as determined by the Director for the California Department of Industrial Relations.

VI. INSURANCE

The Contractor shall carry and maintain during the life of this Agreement, such public liability, property damage and contractual liability, auto, workers' compensation and builders' risk insurance as required by the specifications.

VII. WORKERS' COMPENSATION CERTIFICATION

By execution of this Agreement, the Contractor certifies as follows:

"I am aware of the provisions of Section 3700 of the Labor Code which require every employer to be insured against liability for workers' compensation or to undertake self-insurance in accordance with the provisions of that code, and I will comply with such provisions before commencing the performance of the work of this contract."

VIII. PERFORMANCE AND PAYMENT BONDS

The Contractor shall, before beginning said work, file two bonds with the County, each made payable to the County. These bonds shall be issued by a surety company authorized to do business in the State of California, and shall be maintained during the entire life of the Agreement at the expense of the Contractor. One bond shall be in the amount of one hundred percent (100%) of the Agreement and shall guarantee the faithful performance of the Agreement.

The second bond shall be the payment bond required by California Civil Code Division 3, Part 4, Title 15, Chapter 7, and shall be in the amount of one hundred percent (100%) of the Agreement. Any alterations made in the specifications which are a part of this Agreement or in any provision of this Agreement shall not operate to release any surety from liability on any bond required hereunder and the consent to make such alterations is hereby given, and any surety on said bonds hereby waives the provisions of California Civil Code Sections 2819 and 2845.

IX. INDEMNIFICATION

The Contractor shall defend, indemnify and save harmless the County (including their officers, agents, members, employees, affiliates, and representatives) as set forth in Section 6-2 of the Standard Specifications.

X. MISCELLANEOUS PROVISIONS

This Agreement shall bind and inure to the heirs, devisees, assignees, and successors in interest of Contractor and to the successors in interest of County in the same manner as if such parties had been expressly named herein.

This agreement shall be administered and interpreted under the laws of the State of California and any action brought hereunder shall be brought in the Superior Court in and for the County of Glenn.

A party's failure to insist on strict performance of this contract or to exercise any right or remedy upon breach of this contract shall not constitute a waiver of such performance, right, or remedy. No waiver is binding unless set forth in writing signed by the waiving party.

Modifications or amendments, including those changing the primary personnel committed to the County, or affecting the proposed scope of work that result in an adjustment of the compensation under this agreement, shall be in writing and executed by both parties.

This agreement, all exhibits attached hereto, all other terms or provisions incorporated herein by reference (including the Plans and Specifications), and any notice to proceed issued in accordance with the terms hereof constitute the entire agreement and understanding between the County and Contractor as to the subject matter hereof. It supersedes all prior agreements and representations whether written and oral.

This agreement reflects the contributions of both parties and accordingly the provisions of Civil Code Section 1654 shall not apply in interpreting this Agreement.

All times stated herein or in the contract documents are of the essence hereof.

As used in this instrument the singular includes the plural, and the masculine includes the feminine and the neuter.

This Agreement may create a possessory interest subject to property taxation, and Contractor may be subject to the payment of property taxes levied on such interest.

IN WITNESS WHEREOF, County and Contractor have caused this Agreement to be executed as of the day and year first above written.

COUNTY OF GLENN:	
	Date:
Name/Titile County of Glenn	
CONTRACTOR:	
Authorized Representative	Date:
7 tautorizoa i toprosontativo	
APPROVED AS TO FORM:	
	Date:
William Vanasek, County Counsel County of Glenn	

SECTION 00 60 00 BIDDER'S BOND

KNOW ALL MEN BY THESE PRESENTS, THAT WE, THE UNDERSIGNED

	as principal; and _	as
Surety, are here	by held and bound unto the COUNTY OF GLEN	N, hereinafter called the "County", in the
sum of	dollars (\$ dollars (\$) Which sum is equal to
	hereby jointly and severally bind ourselves, our he	
and assigns.	notes, joining and severally sind surserves, sai ne	, ee.e.e.e.e.e.e.e.e.e.e.e.e.e.e.e
The cor	ndition of the above obligation is such that whereas	the Principal has submitted to the County
	attached hereto and hereby made a part hereof, to	enter into a Contract in writing, for the
construction of:		
North County	Admin Roof Upgrade	
NOW, THEREI	·ORE,	
(a)	If the Bid is rejected, or in the alternative,	
(b)	If the Bid is accepted and the Principal shall sig Contract attached hereto (all completed in accordance)	
	shall in all other respects perform the agreement	
Then th	nis obligation shall be void, otherwise the same s	shall remain in force and effect; it being
	rstood and agreed that the liability of the Surety	
hereunder shall	be the amount of this obligation as herein stated.	
	rety, for value received, hereby stipulates and agree	
	e in no way impaired or affected by any extension	· · ·
•	and said Surety does hereby waive notice of any s	
	the County file an action in a court of law to enfor	
	ver any and all costs and fees associated with the and experts' fees. The parties agree that proper v	
	perior Court of California in Sacramento Count	
	der law (such as California Code of Civil Proce	
waived.		
	THEREOF, the above-bounden parties have exe	
seals this	day of, 20, the na	me and corporate seal of each corporate
	reto affixed and those presents duly signed by its governing body.	undersigned representative, pursuant to
IN PRESENCE		
Principal Signat		(Seal)
Principal Name		,
Business Addre		
Surety Signature	e	
Surety Principa	l Name	
Surety Name Business Addre		(Seal)
		
	nium on this bond is f premium charged \$	
	nust be signed and acknowledged by both the Principal and Sun	
	s, attached hereto. Surety must be authorized and licensed	

SUBMIT BOND OR OTHER CASH GUARANTEE AS PART OF YOUR BID

PROJECT FORMS 00 60 00 - 1

"admitted surety insurer.") 1/05

Attorney-In-Fact.) 1/05

			PERFO	RMA	NCE BO	ND	BOND NO:
KNOW A	LL PERSO	ONS BY THESI	E PRESEN	TS, that			
					NTY OF GLI	ENN, a m	unicipal corporation of the State of
	hereina	fter designated	as the	"Obligee"	", has, on	,	, awarded to
			, here	einafter d	esignated as	the "Princ	
the				Pro	iect hereinaft	ter design	ated as the "Contract": and
W	/HEREAS	S, said Principal	is require	d under	the terms of	said con	tract to furnish a bond for the faithful
performan							
N	OW,	THEREFORE,	WE,	the	Principal,	and	
							, hereinafter designated a
the "Surety	y," are hel	d and firmly box	and unto th	e Oblige	e, in the pena	l sum of _	, hereinafter designated a lawful money of the
United Sta	ites, for the	he payment of	which sum	well an	d truly to be	made, w	ve bind ourselves, our heirs, executors
		uccessors, jointly					
							ve bounden Principal, its heirs, executors
							, and well and truly keep and faithfully
							alterations made as therein provided, or
							ried, and in all respects according to their
							officers and agents as therein stipulated
							in in full force and virtue.
							tract, the above obligation in said amoun
							ce of the said work, during which time i
							assigns shall fail to make full, complete
							om loss or damage made evident during
							ng from or caused by defective material
							tion in the said sum shall remain in ful
							thstanding, the obligation of the Surety
		inue so long as a					
							no change, extension of time, alteration
							er or the specification accompanying the
							nive notice of any such change, extension
							o the specifications. Said Surety hereby
		ns of Sections 28					
							d, the prevailing party shall be entitled to
							ot limited to attorneys' fees and experts
							ill be the Superior Court of California in
					or jurisdicu	on under	law (such as California Code of Civi
		392 et seq.) are e			under the C	lamtmaat l	Country's ablication under this band to
							Surety's obligation under this bond to
							t is not contingent in any manner upon er agreement). Surety must commence
							the Principal's default.
							this instrument under their seals this
offixed on	_ uay or I these pre	ecento duly signe	d by ite un	, 20, ui dersione	e name anu c I raprasantati	orporate	seal of each corporate party being hereto ant to authority of its governing body.
allixed alle	i tilese pre	sents duly signe	d by its un	dersigned	a representati	ve, pursu	ant to authority of its governing body.
Principal					Surety		
By					By		
Signature	for Princip	pal		_	Signat	ure for Si	urety
						221	((77.17)
Title of Sig	•					_	ry/(SEAL)
(This bond	must be sul	omitted in sets of f	our, each be	aring origi	ınal signatures.	. The sign:	ature of the Attorney-In-Fact for the Surety

SUBMIT BOND WITHIN 10 DAYS OF AWARD OF CONTRACT

must be acknowledged by a Notary Public. These bonds must be accompanied by a current power of attorney appointing such

26OJECT FORMS 00 60 00 - 2

PA	AYMENT BOND BOND NO:
KNOW ALL PERSONS BY THESE PRESENTS WHEREAS, the Governing Board of th California, hereinafter designated as the construction of	S, that ne COUNTY OF GLENN, a political subdivision of the State of "Obligee", has on, awarded to, hereinafter designated as "Principal", a contract for the Project; and
if said Principal, or any of its subcontractors, sha	to furnish a bond in connection and with said contract, providing that all fail to pay for any materials, provisions, or other supplies used in, contracted to be done, or for any work or labor done thereon of any to the extent hereinafter set forth:
NOW, THEREFORE, WE, the Principal as Surety, are held and firmly bound unto the Ol United States, for the payment of which sum valuministrators, and successors, jointly and sever	al and
shall fail to pay any of the persons named in Sec due under the Unemployment Insurance Code wany amounts required to be deducted, withheld State of California, from the wages of employees Unemployment Insurance Code of the State of provisions of Section 3225 and following of the same in or to an amount not exceeding the amount of the same in or to an amount not exceeding the amount of the same in or to an amount not exceeding the amount of the same in or to an amount not exceeding the amount of the same in or to an amount not exceeding the amount of the same in or to an amount not exceeding the amount of the same in or to an amount not exceeding the amount of the same in or to an amount not exceeding the amount of the same in or to an amount not exceeding the amount of the same in or to an amount not exceeding the amount of the same in or to an amount not exceeding the amount of the same in or to an amount not exceeding the amount of the same in or to an amount not exceeding the amount of the same in or to an amount not exceeding the amount of the same in or to an amount not exceeding the amount of the same in or to an amount not exceeding the amount of the same in or to an amount not exceeding the amount of the same in or to an amount not exceeding the amount of the same in or to an amount not exceeding the same in or to an amount not exceeding the same in or to an amount not exceed the same in or to an amount not exceed the same in or to an amount not exceed the same in or to an amount not exceed the same in or to an amount not exceed the same in or to an amount not exceed the same in or to an amount not exceed the same in or to an amount not exceed the same in or to an amount not exceed the same in or to an amount not exceed the same in or to an amount not exceed the same in or to an amount not exceed the same in or to an amount not exceed the same in or to an amount not exceed the same in or to an amount not exceed the same in or to an amount not exceed the same in or to an amount not exceed the same in or	
	Code Sections 3247 through 3252 of the State of California and shall apanies, and corporations named in Section 3181 of said Civil Code, signs in any suit brought upon this bond.
or addition to the terms of the contract or to the the same shall, in any way, affect its obligation extension of time, alteration, or addition to the te	eby stipulates and agrees that no change, extension of time, alteration, work to be performed thereunder or the specification accompanying s on this bond, and it does hereby waive notice of any such change, rms of the contract or to the work or to the specifications. Said Surety and 2845 of the Civil Code of the State of California.
IN WITNESS WHEREOF, the above b day of, 20 hereto and these presents duly signed by its under	ounden parties have executed this instrument under their seals this, the name and corporate seal of each corporate party being affixed ersigned representative, pursuant to authority of its governing body.
Principal	Surety
BySignature for Principal	By Signature for Surety
Title of Signatory (This bond must be submitted in sets of four, each beautiful and the sets of four and the sets	Title of Signatory (SEAL) aring original signatures. The signature of the Attorney-In-Fact for the
	ese bonds must be accompanied by a current power of attorney appointing

SUBMIT BOND WITHIN 10 DAYS OF AWARD OF CONTRACT

PROJECT FORMS 00 60 00 - 3

SECTION 00 72 00 GENERAL CONDITIONS OF THE CONTRACT FOR CONSTRUCTION

Project:

Glenn County North County Admin Roof Upgrade 821 E. South St. Orland, CA 95963

Owner:

Glenn County Board of Supervisors 525 W. Sycamore Street, Suite B1 Willows, CA 95988

Contact: Scott H. De Moss, County Administrative Officer

Email: gcboard@countyofglenn.net

Phone: (530) 934-6400

Owner Representative:

Glenn County Administrative Office 525 W. Sycamore Street, Suite B1

Willows, CA 95988

Contact: Scott H. De Moss, County Administrative Officer

Email: gcboard@countyofglenn.net

Phone: (530) 934-6400

Construction Manager:

Glenn County General Services 453 E. County Road 49 ½ Willows, CA 95988

Contact: Samantha Grundy, Accounting and General Services Contact: Ricardo Valdez, Facilities Operations Manager

Email: facilities2@countyofglenn.net

Phone: (530) 934-6545

TABLE OF ARTICLES

- 1 GENERAL PROVISIONS
- 2 OWNER
- 3 CONTRACTOR
- 4 ADMINISTRATION OF THE CONTACT
- 5 SUBCONTRACTORS
- 6 CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS

North County Admin Roof Upgrade

Issue for Bid – Dec 2023

- 7 CHANGES IN THE WORK
- 8 TIME
- 9 PAYMENTS AND COMPLETION
- 10 PROTECTION OF PERSONS AND PROPERTY
- 11 INSURANCE AND BONDS
- 12 UNCOVERING AND CORRECTION OF WORK
- 13 MISCELLANEOUS PROVISIONS
- 14 TERMINATION OR SUSPENSION OF THE CONTRACT
- 15 CLAIMS AND DISPUTES
- 16 STATUTORY AND OTHER REQUIREMENTS

North County Admin Roof Upgrade Issue for Bid – **Dec 2023**

ARTICLE 1 GENERAL PROVISIONS 1.1 BASIC DEFINITIONS 1.1.1 THE CONTRACT DOCUMENTS

The Contract Documents are enumerated in the Agreement between the Owner and Contractor (hereinafter the Agreement) and consist of the Agreement, Conditions of the Contract (General, Supplementary and other Conditions), Drawings, Specifications, Addenda issued prior to execution of the Contract, other documents listed in the Agreement and Modifications issued after execution of the Contract. A Modification is (1) a written amendment to the Contract signed by both parties, (2) a Change Order, (3) a Construction Change Directive or (4) a written order for a minor change in the Work issued by the Construction Manager. Unless specifically enumerated in the Agreement, the Contract Documents do not include the advertisement or invitation to bid, Instructions to Bidders, sample forms, other information furnished by the Owner in anticipation of receiving bids or proposals, the Contractor's bid or proposal, or portions of Addenda relating to bidding requirements.

1.1.2 THE CONTRACT

The Contract Documents form the Contract for Construction. The Contract represents the entire and integrated agreement between the parties hereto and supersedes prior negotiations, representations or agreements, either written or oral. The Contract may be amended or modified only by a Modification. The Contract Documents shall not be construed to create a contractual relationship of any kind between persons or entities other than the Owner and Contractor. The Construction Manager shall, however, be entitled to performance and enforcement of obligations under the Contract intended to facilitate performance of their duties.

1.1.3 THE WORK

The term "Work" means the construction and services required by the Contract Documents, whether completed or partially completed, and includes all other labor, materials, equipment, tools and services provided or to be provided by the Contractor to fulfill the Contractor's obligations. The Work may constitute the whole or a part of the Project.

1.1.4 THE PROJECT

The Project is the total construction of which the Work performed under the Contract Documents may be the whole or a part and which may include construction by the Owner and by Separate Contractors.

1.1.5 THE DRAWINGS

The Drawings are the graphic and pictorial portions of the Contract Documents showing the design, location and dimensions of the Work, generally including plans, elevations, sections, details, schedules and diagrams.

1.1.6 THE SPECIFICATIONS

The Specifications are that portion of the Contract Documents consisting of the written requirements for materials, equipment, systems, standards and workmanship for the Work, and performance of related services.

1.1.7 INSTRUMENTS OF SERVICE

Instruments of Service are representations, in any medium of expression now known or later developed, of the tangible and intangible creative work performed by the Construction Manager and the Construction Manager's consultants under their respective professional services agreements. Instruments of Service may include, without limitation, studies, surveys, models, sketches, drawings, specifications, and other similar materials.

1.1.8 INITIAL DECISION MAKER

Owner is the Initial Decision Maker that renders initial decisions on Claims in accordance with Section 15.2.

1.2 CORRELATION AND INTENT OF THE CONTRACT DOCUMENTS

- **1.2.1** The intent of the Contract Documents is to include all items necessary for the proper execution and completion of the Work by the Contractor. The Contract Documents are complementary, and what is required by one shall be as binding as if required by all; performance by the Contractor shall be required only to the extent consistent with the Contract Documents and reasonably inferable from them as being necessary to produce the indicated results. In the case of conflict between terms of the Contract Documents, the following order of precedence shall apply:
 - 1. Modifications
 - 2. The Agreement
 - 3. The Supplementary Conditions, if any
 - 4. The General Conditions
 - 5. The Specifications
 - 6. The Drawings
- **1.2.2** With respect to the Drawings, figured dimensions shall control over scaled measurements and specific details shall control over typical or standard details.
- **1.2.3** Organization of the Specifications into divisions, sections and articles, and arrangement of Drawings shall not control the Contractor in dividing the Work among Subcontractors or in establishing the extent of Work to be performed by any trade.
- **1.2.4** Unless otherwise stated in the Contract Documents, words that have well-known technical or construction industry meanings are used in the Contract Documents in accordance with such recognized meanings.
- 1.2.5 Whenever the context so requires, the use of the singular number shall be deemed to include the plural and vice versa. Each gender shall be deemed to include any other gender, and each shall include corporation, partnership, trust, or other legal entity whenever the context so requires. The captions and headings of the various subdivisions of the Contract Documents are intended only for reference and convenience and in no way define, limit, or prescribe the scope or intent of the Contract Documents or any subdivision thereof.

1.3 CAPITALIZATION

Terms capitalized in these General Conditions include those that are (1) specifically defined or (2) the titles of numbered articles.

1.4 INTERPRETATION

In the interest of brevity the Contract Documents frequently omit modifying words such as "all" and "any" and articles such as "the" and "an," but the fact that a modifier or an article is absent from one statement and appears in another is not intended to affect the interpretation of either statement.

1.5 OWNERSHIP AND USE OF DRAWINGS, SPECIFICATIONS AND OTHER INSTRUMENT OF SERVICE; BOOKS AND RECORDS

- 1.5.1 The Construction Manager and the Construction Manager's consultants shall be deemed the authors and owners of their respective Instruments of Service, including the Drawings and Specifications, and will retain all common law, statutory and other reserved rights, including copyrights. The Contractor, Subcontractors, Sub-subcontractors, and material or equipment suppliers shall not own or claim a copyright in the Instruments of Service. Submittal or distribution to meet official regulatory requirements or for other purposes in connection with this Project is not to be construed as publication in derogation of the Construction Manager's or Construction Manager's consultants' reserved rights.
- **1.5.2** The Contractor, Subcontractors, Sub-subcontractors and material or equipment suppliers are authorized to use and reproduce the Instruments of Service provided to them solely and exclusively for execution of the Work. All copies made under this authorization shall bear the copyright notice, if any, shown on the Instruments of Service. The Contractor, Subcontractors, Sub-subcontractors, and material or equipment suppliers may not use the Instruments of Service on other projects or for additions to this Project outside the scope of the Work without the specific written consent of the Owner, Construction Manager and the Construction Manager's consultants.
- 1.5.3 The Contractor shall keep and maintain the following (the "Books and Records"):
 - .1 copies of all documents of any kind prepared, issued, or received by Contractor in connection with the Work or the Project including all Contract Documents, bulletins, requests for information, bonds, subcontracts, purchase orders, correspondence, claims, anticipated cost reports, Shop Drawings, Change Orders, Change Order logs, project budgets (and all revisions thereof), estimates of the Contract Sum, handbooks, warranties, guarantees, operating manuals, rate manuals, technical standards and specifications, instructions, permits, licenses, certificates, test reports, notices of lien, documents served in legal proceedings, and insurance documentation;
 - .2 samples received;
 - .3 construction schedules (and all revisions thereof);
 - .4 photographs and/or a video record of the Work, as required by the Contract Documents; and
 - .5 one complete set of the Drawings and Specifications marked to record all changes during the construction and specifying the applicable Change Orders.
- **1.5.4** All samples, Shop Drawings, other submittals, or other documents of any kind prepared by Contractor or any Subcontractor in connection with the work or the Project, and all rights in the foregoing (including rights of use, copyright, and trademark), shall be and remain the sole and

confidential property of Owner (whether or not Owner undertakes, terminates, or completes the Work, or this Agreement is terminated for any reason whatsoever). Submission or description of any document described in the foregoing sentence to any person or entity for purposes of, or in connection with, the Work or the Project shall not be construed as publication in derogation of Owner's rights under this Agreement.

- 1.5.5 As part of final completion, Contractor shall be responsible for collecting from its Subcontractors a complete set of their "as-built" (i.e., as actually constructed) Drawings and Specifications indicating differences and changes from the original (with copies of all Change Orders and Shop Drawings). Contractor shall prepare a coordinated set of such Subcontractor "as-builts" (with Change Orders and Shop Drawings) and deliver one complete copy to Owner and Construction Manager shall then be responsible for reviewing the same, for further coordination (if any), and for converting the same into a digital format if needed. As a part of and condition to final completion (or any earlier termination of this Agreement by either Owner or Contractor for any reason whatsoever), Contractor shall deliver to Owner a complete set of the Books and Records. Upon prior notice to Owner from time to time, Contractor shall have access for six (6) years to the Books and Records which are stored by the Owner after final completion. Subject to the requirements of this Article, Contractor shall be entitled to retain one complete set of the Books and Records for its permanent records.
- **1.5.6** Contractor agrees that Owner has not made, and shall not be deemed to have made, any representations or warranties whatsoever with respect to the Drawings, Specifications, or any other Contract Documents, whether as to design or other adequacy or sufficiency thereof, or otherwise.

1.6 TRANSMISSION OF DATA IN DIGITAL FORM

If the parties intend to transmit Instruments of Service or any other information or documentation in digital form, they shall endeavor to establish necessary protocols governing such transmissions, unless otherwise already provided in the Agreement or the Contract Documents.

ARTICLE 2 OWNER

2.1 GENERAL

- **2.1.1** The Owner is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. The Owner shall designate in writing a representative who shall have express authority to bind the Owner with respect to all matters requiring the Owner's approval or authorization. Except as otherwise provided in Section 4.2.1, the Construction Manager do not have such authority. The term "Owner" means the Owner or the Owner's authorized representative.
- **2.1.2** The Owner shall furnish to the Contractor within fifteen days after receipt of a written request, information necessary and relevant for the Contractor to evaluate, give notice of or enforce mechanic's lien rights. Such information shall include a correct statement of the record legal title to the property on which the Project is located, usually referred to as the site, and the Owner's interest therein.

2.2 INFORMATION AND SERVICES REQUIRED OF THE OWNER

- **2.2.1** Prior to commencement of the Work, the Contractor may request in writing that the Owner provide reasonable evidence that the Owner has made financial arrangements to fulfill the Owner's obligations under the Contract. Thereafter, the Contractor may only request such evidence if (1) the Owner fails to make payments to the Contractor as the Contract Documents require; (2) a change in the Work materially changes the Contract Sum; or (3) the Contractor identifies in writing a reasonable concern regarding the Owner's ability to make payment when due. The Owner shall furnish such evidence as a condition precedent to commencement or continuation of the Work or the portion of the Work affected by a material change. After the Owner furnishes the evidence, the Owner shall not materially vary such financial arrangements without prior notice to the Contractor.
- **2.2.2** The Owner shall secure and pay for necessary approvals, easements, assessments and charges required for construction, use or occupancy of permanent structures or for permanent changes in existing facilities.
- **2.2.3** The Owner shall furnish surveys describing physical characteristics, legal limitations and utility locations for the site of the Project, and a legal description of the site. While Owner makes no representations or warranties regarding the accuracy or completeness of such information, the Contractor shall be entitled to rely on the accuracy of information furnished by the Owner but shall exercise proper precautions relating to the safe performance of the Work.
- **2.2.4** The Owner shall furnish information or services required of the Owner by the Contract Documents with reasonable promptness. The Owner shall also furnish any other information or services under the Owner's control and relevant to the Contractor's performance of the Work with reasonable promptness after receiving the Contractor's written request for such information or services. Owner's failure to be timely in furnishing information may be a Compensable Delay, but not a breach of contract.
- **2.2.5** Unless otherwise provided in the Contract Documents, the Owner shall furnish to the Contractor one copy of the Contract Documents for purposes of making reproductions pursuant to Section 1.5.2.
- **2.2.6** The Owner shall copy or involve the Construction Manager on all communications with Contractor.

2.3 OWNER'S RIGHT TO STOP THE WORK

If the Contractor fails to correct Work that is not in accordance with the requirements of the Contract Documents as required by Section 12.2 or fails to carry out Work in accordance with the Contract Documents, the Owner or Construction Manager may issue a written order to the Contractor to stop the Work, or any portion thereof, until the cause for such order has been eliminated; however, the right of the Owner to stop the Work shall not give rise to a duty on the part of the Owner to exercise this right for the benefit of the Contractor or any other person or entity. Contractor shall not be entitled to any adjustment of Contract Time or Sum as a result of such order.

2.4 OWNER'S RIGHT TO CARRY OUT THE WORK

If the Contractor defaults or neglects to carry out the Work in accordance with the Contract Documents, fails to provide sufficient labor, materials, equipment, tools, and services to maintain the Master Project Schedule, fails to start any activity by its start date as directed by the Construction Manager which will be no earlier than the early start date nor later than the late start date reflected in the Master Project Schedule, or fails to complete any activity by its completion date as directed by the Construction Manager which will be no earlier than the early completion date nor later than the late completion date as reflected in the Master Project Schedule, and then fails within a 10-day period after receipt of written notice from the Owner or Construction Manager to commence and continue correction of such default or neglect with diligence and promptness, the Owner may, without prejudice to other remedies the Owner may have, correct such deficiencies including supplying additional workers to the Contractor in such quantity and for such period as deemed necessary by the Construction Manager, all at the Contractor's expense. In such case an appropriate Change Order shall be issued deducting from payments then or thereafter due the Contractor the reasonable cost of correcting such deficiencies, including Owner's expenses and compensation for the Construction Manager's additional services made necessary by such default, neglect or failure. If payments then or thereafter due the Contractor are not sufficient to cover such amounts, the Contractor shall pay the difference to the Owner.

2.5 OWNER'S RIGHT TO ACCELERATE THE WORK

- **2.5.1** The Owner may direct the acceleration of the Work by the Contractor to meet schedule requirements when the Work has been delayed by a Permitted or Compensable Delay. The Owner will compensate the Contractor for the additional costs incurred by such acceleration to the extent that such costs are directly attributable to the acceleration and are incurred through no fault or negligence of the Contractor.
- **2.5.2** Any acceleration directed by the Owner pursuant to the foregoing provision will be by a Change Order. The Owner will not be obligated, under any circumstances, to direct such acceleration and may elect, at its option, not to accelerate the Work of the Contractor.
- **2.5.3** The Owner may accelerate the work of one or more Separate Contractors to meet schedule requirements when the Work of Contractor does not adhere to the Master Project Schedule and said failure to adhere causes, in whole or in part, a delay in the work of such Separate Contractors and if such delay would otherwise give rise to a time extension. The Owner may reduce the Contract Sum by the amounts incurred due to such acceleration to Separate Contractors.
- **2.5.4** Owner may also require Contractor to accelerate the Work due to delays which are not Permitted or Compensable Delays see Section 3.10.11 below.

ARTICLE 3 CONTRACTOR 3.1 GENERAL

3.1.1 The Contractor is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. The Contractor shall be lawfully licensed, if required in the jurisdiction where the Project is located. The Contractor shall

designate in writing a representative who shall have express authority to bind the Contractor with respect to all matters under this Contract. The term "Contractor" means the Contractor or the Contractor's authorized representative.

- **3.1.2** The Contractor shall perform the Work in accordance with the Contract Documents.
- **3.1.3** The Contractor shall not be relieved of obligations to perform the Work in accordance with the Contract Documents either by activities or duties of the Owner, Construction Manager in the administration of the Contract, or by tests, inspections or approvals required or performed by persons or entities other than the Contractor.
- **3.1.4** The term "Separate Contractors" or the plural term "Contractors" refers to persons or entities who perform construction under General Conditions of the Contract that are administered by the Construction Manager, and that are identical or substantially similar to these General Conditions.
- **3.1.5** The "Contractor" is also referred to as the "Prime Trade Contractor" or "General Contractor" in the Contract Documents.

3.2 REVIEW OF CONTRACT DOCUMENTS AND FIELD CONDITIONS BY CONTRACTOR

- **3.2.1** Execution of the Contract by the Contractor is a representation that the Contractor has visited the site, become generally familiar with local conditions under which the Work is to be performed and correlated personal observations with requirements of the Contract Documents.
- **3.2.2** Because the Contract Documents are complementary, the Contractor shall, before starting each portion of the Work, carefully study and compare the various Contract Documents relative to that portion of the Work, as well as the information furnished by the Owner pursuant to Section 2.2.3, shall take field measurements of any existing conditions related to that portion of the Work, and shall observe any conditions at the site affecting it. These obligations are for the purpose of facilitating coordination and construction by the Contractor and are not for the purpose of discovering errors, omissions, or inconsistencies in the Contract Documents; however, the Contractor shall promptly report to the Construction Manager any errors, inconsistencies or omissions discovered by or made known to the Contractor as a request for information in such form as the Construction Manager may require. It is recognized that the Contractor's review is made in the Contractor's capacity as a contractor and not as a licensed design professional, unless otherwise specifically provided in the Contract Documents.
- **3.2.3** Should Contractor discover any conflicts, omissions, or errors in the Contract Documents; have any questions about the interpretation or clarification of the Contract Documents; question whether Work is within the scope of the Contract Documents; or question that Work required is not sufficiently detailed or explained, then, before proceeding with the Work affected, Contractor shall notify the Construction Manager in writing and request interpretation, clarification, or furnishing of additional detailed instructions.
- **3.2.4** If Contractor performs any construction activity which it knows or should know involves an error, inconsistency, or omission referred to in this Section 3.2, without notifying and

obtaining the written consent of Construction Manager, Contractor shall be responsible for the resultant losses, including, without limitation, the costs of correcting defective Work.

- **3.2.5** The Contractor is not required to ascertain that the Contract Documents are in accordance with applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, but the Contractor shall promptly report to the Construction Manager any nonconformity discovered by or made known to the Contractor as a request for information in such form as the Construction Manager may require.
- **3.2.6** If the Contractor believes that additional cost or time is involved because of clarifications or instructions the Construction Manager issues in response to the Contractor's notices or requests for information pursuant to Sections 3.2.2, 3.2.3 or 3.2.5, the Contractor shall submit a Change Order Request as provided in Section 7.5 or shall make Claims as provided in Article 15. If the Contractor fails to perform the obligations of Sections 3.2.2, 3.2.3 or 3.2.5, the Contractor shall pay such costs and damages to the Owner as would have been avoided if the Contractor had performed such obligations. If the Contractor performs those obligations, the Contractor shall not be liable to the Owner or Construction Manager for damages resulting from errors, inconsistencies or omissions in the Contract Documents, for differences between field measurements or conditions and the Contract Documents, or for nonconformities of the Contract Documents to applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities.

3.3 SUPERVISION AND CONSTRUCTION PROCEDURES

- **3.3.1** The Contractor shall supervise, coordinate and direct the Work, using the Contractor's best skill and attention. The Contractor shall be solely responsible for, and have control over, construction means, methods, techniques, sequences and procedures and for coordinating all portions of the Work under the Contract, unless the Contract Documents give other specific instructions concerning these matters. If the Contract Documents give specific instructions concerning construction means, methods, techniques, sequences or procedures, the Contractor shall evaluate the jobsite safety thereof and, except as stated below, shall be fully and solely responsible for the jobsite safety of such means, methods, techniques, sequences or procedures. If the Contractor determines that such means, methods, techniques, sequences or procedures may not be safe, the Contractor shall give timely written notice to the Owner, Construction Manager and shall not proceed with that portion of the Work without further written instructions from the Construction Manager. If the Contractor is then instructed to proceed with the required means, methods, techniques, sequences or procedures without acceptance of changes proposed by the Contractor, the Owner shall be solely responsible for any loss or damage arising solely from those Owner-required means, methods, techniques, sequences or procedures.
- **3.3.2** The Contractor shall be responsible to the Owner for acts and omissions of the Contractor's employees, Subcontractors and their agents and employees, and other persons or entities performing portions of the Work for, or on behalf of, the Contractor or any of its Subcontractors.
- **3.3.3** The Contractor shall be responsible for inspection of all portions of Work already performed including work by others to determine that such portions are in proper condition to receive subsequent Work.

3.3.4 The Contractor shall inspect portions of the Project related to the Contractor's Work in order to determine that such portions are in proper condition to receive subsequent Work.

3.4 LABOR AND MATERIALS

- **3.4.1** Unless otherwise provided in the Contract Documents, the Contractor shall provide and pay for all labor, materials, equipment, tools, construction equipment and machinery, water, heat, utilities, transportation, and other facilities and services necessary for proper execution and completion of the Work, whether temporary or permanent and whether or not incorporated or to be incorporated in the Work.
- **3.4.2** Except in the case of minor changes in the Work authorized by the Construction Manager in accordance with Sections 3.12.8 or 7.4, the Contractor may make substitutions only with the consent of the Owner, after evaluation by the Construction Manager in accordance with a Change Order or Construction Change Directive.
- **3.4.3** The Contractor shall enforce strict discipline and good order among the Contractor's employees and other persons carrying out the Work. The Contractor shall not permit employment of unfit persons or persons not properly skilled in tasks assigned to them.

3.5 WARRANTY

The Contractor warrants to the Owner, Construction Manager that materials and equipment furnished under the Contract will be of good quality and new unless the Contract Documents require or permit otherwise. The Contractor further warrants that the Work will conform to the requirements of the Contract Documents and will be free from defects, except for those inherent in the quality of the Work the Contract Documents require or permit. Work, materials, or equipment not conforming to these requirements may be considered defective. The Contractor's warranty excludes remedy for damage or defect caused by abuse, alterations to the Work not executed by the Contractor, improper or insufficient maintenance, improper operation, or normal wear and tear and normal usage. If required by the Construction Manager, the Contractor shall furnish satisfactory evidence as to the kind and quality of materials and equipment.

3.6 TAXES

The Contractor shall pay sales, consumer, use and similar taxes for the Work provided by the Contractor that are legally enacted when bids are received or negotiations concluded, whether or not yet effective or merely scheduled to go into effect.

3.7 PERMITS, FEES, NOTICES, AND COMPLIANCE WITH LAWS

- **3.7.1** Owner shall secure and pay permits, fees, licenses, and inspections by government agencies necessary for proper execution and completion of the Work.
- **3.7.2** The Contractor shall comply with and give notices required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities applicable to performance of the Work.

- **3.7.3** If the Contractor performs Work knowing it to be contrary to applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, the Contractor shall assume appropriate responsibility for such Work and shall bear the costs attributable to correction.
- 3.7.4 Concealed or Unknown Conditions. If the Contractor encounters conditions at the site that are (1) subsurface or otherwise concealed physical conditions that differ materially from those indicated in the Contract Documents or (2) unknown physical conditions of an unusual nature, that differ materially from those ordinarily found to exist and generally recognized as inherent in construction activities of the character provided for in the Contract Documents, the Contractor shall immediately provide notice to the Owner, the Construction Manager before conditions are disturbed and in no event later than 3 days after first observance of the conditions. The Construction Manager will promptly investigate such conditions and, if the Construction Manager determine that they differ materially and cause an increase or decrease in the Contractor's cost of, or time required for, performance of any part of the Work, will request Contractor to submit a Change Order Request within 7 days for an adjustment in the Contract Sum or Contract Time, or both, per Section 7.5. If the Construction Manager determine that the conditions at the site are not materially different from those indicated in the Contract Documents and that no change in the terms of the Contract is justified, the Construction Manager shall promptly notify the Owner and Contractor in writing, stating the reasons. If either party disputes the Construction Manager's determination or recommendation that party may proceed as provided in Article 15.
- **3.7.5** If, in the course of the Work, the Contractor encounters human remains or recognizes the existence of burial markers, archaeological sites or wetlands not indicated in the Contract Documents or other documents provided to Contractor, the Contractor shall immediately suspend any operations that would affect them and shall notify the Owner, Construction Manager. Upon receipt of such notice, the Owner shall promptly take any action necessary to obtain any governmental authorization required to resume the operations. The Contractor shall continue to suspend such operations until otherwise instructed by the Owner but shall continue with all other operations that do not affect those remains or features. Any critical path delays directly resulting from such remains or features will be Compensable Delays subject to the requirements of Article 15.

3.8 ALLOWANCES

3.8.1 The Contractor shall include in the Contract Sum all allowances stated in the Contract Documents. Items covered by allowances shall be supplied for such amounts and by such persons or entities as the Owner may direct, but the Contractor shall not be required to employ persons or entities to whom the Contractor has reasonable objection.

3.9 NOT USED

3.10 CONTRACTOR'S CONSTRUCTION SCHEDULES

3.10.1 The Construction Manager has developed an overall "Preliminary Master Project Schedule" indicating major milestones and construction sequences for the Project, showing the general timing for the work of Contractor. This Preliminary Master Project Schedule is for

bidder information and guidance only, and is not intended to serve as the Master Project Schedule that will be utilized for construction. However, the construction milestones and sequences shall be the basis for the Master Project Schedule, unless the Construction Manager modifies them to improve the overall progress and completion by utilizing revised logic and revised schedule. The "Project Time" is the allowed time to perform the construction of the entire Project.

- **3.10.2** Using the schedules submitted by each of the Contractors, the Construction Manager will develop and issue the "Master Project Schedule" showing completion of the Project within the Project Time. The Construction Manager may require additional information from the Contractor during development of the Master Project Schedule.
- **3.10.3** The Construction Manager may impose upon the Contractor, in the initial Master Project Schedule, whatever scheduling requirements are deemed appropriate, consistent with the Preliminary Master Project Schedule, and the Contractor shall comply with any such requirements, at no additional cost to Owner, and reflect same in a "Contractor's Schedule."
- **3.10.4** The Contractor shall develop the Contractor's Schedule and submit updated schedule information to the Construction Manager within the time limits required by the Specifications and in form acceptable to the Construction Manager. The Construction Manager may, at any time, make reasonable adjustments, at no cost to the Owner, to the Master Project Schedule so that the Project may be completed within the Project Time, or if completion within the Project Time is impracticable, to mitigate damages to the Owner resulting from late completion of the Project.
- **3.10.5** The Master Project Schedule shall represent a practical plan to complete the work of the entire Project so that the entire Project can be fully completed within the Project Time. Likewise, the Contractor's Schedule shall represent a practical plan to fully complete the Work within the Contract Time and Master Project Schedule.
- **3.10.6** The Contractor shall prepare and keep current, to the satisfaction of the Construction Manager, a Submittal Schedule, in the form contained in the Exhibits, for each submittal, as required by the Specifications, and that are coordinated with the other activities in the Master Project Schedule.
- **3.10.7** Contractor shall plan, develop, supervise, control, and coordinate the performance of the Work so that its progress and the sequence and timing of Work activities conform to the current Master Project Schedule. Contractor shall continuously obtain from Subcontractors updated information and data about the planning for and progress of the Work and the delivery of equipment, shall coordinate, and monitor the progress of the Work and the delivery of equipment. Contractor shall act as the expeditor to avoid or mitigate potential and actual delays, interruptions, hindrances, or disruptions for its own forces and those forces of Subcontractors, regardless of tier. Contractor shall cooperate with the Construction Manager in the development of the Contractor's Schedule, the Master Project Schedule, and their updates.

Construction Manager's acceptance of, or its review comments about, the Contractor's Schedule

or scheduling data provided by Contractor shall not relieve Contractor of its sole responsibility to plan for, perform, and fully complete its Work within the Contract Time. Acceptance of, or review comments about, the Contractor's Schedule shall not imply the Owner's agreement with (1) any assumption upon which the Contractor's Schedule is based, or (2) any matter underlying or contained in the Contractor's Schedule.

Failure of the Construction Manager to discover errors or omissions in the Contractor's Schedule, or to inform Contractor that Contractor is behind schedule, or to direct or enforce procedures for complying with the Master Project Schedule shall not relieve Contractor from its sole responsibility to perform and complete the Work and shall not be a cause for an adjustment of the Contract Time or the Contract Sum.

- **3.10.8** The Work may require performance in several areas of the project simultaneously in order to fully complete the Project within the Project Time. As each area becomes available, Contractor shall begin work in those respective areas with additional crews if necessary to avoid a reduction of effort in other areas already under construction.
- **3.10.9** Subject to Owner's rights under the Contract or at law, time is of the essence in the Contractor's performance of this Contract. Contractor agrees to promptly commence work when directed by the Construction Manager.
- **3.10.10** In addition to any completion dates required under the Contract, the Contractor agrees to perform the work in accordance with the Construction Manager's Master Project Schedule, including all subsequent modifications to the Master Project Schedule by the Construction Manager. Contractor agrees to perform the work in a way that will not delay the Owner, the Construction Manager, or the progress of the Project, all at Contractor's cost and without additional cost or liability to Owner.
- **3.10.11** If, at any time during Contractor's performance of the work, the actual progress of the Contractor's Work falls behind the Master Project Schedule, then Contractor agrees to immediately take any steps necessary per the Construction Manager's sole discretion to improve progress in the Work or the Project. All these steps will be taken at Contractor's cost and without additional cost or liability to the Owner. If for any reason the Contractor's progress is not in accord with the Construction Manager's current Master Project Schedule, including remedial schedules, or any dates or intervals required elsewhere by the Contract, the Construction Manager may require Contractor to increase its labor force, its supervision force, the number of work shifts, overtime, work on weekends and holidays, the equipment on the Project, revise or modify its construction procedures and sequences and any other measures which the Construction Manager considers necessary, all without additional cost or liability to Owner. Neither notice by the Construction Manager nor the failure to issue notice that Contractor's progress is inadequate shall relieve Contractor from its obligation to achieve the quality of work and rate of progress required by the Construction Manager.

If the Owner incurs expense or loss or it appears that Owner may sustain expense or loss due to Contractor's failure to comply with the above provisions, the Owner or the Construction Manager may either deduct that amount from any progress payment or retention payable to

Contractor and/or delay payment of any sums otherwise owing to Contractor until the situation is remedied or adjusted to the Owner's or the Construction Manager's satisfaction.

3.10.12 The Construction Manager will schedule and coordinate the activities of the Contractor in accordance with the latest approved Master Project Schedule. The Contractor shall cooperate with the Construction Manager in the reasonable determinations of scheduling and performing the Contractor's work to avoid conflict, delay in or interference with the Work or other Contractors, or Separate Contractors, regardless of their float shown on the Master Project Schedule.

3.11 DOCUMENTS AND SAMPLES AT THE SITE

The Contractor shall maintain at the site for the Owner one copy of the Drawings, Specifications, Addenda, Change Orders and other Modifications, in good order and marked currently to indicate field changes and selections made during construction, and one copy of reviewed Shop Drawings, Product Data, Samples and similar required submittals. These shall be available to the Construction Manager and shall be delivered to the Construction Manager for submittal to the Owner upon completion of the Work as a record of the Work as constructed.

3.12 SHOP DRAWINGS, PRODUCT DATA AND SAMPLES

- § 3.12.1 Shop Drawings are drawings, diagrams, schedules and other data specially prepared for the Work by the Contractor or a Subcontractor, Sub-subcontractor, manufacturer, supplier or distributor to illustrate some portion of the Work.
- **3.12.2** Product Data are illustrations, standard schedules, performance charts, instructions, brochures, diagrams and other information furnished by the Contractor to illustrate materials or equipment for some portion of the Work.
- **3.12.3** Samples are physical examples that illustrate materials, equipment or workmanship and establish standards by which the Work will be judged.
- **3.12.4** Shop Drawings, Product Data, Samples and similar submittals are not Contract Documents. Their purpose is to demonstrate the way by which the Contractor proposes to conform to the information given and the design concept expressed in the Contract Documents for those portions of the Work for which the Contract Documents require submittals. Review by the Construction Manager is subject to the limitations of Section 4.2.8. Informational submittals upon which the Construction Manager is not expected to take responsive action may be so identified in the Contract Documents. Submittals that are not required by the Contract Documents may be returned by the Construction Manager without action.
- **3.12.5** The Contractor shall review for compliance with the Contract Documents, approve and submit to the Construction Manager Shop Drawings, Product Data, Samples and similar submittals required by the Contract Documents in accordance with the submittal schedule approved by the Construction Manager or, in the absence of an approved submittal schedule, with reasonable promptness and in such sequence as to cause no delay in the Work or in the activities of the Owner or of Separate Contractors.

- **3.12.6** By approving and submitting Shop Drawings, Product Data, Samples and similar submittals, the Contractor represents to the Owner, Construction Manager that the Contractor has (1) reviewed and approved them, (2) determined and verified materials, field measurements and field construction criteria related thereto, or will do so and (3) checked and coordinated the information contained within such submittals with the requirements of the Work and of the Contract Documents.
- **3.12.7** The Contractor shall perform no portion of the Work for which the Contract Documents require submittal and review of Shop Drawings, Product Data, Samples or similar submittals until the respective submittal has been reviewed and accepted by the Construction Manager.
- **3.12.8** The Work shall be in accordance with reviewed and accepted submittals except that the Contractor shall not be relieved of responsibility for deviations from requirements of the Contract Documents by the Construction Manager's review of Shop Drawings, Product Data, Samples or similar submittals unless the Contractor has specifically informed the Construction Manager in writing of such deviation at the time of submittal and (1) the Construction Manager has given written acceptance to the specific deviation as a minor change in the Work, or (2) a Change Order or Construction Change Directive has been issued authorizing the deviation. The Contractor shall not be relieved of responsibility for errors or omissions in Shop Drawings, Product Data, Samples or similar submittals by the Construction Manager's review thereof.
- **3.12.9** The Contractor shall direct specific attention, in writing or on resubmitted Shop Drawings, Product Data, Samples or similar submittals, to revisions other than those requested by the Construction Manager on previous submittals. In the absence of such written notice, the Construction Manager's review of a resubmission shall not apply to such revisions.
- **3.12.10** If Contractor discovers any conflicts, omissions, or errors in Shop Drawings or other submittals, the Contractor shall notify the Owner's Representative and receive instruction before proceeding with the affected Work.
- **3.12.11** The Contractor shall not be required to provide professional services that constitute the practice of engineering unless such services are specifically required by the Contract Documents for a portion of the Work or unless the Contractor needs to provide such services in order to carry out the Contractor's responsibilities for construction means, methods, techniques, sequences and procedures. The Contractor shall not be required to provide professional services in violation of applicable law. If professional design services or certifications by a design professional related to systems, materials or equipment are specifically required of the Contractor by the Contract Documents, the Owner, Construction Manager and the will specify all performance and design criteria that such services must satisfy. The Contractor shall cause such services or certifications to be provided by a properly licensed design professional, whose signature and seal shall appear on all drawings, calculations, specifications, certifications, Shop Drawings and other submittals prepared by such professional. Shop Drawings and other submittals related to the Work designed or certified by such professional, if prepared by others, shall bear such professional's written approval when submitted to the Construction Manager. The Owner and the Construction Manager shall be entitled to rely upon the adequacy, accuracy

and completeness of the services, certifications and approvals performed or provided by such design professionals, provided the Owner or Construction Manager have specified to the Contractor all performance and design criteria that such services must satisfy. Pursuant to this Section 3.12.11, the Construction Manager will review, accept or take other appropriate action on submittals only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Contractor shall not be responsible for the adequacy of the performance and design criteria specified in the Contract Documents.

3.13 USE OF SITE

The Contractor shall confine operations at the site to areas permitted by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities and the Contract Documents and shall not unreasonably encumber the site with materials or equipment.

3.14 CUTTING AND PATCHING

- **3.14.1** Contractor shall do all cutting, fitting, or patching of the Work required to make all parts of the Work come together properly and to allow the Work to receive or be received by work of Separate Contractors shown upon, or reasonably implied by, the Contract Documents. All areas requiring cutting, fitting and patching shall be restored to the condition existing prior to the cutting, fitting and patching, unless otherwise required by the Contract Documents.
- **3.14.2** The Contractor shall not damage or endanger a portion of the Work or fully or partially completed construction of the Owner or Separate Contractors by cutting, patching or otherwise altering such construction, or by excavation. The Contractor shall not cut or otherwise alter such construction by the Owner or a Separate Contractor except with written consent of the Owner and of such Separate Contractor; such consent shall not be unreasonably withheld. The Contractor shall not unreasonably withheld from the Owner or a Separate Contractor the Contractor's consent to cutting or otherwise altering the Work.

3.15 CLEANING UP

- **3.15.1** The Contractor shall keep the premises and surrounding area free from accumulation of waste materials or rubbish caused by operations under the Contract. At completion of the Work, the Contractor shall remove waste materials, rubbish, the Contractor's tools, construction equipment, machinery and surplus materials from and about the Project.
- **3.15.2** If the Contractor fails to clean up as provided in the Contract Documents, the Owner may do so and Owner shall be entitled to reimbursement from the Contractor.
- **3.15.3** Personnel of Contractor and its Subcontractors shall not occupy, live upon, or otherwise make use of the Project site during any time that Work is not being performed at the Project site, except as otherwise provided in the Contract Documents.

3.16 ACCESS TO WORK

The Contractor shall provide the Owner and Construction Manager, their consultants and other persons authorized by the Owner access to the Work in preparation and progress wherever

located. The Contractor shall provide safe and proper facilities for such access and for inspection.

3.17 ROYALTIES, PATENTS AND COPYRIGHTS

The Contractor shall pay all royalties and license fees. The Contractor shall defend suits or claims for infringement of copyrights and patent rights and shall hold the Owner and Construction Manager harmless from loss on account thereof, but shall not be responsible for such defense or loss when a particular design, process or product of a particular manufacturer or manufacturers is required by the Contract Documents, or where the copyright violations are contained in Drawings, Specifications or other documents prepared by the Owner or Construction Manager. However, if the Contractor has reason to believe that the required design, process or product is an infringement of a copyright or a patent, the Contractor shall be responsible for such loss unless such information is promptly furnished to the Construction Manager.

3.18 INDEMNIFICATION

3.18.1 Subject to Section 3.18.7, Contractor agrees to comply with the Indemnity Requirements listed in the Construction Agreement. Additionally, the Contractor shall indemnify and hold harmless each and every Indemnitee (as defined below) from and against (and to pay) all Loss-And-Expense which any Indemnitee may suffer, incur or pay out, or which may be asserted against any Indemnitee in whole or in part, by reason of, or in connection with, the following:

- 1. Any bodily injury, sickness, disease or death of or to any person or persons occurring in connection with (or arising out of or resulting from) the Work (including any corrective or warranty Work), whether on the Site or elsewhere, to the extent caused by or resulting from the negligence or willful misconduct of Contractor or those for whom it is responsible;
- 2. Any damage to or destruction of any property, including any utilities or any property of Owner other than the Work (which is intended to be covered by builders' risk insurance pursuant to Article 11) or any other person or entity, occurring in connection with (or arising out of, or resulting from) the Work, whether on the Site or elsewhere, to the extent caused by or resulting from the negligence or willful misconduct of Contractor or those for whom it is responsible (except for Loss-and-Expense arising as a result of damage to or destruction of property which is covered by Owner's builder's risk insurance);
- **3.** Any loss of benefits under any manufacturer's guarantee or service agreement resulting from the fault, inaccuracy, error, or omission of Contractor or any Subcontractor;
- **4.** Any materially untrue or incorrect statement or representation of Contractor in any Application for Payment, or in any other document submitted by Contractor with respect to the Work, the Project, or the Contract (or for purposes of securing the Contract);

- **5.** Any failure of Contractor to pay Subcontractors or suppliers, provided that Owner is not in default of its payment obligations under the Contract;
- **6.** Any failure of Contractor to comply with all applicable laws, rules or regulations of governmental entities; or
- 7. Any failure of Contractor to secure and maintain the insurance required by Article 11.
- **3.18.2** Contractor shall defend any legal proceedings commenced against any Indemnitee concerning any matter covered by any indemnity or obligation under this Article 3. Contractor shall give Owner copies of documents served in any such legal proceeding and, whenever requested by Owner, shall advise promptly as to the status of such legal proceeding. If Contractor fails to defend diligently any such legal proceeding, Owner shall have the right (but no obligation) to defend the same at Contractor's expense. Contractor shall not settle any such legal proceeding without Owner's prior written consent (unless the effect of such settlement shall be to release every Indemnitee against whom liability has been asserted from all liability whatsoever with respect to such legal proceeding, without cost or contribution from any Indemnitee).
- **3.18.3** Contractor shall notify Owner promptly of every legal proceeding or claim of which Contractor has actual knowledge which may be covered by any indemnity or obligation under this Article 3 and/or which may be covered by any insurance policy required under Article 11. Contractor shall also give timely notice of such legal proceedings and claims to each insurer which has issued an applicable policy.
- **3.18.4** Contractor's indemnities and obligations under this Contract shall not be limited or defined in any fashion whatsoever by the amount of insurance required under the Contract Documents or by any limitations or restrictions on the amount or type of damages, compensation or benefits payable to, by or for Contractor under workers' compensation acts, disability benefit acts or any other laws relating to employee benefits (although actual recoveries of insurance proceeds by an Indemnitee, net of reasonable fees and costs—including attorney's fees—of collection, shall be applied to reduce Contractor's obligation to such Indemnitee with respect to the subject matter of such recovery). No Indemnitee's right to indemnity under this Contract shall be diminished, waived, or discharged by the exercise of any other remedy allowed under this Contract or by law.
- **3.18.5** This Article 3 and the obligations of Contractor hereunder shall survive Substantial Completion, final completion, all payments (including final payment) to Contractor, and any termination of this Contract.
- **3.18.6** Contractor's obligations under this Article 3 shall be construed as protecting the Indemnitees to the fullest extent permitted by law, subject to Section 3.18.7.
- **3.18.7** Notwithstanding anything to the contrary in this Article 3 (but subject to Section 3.18.2), Contractor shall <u>not</u> be required to indemnify against any claims, damages, losses or expenses: (a) as to Construction Manager, to the extent that such loss and expense is the result of

Construction Manager's professional malpractice or professional negligence; or (b) as to any Indemnitee, to the extent that such Loss-and-Expense is the result of the negligence or willful misconduct of such Indemnitee. (This Section shall not excuse any insurer providing policies of insurance required by the Contract Documents from defending any such Indemnitee as required under Section 3.18.2.)

- **3.18.8** "Indemnitee" means Owner, Owner's affiliates, Construction Manager, Construction Manager's consultants, Construction Manager, and the members, managers, and the directors, shareholders, agents, officers and employees of the foregoing. "Loss-And-Expense" means loss, liability, obligation, damage, delay, penalty, judgment, cost, fee, claim, charge, tax, or expense of every kind (including related reasonable fees and costs of attorneys or otherwise).
- **3.18.9** The obligations of the Contractor under this Section 3.18 shall not extend to the liability of the Construction Manager, their consultants, and agents and employees of any of them arising out of (1) the preparation or approval of maps, drawings, opinions, reports, surveys, Change Orders, designs or specifications, or (2) the giving of or the failure to give directions or instructions by the Construction Manager, their consultants, and agents and employees of any of them provided such giving or failure to give is the primary cause of the injury or damage.

3.19 LIABILITY FOR AND REPAIR OF DAMAGED WORK

- **3.19.1** The Contractor shall be liable for any and all damages and losses to the Project (whether by fire, theft, vandalism, earthquake, flood or otherwise) prior to Owner's acceptance of the Project as fully completed except that the Contractor shall not be liable for:
 - 1. Losses covered by the builder's risk property insurance provided by the Owner pursuant to Article 11 of the General Conditions, except that the Contractor shall be liable for any deductible(s) and any amounts exceeding policy limits.
 - 2. Earthquake, tidal wave, or flood, provided that the loss was not caused in whole or in part by the negligent acts or omissions of Contractor, its officers, agents or employees (including all Subcontractors and suppliers of all tiers). As used herein, "flood" shall have the same meaning as in the builder's risk property insurance.

ARTICLE 4 ADMINISTRATION OF THE CONTRACT 4.1 CONSTRUCTION MANAGER 4.1.1 NOT USED

- **4.1.2** If the employment of the Construction Manager is terminated, the Owner shall employ a successor Construction Manager whose status under the Contract Documents shall be that of the Construction Manager.
- **4.1.3** The Owner shall retain qualified and licensed professionals to prepare the construction documents and assist with the administration of their construction.

4.2 ADMINISTRATION OF THE CONTRACT

4.2.1 The Construction Manager will provide administration of the Contract as described in the Contract Documents, and will be the Owner's representatives (1) during construction, (2) until final payment is due and (3) with the Owner's concurrence, from time to time during the

correction period described in Section 12.2. The Construction Manager will advise and consult with the Owner and will have authority to act on behalf of the Owner only to the extent provided in the Contract Documents, unless otherwise modified by written instrument in accordance with other provisions of the Contract.

4.2.2 NOT USED

4.2.3 NOT USED

4.2.4 Construction Manager will have the right to visit the Project site at such intervals as deemed appropriate by the Construction Manager. However, no actions taken during such Project site visit by Construction Manager shall relieve Contractor of its obligations as described in the Contract Documents.

4.2.5 COMMUNICATIONS FACILITATING CONTRACT ADMINISTRATION

Owner shall copy or involve Construction Manager on all communications with Contractor. the Communications by and with Subcontractors and material suppliers shall be through the Contractor. Communications by and with Separate Contractors shall be through the Construction Manager and shall be contemporaneously provided to the Owner.

4.2.6 NOT USED

- **4.2.7** The Construction Manager has authority to reject Work that does not conform to the Contract Documents, and to require additional inspection or testing, in accordance with Sections 13.5.2 and 13.5.3, whether or not such Work is fabricated, installed or completed, but will take such action only after notifying the Owner. Subject to review by the Owner, the Construction Manager will have the authority to reject Work which does not conform to the Contract Documents. Whenever the Construction Manager considers it necessary or advisable for implementation of the intent of the Contract Documents, the Construction Manager has authority to require additional inspection or testing of the Work in accordance with Sections 13.5.2 and 13.5.3, whether or not such Work is fabricated, installed or completed. The foregoing authority of the Construction Manager will be subject to the provisions of Sections 4.2.13 through 4.2.15 inclusive.
- **4.2.8** The Construction Manager will review and accept or take other appropriate action upon the Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Construction Manager's review of the Contractor's submittals shall not relieve the Contractor of the obligations under Sections 3.3, 3.5 and 3.12. The Construction Manager's review shall not constitute approval of safety precautions or, unless otherwise specifically stated by the Construction Manager, of any construction means, methods, techniques, sequences or procedures. The Construction Manager's acceptance of a specific item shall not indicate acceptance of an assembly of which the item is a component.
- **4.2.9** The Construction Manager will prepare Change Orders and Construction Change Directives, and the Construction Manager may authorize minor changes in the Work as provided

in Section 7.4. The Construction Manager will investigate and make determinations and recommendations regarding concealed and unknown conditions as provided in Section 3.7.4.

4.2.10 NOT USED

- **4.2.11** If the Owner and Construction Manager agree, the Construction Manager will provide one or more project representatives to assist in carrying out the Construction Manager's responsibilities at the site. The duties, responsibilities and limitations of authority of such project representatives shall be as set forth in an exhibit to be incorporated in the Contract Documents.
- **4.2.12** The Construction Manager will interpret matters concerning performance under, and requirements of, the Contract Documents on written request of either the Owner, Construction Manager or Contractor. The Construction Manager's response to such requests will be made in writing within any time limits agreed upon or otherwise with reasonable promptness.
- **4.2.13** Interpretations and decisions of the Construction Manager will be consistent with the intent of, and reasonably inferable from, the Contract Documents and will be in writing or in the form of drawings.

4.2.14 NOT USED

- **4.2.15** The Construction Manager will review and respond to requests for information about the Contract Documents. The Construction Manager's response to such requests will be made in writing within any time limits agreed upon or otherwise with reasonable promptness. If appropriate, the Construction Manager will prepare and issue supplemental Drawings and Specifications in response to the requests for information.
- **4.2.16** The Construction Manager will determine in general that the Work is being performed in accordance with the requirements of the Contract Documents, will keep the Owner informed of the progress of the Work, and will endeavor to guard the Owner against defects and deficiencies in the Work.
- **4.2.17** The Construction Manager will provide for coordination of the activities of Separate Contractors and of the Owner's own forces with the Work of the Contractor, who shall cooperate with them. The Contractor shall participate with Separate Contractors and the Construction Manager and Owner in reviewing their construction schedules when directed to do so. The Contractor shall make any revisions to the Contractor's Schedule deemed necessary after a joint review and mutual agreement.
- **4.2.18** The Construction Manager will schedule and coordinate the activities of the Contractors in accordance with the latest approved Master Project schedule.
- **4.2.19** The Construction Manager will review and certify all Applications for Payment by the Contractor, including final payment. The Construction Manager will assemble each of the Contractor's Applications for Payment with similar Applications from other Contractors into a Project Application and Project Certificate for Payment.

4.3 CONSTRUCTION MANAGER

4.3.1 The Construction Manager is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. The term "Construction Manager" means the Construction Manager identified on the cover page or the Construction Manager's authorized representative, or their successors when designated by written notice to Contractor.

4.3.2 NOT USED.

4.3.3 In case of termination of employment of the Construction Manager, the Owner shall appoint a new construction manager whose status under the Contract Documents shall be that of the former construction manager.

ARTICLE 5 SUBCONTRACTORS 5.1 DEFINITIONS

- **5.1.1** A Subcontractor is a person or entity who has a direct contract with the Contractor to perform a portion of the Work at the site. The term "Subcontractor" is referred to throughout the Contract Documents as if singular in number and means a Subcontractor or an authorized representative of the Subcontractor. The term "Subcontractor" does not include a Separate Contractor or subcontractors of a Separate Contractor.
- **5.1.2** A Sub-subcontractor is a person or entity who has a direct or indirect contract with a Subcontractor to perform a portion of the Work at the site. The term "Sub-subcontractor" is referred to throughout the Contract Documents as if singular in number and means a Sub-subcontractor or an authorized representative of the Sub-subcontractor. Unless otherwise noted or inapplicable in a particular context, the term Subcontractor includes Subcontractors at all tiers and the term Contractor includes all Subcontractors.

5.2 AWARD OF SUBCONTRACTS AND OTHER CONTRACTS FOR PORTIONS OF THE WORK

- **5.2.1** Unless otherwise stated in the Contract Documents or the bidding requirements, the Contractor, as soon as practicable after award of the Contract, shall furnish in writing to the Owner through the Construction Manager the names of persons or entities for review by the Owner, Construction Manager (including those who are to furnish materials or equipment fabricated to a special design) proposed for each principal portion of the Work. The Construction Manager may reply within 14 days to the Contractor in writing stating (1) whether the Owner or the Construction Manager has reasonable objection to any such proposed person or entity or (2) that the Owner or Construction Manager requires additional time for review. Failure of the Owner or Construction Manager to reply within the 14 day period shall constitute notice of no reasonable objection.
- **5.2.2** The Contractor shall not contract with a proposed person or entity to whom the Owner or Construction Manager has made reasonable and timely objection. The Contractor shall not be required to contract with anyone to whom the Contractor has made reasonable objection.

5.2.3 If the Owner or Construction Manager has reasonable objection to a person or entity proposed by the Contractor, the Contractor shall propose another to whom the Owner or Construction Manager has no reasonable objection. If the proposed but rejected Subcontractor was reasonably capable of performing the Work, the Contract Sum and Contract Time shall be increased or decreased by the difference, if any, occasioned by such change, and an appropriate Change Order shall be issued before commencement of the substitute Subcontractor's Work. However, no increase in the Contract Sum or Contract Time shall be allowed for such change unless the Contractor has acted promptly and responsively in submitting names as required.

5.2.4 The Contractor shall not substitute a Subcontractor, person or entity previously selected if the Owner or Construction Manager makes reasonable objection to such substitution.

5.3 SUBCONTRACTUAL RELATIONS

By appropriate written agreement, the Contractor shall require each Subcontractor, to the extent of the Work to be performed by the Subcontractor, to be bound to the Contractor by terms of the Contract Documents, and to assume toward the Contractor all the obligations and responsibilities, including the responsibility for safety of the Subcontractor's Work, which the Contractor, by these Documents, assumes toward the Owner and Construction Manager. Each subcontract agreement shall preserve and protect the rights of the Owner and Construction Manager under the Contract Documents with respect to the Work to be performed by the Subcontractor so that subcontracting thereof will not prejudice such rights, and shall allow to the Subcontractor, unless specifically provided otherwise in the subcontract agreement, the benefit of all rights, remedies and redress against the Contractor that the Contractor, by the Contract Documents, has against the Owner. The Contractor shall require each Subcontractor to enter into similar agreements with Sub-subcontractors. The Contractor shall make available to each proposed Subcontractor, prior to the execution of the subcontract agreement, copies of the Contract Documents to which the Subcontractor will be bound, and, upon written request of the Subcontractor, identify to the Subcontractor terms and conditions of the proposed subcontract agreement that may be at variance with the Contract Documents. Subcontractors will similarly make copies of applicable portions of such documents available to their respective proposed Sub-subcontractors.

5.4 CONTINGENT ASSIGNMENT OF SUBCONTRACTS

5.4.1 Each subcontract agreement for a portion of the Work is assigned by the Contractor to the Owner, provided that

- 1. Assignment is effective only after termination of the Contract by the Owner for cause pursuant to Section 14.2 and only for those subcontract agreements that the Owner accepts by notifying the Subcontractor and Contractor in writing; and
- 2. Assignment is subject to the prior rights of the surety, if any, obligated under bond relating to the Contract.

When the Owner accepts the assignment of a subcontract agreement, the Owner assumes the Contractor's rights and obligations under the subcontract from the date of assignment.

- **5.4.2** Upon such assignment, if the Work has been suspended for more than 30 days, the Subcontractor's compensation shall be adjusted for the documented increases in cost resulting from the suspension.
- **5.4.3** Upon such assignment to the Owner under this Section 5.4, the Owner may further assign the subcontract to a successor contractor or other entity.

ARTICLE 6 CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS 6.1

OWNER'S RIGHT TO PERFORM CONSTRUCTION AND TO AWARD SEPARATE CONTRACTS

- **6.1.1** The Owner reserves the right to perform construction or operations related to the Project with the Owner's own forces, which include persons or entities under separate contracts not administered by the Construction Manager. The Owner further reserves the right to award separate contracts in connection with other portions of the Project or other construction or operations on the site under Conditions of the Contract identical or substantially similar to these including those portions related to insurance and waiver of subrogation.
- **6.1.2** When separate contracts are awarded for different portions of the Project or other construction or operations on the site, the term "Contractor" in the Contract Documents in each case shall mean the Contractor who executes each separate Owner-Contractor Agreement.
- **6.1.3** See Section 4.2.17.
- **6.1.4** Not Used.
- **6.1.5** The Project, of which the Work is a part, will involve other contracts for work to be performed and also work to be performed by the Owner on the same site. By entering into this Contract, Contractor acknowledges that Owner has the right to enter into such other contracts and to perform work, and that the work of said contracts or the Owner may (i) be in close proximity to and/or performed contemporaneously with the work of this Contract, and (ii) result in delays in or disruptions to Contractor's Work. The Contractor further agrees as follows:
 - 1. The Owner shall cause Separate Contractors to afford the Contractor reasonable opportunity for the introduction and storage of its materials and the execution of its work. The Contractor shall properly connect and coordinate its construction and operations with the construction and operations of Separate Contractors and Owner forces, as required by the Contract Documents.
 - 2. The Contractor shall cooperate with Separate Contractors and the Owner on the project site and will do nothing to delay, hinder, disrupt, or interfere with the work of Separate Contractors, or the Owner. Contractor shall coordinate its work with the work of any Separate Contractor and agrees to attend any coordination meetings scheduled for this purpose by the Owner's Representative. Any dispute between the Contractor and any Separate Contractor over how the work of the various trades should be coordinated, shall be promptly submitted by Contractor to the Owner's Representative. The

Contractor agrees to cooperate with the development of, and to be bound by, any reasonable coordination plan directed by Owner's Representative to address the dispute, even if Contractor does not agree with the coordination plan so developed. The Contractor agrees that if its work is delayed, hindered, disrupted or interfered with by a Separate Contractor to the extent such delays, hindrances, disruptions, and interferences result in Contractor working beyond the Contract Time, through no fault of the Contractor, the Contract shall be subject to a time extension, but no compensation from the Owner, provided the Contractor complies with the requirements of the Contract for seeking a time extension, including without limitation, the requirements set forth in Articles 4, 7 and 8 of the General Conditions.

6.2 MUTUAL RESPONSIBILITY

- **6.2.1** The Contractor shall afford the Owner, its forces, Construction Manager and Separate Contractors reasonable opportunity for introduction and storage of their materials and equipment and performance of their activities, and shall connect and coordinate the Contractor's construction and operations with theirs as required by the Contract Documents.
- **6.2.2** If part of the Contractor's Work depends for proper execution or results upon construction or operations by the Owner or a Separate Contractor, the Contractor shall, prior to proceeding with that portion of the Work, promptly report to the Construction Manager apparent discrepancies or defects in such other construction that would render it unsuitable for such proper execution and results. Failure of the Contractor so to report shall constitute an acknowledgment that the Owner's or Separate Contractor's completed or partially completed construction is fit and proper to receive the Contractor's Work, except as to defects not then reasonably discoverable.
- **6.2.3** The Contractor shall reimburse the Owner for costs the Owner incurs that are payable to a Separate Contractor because of the Contractor's delays, improperly timed activities or defective construction. The Owner shall be responsible to the Contractor for costs the Contractor incurs because of a Separate Contractor's delays, improperly timed activities, damage to the Work or defective construction.
- **6.2.4** The Contractor shall promptly remedy damage the Contractor wrongfully causes to completed or partially completed construction or to property of the Owner or Separate Contractors as provided in Section 10.2.5.
- **6.2.5** Each Separate Contractor shall have the same responsibilities for cutting and patching as are described for the Contractor in Section 3.14.

6.3 OWNER'S RIGHT TO CLEAN UP

If a dispute arises among the Contractor, Separate Contractors and the Owner as to the responsibility under their respective contracts for maintaining the premises and surrounding area free from waste materials and rubbish, the Owner may clean up and allocate the cost among those responsible as the Construction Manager.

ARTICLE 7 CHANGES IN THE WORK 7.1 GENERAL

- **7.1.1** Changes in the Work may be accomplished after execution of the Contract, and without invalidating the Contract, by Change Order, Construction Change Directive or order for a minor change in the Work, subject to the limitations stated in this Article 7 and elsewhere in the Contract Documents.
- **7.1.2** A Change Order shall be based upon agreement among the Owner and Contractor; a Construction Change Directive by the Owner may or may not be agreed to by the Contractor; an order for a minor change in the Work may be issued by the Construction Manager alone.
- **7.1.3** Changes in the Work shall be performed under applicable provisions of the Contract Documents, and the Contractor shall proceed promptly, unless otherwise provided in the Change Order, Construction Change Directive or order for a minor change in the Work.
- **7.1.4** In the Contract Documents, "changes in the Work" means any Work in addition to the original Work under this Contract which adds scope or program to the original Work under this Contract. For the avoidance of doubt, "changes in the Work" does not include (a) development of details reasonably inferable from or implied by, but not explicit in, Drawings or Specifications; (b) the inclusion or exclusion of any item which should be included or excluded in accordance with good construction practice, whether or not shown or set forth in a drawing or specification; and (c) any matters resulting from, or attributable to, mistake, error, omission, delay, or negligence of Contractor or any Subcontractor.

7.2 CHANGE ORDERS

- **7.2.1** A Change Order is a written instrument signed by the Owner and Contractor stating their agreement upon all of the following:
 - 1. The change in the Work;
 - 2. The amount of the adjustment, if any, in the Contract Sum; and
 - 3. The extent of the adjustment, if any, in the Contract Time.

7.3 CONSTRUCTION CHANGE DIRECTIVES

- **7.3.1** A Construction Change Directive is a written order prepared by the Construction Manager and signed by the Owner directing a change in the Work prior to agreement on adjustment, if any, in the Contract Sum or Contract Time, or both. The Owner may by Construction Change Directive, without invalidating the Contract, order changes in the Work within the general scope of the Contract consisting of additions, deletions or other revisions, the Contract Sum and Contract Time being adjusted accordingly.
- **7.3.2** A Construction Change Directive shall be used in the absence of total agreement on the terms of a Change Order.
- **7.3.3** If the Construction Change Directive provides for an adjustment to the Contract Sum, the adjustment shall be based on one of the following methods:
 - 1. Mutual acceptance of a lump sum properly itemized and supported by sufficient substantiating data to permit evaluation;

- 2. Unit prices stated in the Contract Documents or subsequently agreed upon;
- 3. Cost to be determined in a manner agreed upon by the parties and a mutually acceptable fixed or percentage fee; or
- **4.** Costs as provided in Section 7.3.7 plus the Contractor Fee on such costs.
- **7.3.4** If unit prices are stated in the Contract Documents or subsequently agreed upon, and if quantities originally contemplated are materially changed in a proposed Change Order or Construction Change Directive so that application of such unit prices to quantities of Work proposed will cause substantial inequity to the Owner or Contractor, the applicable unit prices shall be equitably adjusted.
- **7.3.5** Upon receipt of a Construction Change Directive, the Contractor shall promptly proceed with the change in the Work involved and advise the Construction Manager of the Contractor's agreement or disagreement with the method, if any, provided in the Construction Change Directive for determining the proposed adjustment in the Contract Sum or Contract Time within 7 calendar days after receipt of the Construction Change Directive.
- **7.3.6** A Construction Change Directive signed by the Contractor indicates the Contractor's agreement therewith, including adjustment in Contract Sum and Contract Time or the method for determining them. Such agreement shall be effective immediately and shall be recorded as a Change Order.
- **7.3.7** If the Contractor does not respond promptly or disagrees with the method for adjustment in the Contract Sum, Contractor shall nonetheless proceed with the Work if so directed by Owner, and the Contractor shall keep and present, in such form as the Construction Manager may prescribe, an itemized accounting together with appropriate supporting data of Contractor's costs for the changed or disputed work. Unless otherwise provided in the Contract Documents, costs for the purposes of this Section 7.3.7 shall mean actual costs incurred or to be incurred by Contractor and each Subcontractor regardless of tier, and shall be limited to the following categories:
 - 1. Straight-time wages or salaries for employees employed at the Project site, or at fabrication sites off the Project site, in the direct performance of the applicable Work.
 - 2. Fringe benefits and payroll taxes for employees employed at the Project site, or at fabrication sites off the Project site, in the direct performance of the applicable Work.
 - 3. Overtime wages or salaries for overtime Work specifically authorized in writing by Owner for employees employed at the Project site, or at fabrication sites off the Project site, in the direct performance of the applicable Work.
 - 4. Fringe benefits and payroll taxes for overtime Work specifically authorized in writing by Owner for employees employed at the Project site, or at fabrication sites off the Project site, in the direct performance of the applicable Work.

- 5. Costs (including sales tax) of materials and consumable items which are furnished and incorporated into the applicable Work. Such costs shall be charged at the lowest price available to the Contractor but in no event shall such costs exceed competitive costs obtainable from other subcontractors, suppliers, manufacturers, and distributors in the area of the Project site. All discounts, rebates, and refunds and all returns from sale of surplus materials and consumable items shall accrue to Owner and Contractor shall make provisions so that they may be obtained.
- 6. Rental charges for necessary machinery and equipment, whether owned or hired, as authorized in writing by Construction Manager, exclusive of hand tools, used directly in the performance of the applicable Work. Such rental charges shall not exceed the current U. S. Army Corp of Engineers scheduled charges for the area in which the work is performed. Contractor shall attach a schedule of rental charges to the Cost Proposal. The charges for any machinery and equipment shall cease when the use thereof is no longer necessary for the applicable Work.
- 7. Additional costs of royalties and permits due to the performance of the applicable Work.
- **8.** Cost for additional premiums for insurance and bonds, which may not exceed in aggregate 1.5% of items .1 through .7 above.
- **7.3.8** Pending final determination of the total cost of a Construction Change Directive to the Owner, the Contractor may request payment for Work completed under the Construction Change Directive in Applications for Payment. The Construction Manager will make an interim determination for purposes of monthly certification for payment for those costs and certify for payment the amount that the Construction Manager determines, in the Construction Manager's judgment, to be reasonably justified. Construction Manager's interim determination of cost shall adjust the Contract Sum on the same basis as a Change Order, subject to the right of either party to disagree and assert a Claim in accordance with Article 15.
- **7.3.9** When the Owner and Contractor agree with a determination made by the Construction Manager concerning the adjustments in the Contract Sum and Contract Time, or otherwise reach agreement upon the adjustments, such agreement shall be effective immediately and the Construction Manager will prepare a Change Order. Change Orders may be issued for all or any part of a Construction Change Directive.
- **7.3.10** The term "Contractor Fee" shall mean the full amount of compensation, both direct and indirect (including without limitation all overhead and profit), to be paid to Contractor for its own Work and the Work of all Subcontractors pursuant to a Change Order or Construction Change Directive, for all costs and expenses not included in paragraphs 7.3.7.1, .2,. 3, and .4 above. The Contractor Fee shall not be compounded. The Contractor Fee shall be computed as follows:

- 1. Fifteen percent (15%) for the cost of that portion of the Work to be performed by the Contractor with its own forces.
- 2. Fifteen percent (15%) of the cost of that portion of the Work to be performed by a Subcontractor, regardless of tier, with its own forces, plus five percent (5%) of such cost for the Contractor. Total combined Contractor and Subcontractor's mark-up shall not exceed twenty percent (20%).
- **7.3.11** Cost of Work for purposes of this Section 7.3 shall not include any of the following:
 - 1. Superintendent(s).
 - 2. Assistant Superintendent(s.)
 - **3.** Project Engineer(s).
 - 4. Project Manager(s).
 - **5.** Scheduler(s).
 - **6.** Estimator(s).
 - 7. Drafting or Detailing
 - **8.** Small Tools (replacement value does not exceed \$300).
 - 9. Office Expenses including staff, materials and supplies.
 - **10.** On-site or off-site trailer and storage rental and expenses.
 - 11. Site fencing.
 - 12. Data processing personnel and equipment.
 - **13.** Utilities including gas, electric, sewer, water, telephone, facsimile, copier equipment.
 - **14.** Federal, state or local business income and franchise taxes.
 - **15.** Overhead and Profit.
 - **16.** Costs and expenses of any kind or item not specifically and expressly included in Article 7.3.7 above.

7.4 MINOR CHANGES IN THE WORK

The Construction Manager has authority to order minor changes in the Work not involving adjustment in the Contract Sum or extension of the Contract Time and not inconsistent with the intent of the Contract Documents. Such changes will be effected by written order signed by the Construction Manager and delivered by the Construction Manager and shall be binding on the Contractor.

7.5 CONTRACTOR CHANGE ORDER REQUEST

- **7.5.1** Contractor may request changes to the Contract Sum and/or Contract Time for those reasons specifically allowed under the Contract.
- **7.5.2** Contractor entitlement to an adjustment of the Contract Sum and/or Contract Time is conditioned upon Contractor submitting a timely Change Order Request.
- **7.5.3** A Change Order Request will be deemed timely submitted if, and only if, it is submitted within 7 days of the date the Contractor discovers, or reasonably should discover, the circumstances giving rise to the Change Order Request, unless additional time is allowed in writing by Owner for submission of the Change Order Request.

- **7.5.4** A Change Order Request must state that it is a Change Order Request, state the reason for the request, cite contractual justification for the request, and specify the amount of any requested adjustment of the Contract Sum or Contract Time. If the Contractor requests an adjustment to the Contract Sum, the Contractor shall submit a cost proposal with the Change Order Request that includes a complete and itemized breakdown of all costs allowed in Section 7.3.7 and Contractor's Fee under Section 7.3.10 that substantiates the Contractor's cost proposal. Quantities, description of work items, unit costs for each work items, tabulations, mark-ups, etc. shall be clearly indicated and described in the cost proposal.
- **7.5.5** Upon request of Construction Manager, Contractor shall submit such additional information as may be requested by Construction Manager for the purpose of evaluating the Change Order Request. Such additional information may include:
 - 1. Actual cost records for any changed or extra costs (including without limitation, payroll records, material and rental invoices and the like), shall be submitted by the deadline established by the Construction Manager, who may require such actual cost records to be submitted and reviewed, on a daily basis, by the Construction Manager.
 - 2. Daily time and material sheets which shall be signed by the Construction Manager.
 - **3.** Any other information requested by the Construction Manager or Owner for the purpose of evaluating the Change Order Request, which shall be submitted by the deadline established by the Construction Manager.

ARTICLE 8 TIME 8.1 DEFINITIONS

- **8.1.1** Unless otherwise provided, Contract Time is the period of time, including authorized adjustments, allotted in the Contract Documents for Substantial Completion of the Work.
- **8.1.2** The date of commencement of the Work is the date established in the Agreement.
- **8.1.3** The date of Substantial Completion is the date certified by the Construction Manager in accordance with Section 9.8.
- **8.1.4** The term "day" as used in the Contract Documents shall mean calendar day unless otherwise specifically defined.

8.2 PROGRESS AND COMPLETION

- **8.2.1** Time limits stated in the Contract Documents are of the essence of the Contract. By executing the Agreement the Contractor confirms that the Contract Time is a reasonable period for performing the Work.
- **8.2.2** The Contractor shall not knowingly, except by agreement or instruction of the Owner in writing, prematurely commence operations on the site or elsewhere prior to the effective date of

insurance required by Article 11 to be furnished by the Contractor and Owner. The date of commencement of the Work shall not be changed by the effective date of such insurance.

- **8.2.3** The Contractor agrees that the Owner is purchasing the right to have the Contractor fully mobilized on the site from the earliest date for commencement of the Work shown on the Preliminary Master Project Schedule to the date shown for full completion of the Project as shown in the bidding documents.
- **8.2.4** The Contractor agrees that the Owner is purchasing the right to have Contractor work on the Project site shared by the Contractor and Separate Contractors. The Contractor recognizes that as a result of working at a shared Project site there will be a loss of productivity and disruption commensurate with a project of the type, size and complexity of the Project. The Contractor agrees that the Contract Sum includes full compensation for such loss of productivity and disruption.
- **8.2.5** The Contractor shall proceed expeditiously with adequate forces and shall achieve Substantial Completion within the Contract Time. If Construction Manager determines and notifies the Contractor that Contractor's progress is such that the Contractor will not achieve full completion of the Work within the Contract Time, the Contractor shall immediately and at no additional cost to the Owner, take all measures necessary, including working such overtime, additional shifts, Sundays, or holidays as may be required to ensure that the entire Project is completed within the Contract Time. Upon receipt of such notice from Construction Manager, Contractor shall immediately notify Construction Manager of all measures to be taken to ensure full Completion of the Work within the Contract Time. The Contractor shall reimburse the Owner for any extra costs or expenses (including the reasonable value of any services provided by Owner's employees) incurred by Owner as the result of such measures.

8.3 DELAYS AND EXTENSIONS OF TIME

- **8.3.1** The Contract Time and Contractor's Schedule will be extended for Permitted Delays and Compensable Delays only in compliance with the applicable provisions of these General Conditions. If Contractor experiences a Permitted Delay, then Contractor will be entitled to request a Change Order pursuant to Section 7.5 and Article 15 for an extension of the Contract Time for the number of days the critical path to Substantial Completion was delayed. Notwithstanding the above, Contractor will not be entitled to any such extension of time unless Contractor (1) notifies Owner and Construction Manager in writing of the cause or causes of the delay in accordance with Article 15 and (2) demonstrates that it could not have anticipated or avoided the delay and has used all available means to minimize the consequences of the delay.
- **8.3.1.1** "Permitted Delay" means any critical path delay in Substantial Completion of the Work beyond the expiration of the Contract Time for Substantial Completion caused by conditions or events beyond the reasonable control of, and without the fault or neglect of, Contractor, its Subcontractors and those for whom they are responsible, and which delay was not and could not in the exercise of reasonable diligence have been avoided. Force Majeure Events (defined below) are Permitted Delays. However, the financial inability of Contractor, a Subcontractor or supplier or any default of any of them, without limitation, will not be deemed a Permitted Delay. Contractor's sole remedy for a Permitted Delay is an extension of the Contract Time, in

accordance with the procedures of the Contract Documents. Permitted Delays will not entitle Contractor to any adjustment of the Contract Sum (unless such delay also qualifies as a Compensable Delay).

- **8.3.1.2** "Force Majeure Events" are natural disasters, union labor strikes that cannot be resolved through use of a dual gate or other measures in Contractor's reasonable control, delays or disruptions in utility service and/or connections due to acts or omissions of Owner or Separate Contractors, civil disobedience, an act of terror or war, or unavoidable casualties or catastrophic events, beyond the control of Contractor and its Subcontractors, and not due to any act or omission of them, that necessarily extends the Contract Time.
- **8.3.2** "Compensable Delay" means any Permitted Delay to the extent caused by (a) the wrongful or negligent acts or omissions of Owner, Construction Manager, or Separate Contractors; (b) changes in the scope of Work directed by Owner for reasons other than a mistake, error or problem resulting from the act or omission of Contractor or any of its Subcontractors; (c) events or conditions that provide for an adjustment of the Contract Sum pursuant to Sections 2.5.1, 3.7.4, 3.7.5, 3.19.1, 9.7 or 14.3; or (d) any other item that is specified as a Compensable Delay elsewhere in the Contract; provided that any such delay is unreasonable under the circumstances involved and not within the contemplation of Contractor on the effective date of this Contract. Contractor's sole remedy for a Compensable Delay is an extension of the Contract Time under Section 8.3.1 and an adjustment of the Contract Sum in accordance with Articles 7 and 15 to the extent that Contractor incurs additional cost allowed under Section 7.3.7 as a result of the Compensable Delay extending the completion of the Work beyond the Contract Time.
- **8.3.3** Contractor expressly agrees that neither Contractor nor any Subcontractor shall have any right to claim or assert a right to an extension of the Contract Time, an adjustment of the Contract Sum or any other legal or equitable relief or damages for any Project delay that is not a Permitted Delay or a Compensable Delay (an "Inexcusable Delay").
- **8.3.4** Any Claim for an extension of the Contract Time due to a Permitted Delay or a Compensable Delay is subject to the following:
 - 1. If more than one Permitted Delay occurs concurrently, or if a Permitted Delay and a Compensable Delay occur concurrently, the maximum extension of the Contract Time is the number of days from the commencement of the first delay to the cessation of the delay which ends last, and Contractor will not be entitled to any adjustment in the Contract Sum;
 - 2. If an Inexcusable Delay occurs concurrently with either a Permitted Delay or a Compensable Delay, the maximum extension of the Contract Time is the number of days during which the delays are concurrent, and Contractor will not be entitled to any adjustment in the Contract Sum; and
 - **3.** Delays in the prosecution of parts or classes of the Work which do not prevent or delay Substantial Completion of the whole Work within the Contract Time are not Permitted or Compensable Delays.

8.3.5 A "delay" means any and every kind of delay, obstruction, hindrance, interference, loss of productivity, or inefficiency of any kind.

ARTICLE 9 PAYMENTS AND COMPLETION 9.1 CONTRACT SUM

The Contract Sum is stated in the Agreement and, including authorized adjustments, is the total amount payable by the Owner to the Contractor for performance of the Work under the Contract Documents.

9.2 SCHEDULE OF VALUES

Where the Contract is based on a stipulated sum or Guaranteed Maximum Price, the Contractor shall submit to the Construction Manager, before the first Application for Payment, a schedule of values allocating the entire Contract Sum to the various portions of the Work and prepared in such form and supported by such data to substantiate its accuracy as the Construction Manager may require. This schedule, unless objected to by the Construction Manager shall be used as a basis for reviewing the Contractor's Applications for Payment.

9.3 APPLICATIONS FOR PAYMENT

- 9.3.1 At least ten days before the date established for each progress payment, the Contractor shall submit to the Construction Manager an itemized Application for Payment prepared in accordance with the schedule of values, if required under Section 9.2., for completed portions of the Work. Such application 5 and supported by such data substantiating the Contractor's right to payment as the Owner or Construction Manager may require, such as copies of requisitions from Subcontractors and material suppliers, and shall reflect retainage if provided for in the Contract Documents.
- **9.3.1.1** As provided in Section 7.3.8, such applications may include requests for payment on account of changes in the Work that have been properly authorized by Construction Change Directives, or by interim determinations of the Construction Manager, but not yet included in Change Orders.
- **9.3.1.2** Applications for Payment shall not include requests for payment for portions of the Work for which the Contractor does not intend to pay a Subcontractor or material supplier, unless such Work has been performed by others whom the Contractor intends to pay.
- **9.3.1.3** Each Application for Payment shall also include the following: (a) Contractor's affidavit covering all payments by Owner pursuant to previous Applications for Payment; (b) statutory conditional waivers and releases of lien upon progress payment from Contractor and each Subcontractor submitting 20-day preliminary notices; (c) statutory unconditional waivers and releases for the previous progress payment from Contractor and each Subcontractor submitting 20-day preliminary notices; (d) Contractor's updated schedule of values; and (e) an updated Contractor's Schedule showing progress towards major milestones. Contractor acknowledges that failure to provide all the required documentation shall result in a delay in the next progress payment to Contractor.

- **9.3.1.4** Contractor will notify Construction Manager of all 20-day preliminary notices receives from Subcontractors or suppliers. Construction Manager will maintain a written record of all 20-day preliminary notices received by it including the manner of receipt, date of receipt, and name and address of person or entity serving the notice.
- **9.3.2** Unless otherwise provided in the Contract Documents, payments shall be made on account of materials and equipment delivered and suitably stored at the site for subsequent incorporation in the Work. If approved in advance by the Owner, payment may similarly be made for materials and equipment suitably stored off the site at a location agreed upon in writing. Payment for materials and equipment stored on or off the site shall be conditioned upon compliance by the Contractor with procedures satisfactory to the Owner to establish the Owner's title to such materials and equipment or otherwise protect the Owner's interest, and shall include the costs of applicable insurance, storage and transportation to the site for such materials and equipment stored off the site.
- **9.3.3** The Contractor warrants that title to all Work covered by an Application for Payment will pass to the Owner no later than the time of payment. The Contractor further warrants that upon submittal of an Application for Payment all Work for which Certificates for Payment have been previously issued and payments received from the Owner shall, to the best of the Contractor's knowledge, information and belief, be free and clear of liens, claims, security interests or encumbrances in favor of the Contractor, Subcontractors, material suppliers, or other persons or entities making a claim by reason of having provided labor, materials and equipment relating to the Work.

9.4 CERTIFICATES FOR PAYMENT

- **9.4.1** In accordance with Section 4.2.19, the Construction Manager will, within seven days after receipt of the Contractor's Application for Payment, review and certificate of amount due. The Construction Manager will, within seven days after receipt of the Application for Payment, either issue to the Owner a Certificate for Payment, with a copy to the Contractor, for such amount as the Construction Manager determines is properly due, or notify the Contractor and Owner in writing of the Construction Manager's reasons for withholding certification in whole or in part as provided in Section 9.5.1.
- 9.4.2 The issuance of a Certificate for Payment will constitute representations made separately by the Construction Manager to the Owner, based on the Construction Manager's evaluation of the Work and the data comprising the Application for Payment, that, to the best of the Construction Manager's knowledge, information and belief, the Work has progressed to the point indicated and that the quality of the Work is in accordance with the Contract Documents. The foregoing representations are subject to an evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, to results of subsequent tests and inspections, to correction of minor deviations from the Contract Documents prior to completion and to specific qualifications expressed by the Construction Manager. The issuance of a Certificate for Payment will further constitute a representation that the Contractor is entitled to payment in the amount certified. However, the issuance of a Certificate for Payment will not be a representation that the Construction Manager has (1) made exhaustive or continuous on-site inspections to check the quality or quantity of the Work, (2) reviewed construction means, methods,

techniques, sequences or procedures, (3) reviewed copies of requisitions received from Subcontractors and material suppliers and other data requested by the Owner to substantiate the Contractor's right to payment, or (4) made examination to ascertain how or for what purpose the Contractor has used money previously paid on account of the Contract Sum.

9.5 DECISIONS TO WITHHOLD CERTIFICATION

- 9.5.1 The Construction Manager may withhold a Certificate for Payment in whole or in part, to the extent reasonably necessary to protect the Owner, if in the Construction Manager's or opinion the representations to the Owner required by Section 9.4.2 cannot be made. If the Construction Manager is unable to certify payment in the amount of the Application, the Construction Manager will notify the Contractor and Owner as provided in Section 9.4.1. If the Contractor, Construction Manager cannot agree on a revised amount, the Construction Manager will promptly issue a Certificate for Payment for the amount for which the Construction Manager is able to make such representations to the Owner. The Construction Manager may also withhold a Certificate for Payment or, because of subsequently discovered evidence, may nullify the whole or a part of a Certificate for Payment previously issued, to such extent as may be necessary in the Construction Manager's opinion to protect the Owner from loss for which the Contractor is responsible, including loss resulting from acts and omissions described in Section 3.3.2, because of
 - 1. Defective Work not remedied;
 - 2. Third party claims filed or reasonable evidence indicating probable filing of such claims unless security acceptable to the Owner is provided by the Contractor;
 - **3.** Failure of the Contractor to make payments properly to Subcontractors or for labor, materials or equipment;
 - **4.** Reasonable evidence that the Work cannot be completed for the unpaid balance of the Contract Sum:
 - **5.** Damage to the Owner or a Separate Contractor;
 - **6.** Reasonable evidence that the Work will not be completed within the Contract Time, and that the unpaid balance would not be adequate to cover actual or liquidated damages for the anticipated delay; or
 - 7. Repeated failure to carry out the Work in accordance with the Contract Documents.
- **9.5.2** When the above reasons for withholding certification are removed, certification will be made for amounts previously withheld with no interest due on withheld amounts.
- **9.5.3** If the Construction Manager withholds certification for payment under Section 9.5.1.3, the Owner may, at its sole option, issue joint checks to the Contractor and to any Subcontractor or material or equipment suppliers to whom the Contractor failed to make payment for Work properly performed or material or equipment suitably delivered. If the Owner makes payments by joint check, the Owner shall notify the Construction Manager and Construction Manager will reflect such payment on the next Certificate for Payment.

9.6 PROGRESS PAYMENTS

- **9.6.1** After the Construction Manager have issued a Certificate for Payment, the Owner shall make payment in the manner and within the time provided in the Contract Documents, and shall so notify the Construction Manager.
- **9.6.2** The Contractor shall pay each Subcontractor no later than seven days after receipt of payment from the Owner the amount to which the Subcontractor is entitled, reflecting percentages actually retained from payments to the Contractor on account of the Subcontractor's portion of the Work. The Contractor shall, by appropriate agreement with each Subcontractor, require each Subcontractor to make payments to Sub-subcontractors in a similar manner.
- **9.6.3** The Construction Manager will, on request, furnish to a Subcontractor, if practicable, information regarding percentages of completion or amounts applied for by the Contractor and action taken thereon by the Owner, Construction Manager on account of portions of the Work done by such Subcontractor.
- **9.6.4** The Owner has the right to request written evidence from the Contractor that the Contractor has properly paid Subcontractors and material and equipment suppliers' amounts paid by the Owner to the Contractor for subcontracted Work. If the Contractor fails to furnish such evidence within seven days, the Owner shall have the right to contact Subcontractors to ascertain whether they have been properly paid. Neither the Owner nor Construction Manager shall have an obligation to pay or to see to the payment of money to a Subcontractor, except as may otherwise be required by law.
- **9.6.5** Contractor payments to material and equipment suppliers shall be treated in a manner similar to that provided in Sections 9.6.2, 9.6.3 and 9.6.4.
- **9.6.6** A Certificate for Payment, a progress payment, or partial or entire use or occupancy of the Project by the Owner shall not constitute acceptance of Work not in accordance with the Contract Documents.
- **9.6.7** Unless the Contractor provides the Owner with a payment bond in the full penal sum of the Contract Sum, payments received by the Contractor for Work properly performed by Subcontractors and suppliers shall be held by the Contractor for those Subcontractors or suppliers who performed Work or furnished materials, or both, under contract with the Contractor for which payment was made by the Owner. Nothing contained herein shall require money to be placed in a separate account and not commingled with money of the Contractor, shall create any fiduciary liability or tort liability on the part of the Contractor for breach of trust or shall entitle any person or entity to an award of punitive damages against the Contractor for breach of the requirements of this provision.
- **9.6.8** If any Subcontractor or any other person or entity claiming to have been employed (directly or indirectly) by or through Contractor or any Subcontractor files a lien affecting Owner or the Project (unless as a result of Owner's failure to pay an amount properly due under this Agreement), Contractor shall satisfy, remove or discharge such lien at Contractor's expense by

bonding, payment or otherwise within 15 days after notice to Contractor of the filing thereof. If Contractor fails to do so, Owner may satisfy, remove or discharge such lien; and, at Owner's election:

- 1. Owner may deduct the cost thereof (including all applicable fees and costs) from Contractor's next succeeding Application for Payment (or may invalidate any previously-approved but unpaid applications for payment) until Owner recoups the total thereof; or
- **2.** Contractor shall pay the same to Owner upon demand.

9.7 FAILURE OF PAYMENT

If the Construction Manager does not issue a Certificate for Payment, through no fault of the Contractor, within seven days after receipt of the Contractor's Application for Payment, or if the Owner does not pay the Contractor within 15 business days after the date established in the Contract Documents the amount certified by the Construction Manager or awarded by binding dispute resolution, then the Contractor may, upon 10 additional business days' written notice to the Owner, Construction Manager, stop the Work until payment of the amount owing has been received. The Contract Time shall be extended appropriately and the Contract Sum shall be increased by the amount of the Contractor's reasonable costs of shut-down, delay and start-up, plus interest as provided for in the Contract Documents.

9.8 SUBSTANTIAL COMPLETION

9.8.1 "Substantial Completion" shall occur (and the Work shall be "substantially complete") when all of the following conditions have been completed to Owner's full satisfaction:

- (a) All Work, excepting minor punch list work and the Close-Out Requirements, has been completed in compliance with the Contract Documents, and the Work is in a condition ready for final inspection by all governmental entities having jurisdiction (although such inspection may not occur until the rest of the Project is ready for occupancy);
- **(b)** All equipment specified or required under the Contract Documents for Contractor to install are connected and functioning properly;
- (c) Completion of Contractor's punch list items and the Close-Out Requirements will cause minimal interference to the rest of the Project;
- (d) Contractor and Construction Manager have agreed in writing upon the final punch list;
- (e) Contractor has delivered to Owner a claim statement setting forth in detail all Contractor's claims against Owner or any Indemnitee connected with, or arising out of, this Contract or the Work and arising out of or based on events prior to the date when Contractor gives such statement to Owner;
- **(f)** Contractor has agreed in writing that Contractor will achieve final completion on a specified date; and

(g) Contractor has delivered to Owner all required manufacturer's (or Subcontractor's) warranties.

Alternatively, Substantial Completion shall occur on any date reasonably certified by Owner, who shall have discretion to reasonably waive any of the foregoing conditions.

- **9.8.1.1** "Close-Out Requirements" means Contractor's obligations for closing out the Work and completing and furnishing to Owner documentation of the Work as required under the Contract Documents (including such matters as submittal of record drawings and operating and other manuals; clean-up and removal from the construction site; and all other matters which the Contract Documents require Contractor to do and perform as part of the completion of Contractor's obligations).
- **9.8.2** When the Contractor considers that the Work, or a portion thereof which the Owner agrees to accept separately, is substantially complete, the Contractor and Construction Manager shall jointly prepare and submit to the Construction Manager a comprehensive list of items to be completed or corrected prior to final payment ("punch list"). Failure to include an item on such list does not alter the responsibility of the Contractor to complete all Work in accordance with the Contract Documents.
- **9.8.3** Upon receipt of the punch list, the Construction Manager will make an inspection to determine whether the Work or designated portion thereof is substantially complete. If the Construction Manager's inspection discloses any item, whether or not included on the punch list, which is not sufficiently complete in accordance with the Contract Documents so that the Work as a whole is not substantially complete, the Contractor shall, before issuance of the Certificate of Substantial Completion, complete or correct such item upon notification by the Construction Manager. In such case, the Contractor shall then submit a request for another inspection by the Construction Manager to determine Substantial Completion.
- **9.8.4** When the Work or designated portion thereof is substantially complete, the Construction Manager will prepare a Certificate of Substantial Completion that shall establish the date of Substantial Completion, shall establish responsibilities of the Owner and Contractor for security, maintenance, heat, utilities, damage to the Work and insurance, and shall fix the time within which the Contractor shall finish all items on the list accompanying the Certificate. Warranties required by the Contract Documents shall commence on the date of Substantial Completion of the Work or designated portion thereof unless otherwise provided in the Certificate of Substantial Completion.
- **9.8.5** The Certificate of Substantial Completion shall be submitted to the Owner and Contractor for their written acceptance of responsibilities assigned to them in such Certificate. Upon such acceptance and consent of surety, if any, the Owner shall make payment of retainage applying to such Work or designated portion thereof. Such payment shall be adjusted for Work that is incomplete or not in accordance with the requirements of the Contract Documents.

9.9 PARTIAL OCCUPANCY OR USE

- **9.9.1** The Owner may occupy or use any completed or partially completed portion of the Work at any stage when such portion is designated by separate agreement with the Contractor, provided such occupancy or use is consented to by the insurer, if required, and authorized by public authorities having jurisdiction over the Project. Such partial occupancy or use may commence whether or not the portion is substantially complete, provided the Owner and Contractor have accepted in writing the responsibilities assigned to each of them for payments, retainage, if any, security, maintenance, heat, utilities, damage to the Work and insurance, and have agreed in writing concerning the period for correction of the Work and commencement of warranties required by the Contract Documents. When the Contractor considers a portion substantially complete, the Construction Manager and Contractor shall jointly prepare and submit a punch list as provided under Section 9.8.2. Consent of the Contractor to partial occupancy or use shall not be unreasonably withheld. The stage of the progress of the Work shall be determined by written agreement between the Owner and Contractor.
- **9.9.2** Immediately prior to such partial occupancy or use, the Owner and Construction Manager, Contractor shall jointly inspect the area to be occupied or portion of the Work to be used in order to determine and record the condition of the Work.
- **9.9.3** Unless otherwise agreed upon, partial occupancy or use of a portion or portions of the Work shall not constitute acceptance of Work not complying with the requirements of the Contract Documents.

9.10 FINAL COMPLETION AND FINAL PAYMENT

- **9.10.1** Upon completion of the Work, the Contractor shall forward to the Construction Manager a written notice that the Work is ready for final inspection and acceptance and shall also forward to the Construction Manager a final Contractor's Application for Payment. Upon receipt of the Contractor's written notice that the Work is ready for final inspection and acceptance and receipt of a final Application for Payment, the Construction Manager will promptly make such inspection. When the Construction Manager finds the Work acceptable under the Contract Documents and the Contract fully performed and the Construction Manager will promptly issue a final Certificate for Payment stating that to the best of the their knowledge, information and belief, and on the basis of the their on-site visits and inspections, the Work has been completed in accordance with terms and conditions of the Contract Documents and that the entire balance found to be due the Contractor and noted in the final Certificate is due and payable. The Construction Manager's final Certificate for Payment will constitute a further representation that conditions listed in Section 9.10.2 as precedent to the Contractor's being entitled to final payment have been fulfilled.
- **9.10.2** Neither final payment nor any remaining retained percentage shall become due until the Contractor submits to the Construction Manager (1) an affidavit that payrolls, bills for materials and equipment, and other indebtedness connected with the Work for which the Owner or the Owner's property might be responsible or encumbered (less amounts withheld by Owner) have been paid or otherwise satisfied, (2) a certificate evidencing that insurance required by the Contract Documents to remain in force after final payment is currently in effect and will not be canceled or allowed to expire until at least 30 days' prior written notice has been given to the

Owner, (3) a written statement that the Contractor knows of no substantial reason that the insurance will not be renewable to cover the period required by the Contract Documents, (4) consent of surety, if any, to final payment, (5) statutory waiver and release upon final payment, and (6), if required by the Owner, other data establishing payment or satisfaction of obligations, such as receipts, releases and waivers of claims, security interests or encumbrances arising out of the Contract, to the extent and in such form as may be designated by the Owner. If a Subcontractor refuses to furnish a release or waiver required by the Owner, the Contractor may furnish a bond satisfactory to the Owner to indemnify the Owner against such lien. If such lien remains unsatisfied after payments are made, the Contractor shall refund to the Owner all money that the Owner may be compelled to pay in discharging such lien, including all costs and reasonable attorneys' fees.

9.10.3 If, after Substantial Completion of the Work, final completion thereof is materially delayed through no fault of the Contractor or by issuance of Change Orders affecting final completion, and the Construction Manager so confirm, the Owner shall, upon application by the Contractor and certification by the Construction Manager, and without terminating the Contract, make payment of the balance due for that portion of the Work fully completed and accepted. If the remaining balance for Work not fully completed or corrected is less than retainage stipulated in the Contract Documents, and if bonds have been furnished, the written consent of surety to payment of the balance due for that portion of the Work fully completed and accepted shall be submitted by the Contractor to the Construction Manager prior to certification of such payment. Such payment shall be made under terms and conditions governing final payment, except that it shall not constitute a waiver of claims.

9.10.4 Not used.

9.10.5 Acceptance of final payment by the Contractor, a Subcontractor or material supplier shall constitute a waiver of claims by that payee except those previously made in writing and identified by that payee as unsettled at the time of final Application for Payment.

ARTICLE 10 PROTECTION OF PERSONS AND PROPERTY 10.1 SAFETY PRECAUTIONS AND PROGRAMS

The Contractor is solely responsible for initiating, maintaining and supervising all safety precautions and programs in connection with the performance of the Contract. The Contractor shall submit one (1) copy of its written safety program to Construction Manager prior to starting work on site. The Contractor shall coordinate its safety program with the safety programs of Separate Contractors so as to provide for a safe project for all involved. The Contractor understands and agrees that Construction Manager is only responsible for the safety of Construction Manager's own employees, subcontractors and agents, and that Construction Manager has no other control nor responsibility for the safety of any other person at the Project site whatsoever.

10.2 SAFETY OF PERSONS AND PROPERTY

- **10.2.1** The Contractor shall take reasonable precautions for safety of, and shall provide reasonable protection to prevent damage, injury or loss to
 - 1. Employees on the Work and other persons who may be affected thereby;

- 2. The Work and materials and equipment to be incorporated therein, whether in storage on or off the site, under care, custody or control of the Contractor or the Contractor's Subcontractors or Sub-subcontractors;
- **3.** Other property at the site or adjacent thereto, such as trees, shrubs, lawns, walks, pavements, roadways, structures and utilities not designated for removal, relocation or replacement in the course of construction; and
- **4.** Construction or operations by the Owner or Separate Contractors.
- **10.2.2** The Contractor shall comply with and give notices required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities bearing on safety of persons or property or their protection from damage, injury or loss.
- **10.2.3** The Contractor shall erect and maintain, as required by existing conditions and performance of the Contract, reasonable safeguards for safety and protection, including posting danger signs and other warnings against hazards, promulgating safety regulations and notifying owners and users of adjacent sites and utilities.
- **10.2.4** When use or storage of explosives or other hazardous materials or equipment or unusual methods are necessary for execution of the Work, the Contractor shall exercise utmost care and carry on such activities under supervision of properly qualified personnel.
- **10.2.5** The Contractor shall promptly remedy damage and loss (other than damage or loss insured under property insurance required by the Contract Documents) to property referred to in Sections 10.2.1.2, 10.2.1.3 and 10.2.1.4 caused in whole or in part by the Contractor, a Subcontractor, a Sub-subcontractor, or anyone directly or indirectly employed by any of them, or by anyone for whose acts they may be liable and for which the Contractor is responsible under Sections 10.2.1.2, 10.2.1.3 and 10.2.1.4, except damage or loss attributable to acts or omissions of the Owner or Construction Manager or anyone directly or indirectly employed by either of them, or by anyone for whose acts either of them may be liable, and not attributable to the fault or negligence of the Contractor. The foregoing obligations of the Contractor are in addition to the Contractor's obligations under Section 3.18.
- **10.2.6** The Contractor shall designate a responsible member of the Contractor's organization at the site whose duty shall be the prevention of accidents. This person shall be the Contractor's superintendent unless otherwise designated by the Contractor in writing to the Owner and Construction Manager.
- **10.2.7** The Contractor shall not permit any part of the construction or site to be loaded so as to cause damage or create an unsafe condition.

10.2.8 INJURY OR DAMAGE TO PERSON OR PROPERTY

If either party suffers injury or damage to person or property because of an act or omission of the other party, or of others for whose acts such party is legally responsible, written notice of such

North County Admin Roof Upgrade Issue for Bid – **Dec 2023**

injury or damage, whether or not insured, shall be given to the other party within a reasonable time not exceeding 21 days after discovery. The notice shall provide sufficient detail to enable the other party to investigate the matter.

10.3 HAZARDOUS MATERIALS

10.3.1 The Contractor is responsible for compliance with any requirements included in the Contract Documents regarding hazardous materials. If the Contractor encounters a hazardous material or substance not addressed in the Contract Documents and if reasonable precautions will be inadequate to prevent foreseeable bodily injury or death to persons resulting from a material or substance, including but not limited to asbestos or polychlorinated biphenyl (PCB), encountered on the site by the Contractor, the Contractor shall, upon recognizing the condition, immediately stop Work in the affected area and report the condition to the Owner, Construction Manager in writing.

10.3.2 Upon receipt of the Contractor's written notice, the Owner shall obtain the services of a licensed laboratory to verify the presence or absence of the material or substance reported by the Contractor and, in the event such material or substance is found to be present, to cause it to be rendered harmless. Unless otherwise required by the Contract Documents, the Owner shall furnish in writing to the Contractor and Construction Manager the names and qualifications of persons or entities who are to perform tests verifying the presence or absence of such material or substance or who are to perform the task of removal or safe containment of such material or substance. The Contractor and the Construction Manager will promptly reply to the Owner in writing stating whether or not either has reasonable objection to the persons or entities proposed by the Owner. If the Contractor or Construction Manager has an objection to a person or entity proposed by the Owner, the Owner shall propose another to whom the Contractor and the Construction Manager have no reasonable objection. When the material or substance has been rendered harmless, Work in the affected area shall resume upon written agreement of the Owner and Contractor. By Change Order, the Contract Time shall be extended appropriately and the Contract Sum shall be increased in the amount of the Contractor's reasonable additional costs of shut-down, delay and start-up.

10.3.3 To the fullest extent permitted by law, the Owner shall indemnify and hold harmless the Contractor, Construction Manager, their consultants, and agents and employees of any of them from and against claims, damages, losses and expenses, including but not limited to attorneys' fees, arising out of or resulting from performance of the Work in the affected area if in fact the material or substance presents the risk of bodily injury or death as described in Section 10.3.1 and has not been rendered harmless, provided that such claim, damage, loss or expense is attributable to bodily injury, sickness, disease or death, or to injury to or destruction of tangible property (other than the Work itself), except to the extent that such damage, loss or expense is due to the fault or negligence of the party seeking indemnity.

10.3.4 The Owner shall not be responsible under this Section 10.3 for materials or substances the Contractor brings to the site unless such materials or substances are required by the Contract Documents. The Owner shall be responsible for materials or substances required by the Contract Documents, except to the extent of the Contractor's fault or negligence in the use and handling of such materials or substances.

- 10.3.5 The Contractor shall indemnify the Owner for the cost and expense the Owner incurs (1) for remediation of a material or substance the Contractor brings to the site and negligently handles, or (2) where the Contractor fails to perform its obligations under Section 10.3.1, except to the extent that the cost and expense are due to the Owner's fault or negligence.
- **10.3.6** If, without negligence on the part of the Contractor, the Contractor is held liable by a government agency for the cost of remediation of a hazardous material or substance solely by reason of performing Work as required by the Contract Documents, the Owner shall indemnify the Contractor for all cost and expense thereby incurred.

10.4 EMERGENCIES

In an emergency affecting safety of persons or property, the Contractor shall act, at the Contractor's discretion, to prevent threatened damage, injury or loss. Additional compensation or extension of time claimed by the Contractor on account of an emergency shall be determined as provided in Article 15 and Article 7.

ARTICLE 11 INSURANCE AND BONDS 11.1 CONTRACTOR'S LIABILITY INSURANCE

- 11.1.1 The Contractor shall purchase from and maintain in a company or companies lawfully authorized to do business in the jurisdiction in which the Project is located such insurance as will protect the Contractor from claims set forth below which may arise out of or result from the Contractor's operations and completed operations under the Contract and for which the Contractor may be legally liable, whether such operations be by the Contractor or by a Subcontractor or by anyone directly or indirectly employed by any of them, or by anyone for whose acts any of them may be liable:
 - 1. Claims under workers' compensation, disability benefit and other similar employee benefit acts that are applicable to the Work to be performed;
 - 2. Claims for damages because of bodily injury, occupational sickness or disease, or death of the Contractor's employees;
 - **3.** Claims for damages because of bodily injury, sickness or disease, or death of any person other than the Contractor's employees;
 - **4.** Claims for damages insured by usual personal injury liability coverage;
 - 5. Claims for damages, other than to the Work itself, because of injury to or destruction of tangible property, including loss of use resulting therefrom;
 - **6.** Claims for damages because of bodily injury, death of a person or property damage arising out of ownership, maintenance or use of a motor vehicle;
 - 7. Claims for bodily injury or property damage arising out of completed operations; and
 - **8.** Claims involving contractual liability insurance applicable to the Contractor's obligations under Section 3.18.
- **11.1.2** The insurance required by Section 11.1.1 shall be written for not less than limits of liability specified in the Supplementary Conditions or required by law, whichever coverage is greater. Commercial general liability and automobile liability insurance coverages must be written on the occurrence basis. Coverages, whether written on an occurrence or claims-made

basis, shall be maintained without interruption from the date of commencement of the Work until the date of final payment and termination of any coverage required to be maintained after final payment, and, with respect to the Contractor's completed operations coverage, until the expiration of the period for correction of Work or for such other period for maintenance of completed operations coverage as specified in the Contract Documents.

- 11.1.3 Certificates of insurance acceptable to the Owner shall be filed with the Owner prior to commencement of the Work and thereafter upon renewal or replacement of each required policy of insurance. The insurance policies required by this Section 11.1 shall contain a provision that coverages afforded under the policies will not be canceled or allowed to expire until at least 30 days' prior written notice has been given to the Owner. An additional certificate evidencing continuation of liability coverage, including coverage for completed operations, shall be submitted with the final Application for Payment as required by Section 9.10.2 and thereafter upon renewal or replacement of such coverage until the expiration of the time required by Section 11.1.2. Information concerning reduction of coverage on account of revised limits or claims paid under the General Aggregate, or both, shall be furnished by the Contractor with reasonable promptness and prior to the effectiveness of any reduced limits. Failure to so notify is a material breach of this Contract.
- 11.1.4 The Contractor shall cause the commercial liability coverage and automobile liability coverage required by the Contract Documents to include (1) the Owner, the Owner's consultants, the Construction Manager, the Construction Manager's consultants and the employees, officers and affiliates of the foregoing as additional insureds for claims arising from the Contractor's operations; and (2) the Owner as an additional insured for claims arising from the Contractor's completed operations.

11.2 OWNER'S LIABILITY INSURANCE

The Owner shall be responsible for purchasing and maintaining the Owner's usual liability insurance.

11.3 BUILDER'S RISK PROPERTY INSURANCE

- 11.3.1 The Owner will provide builder's risk property insurance, with a \$25,000 deductible as required by the General Conditions if the requirements of the Project are not excluded by such coverage. Bidder agrees that the Owner's provision of builder's risk property insurance containing said provisions meets the Owner's obligation to provide builder's risk property insurance under the Contract and, in the event of a conflict between the provisions of the policy and any summary or description of the provisions contained herein or otherwise, the provisions of the policy shall control and shall be conclusively presumed to fulfill the Owner's obligation to provide such insurance. See also, Section 3.19.1.1.
- **11.3.2** A copy of the Owner's builder's risk property insurance policy for the Project will be provided to the bidder awarded the Contract.
- **11.3.3** A loss insured under the Owner's property insurance shall be adjusted solely by the Owner and made payable to the Owner, subject to requirements of any applicable mortgagee clause. Owner shall apply proceeds to paying Contractor for damaged Work pursuant to Article

7. The Contractor shall pay Subcontractors their just shares of insurance proceeds received by the Contractor, and by appropriate written agreements shall require Subcontractors to make payments to their Sub-subcontractors in similar manner.

11.3.4 WAIVERS OF SUBROGATION

The Owner and Contractor waive all rights against (1) each other and any of their subcontractors, sub-subcontractors, agents and employees, each of the other, and (2) the Construction Manager, Construction Manager's consultants, Separate Contractors described in Article 6, if any, and any of their subcontractors, sub-subcontractors, agents and employees, for damages caused by fire or other causes of loss to the extent covered by property insurance obtained pursuant to this Section 11.3 or other property insurance applicable to the Work, except such rights as they have to proceeds of such insurance. The Owner or Contractor, as appropriate, shall require of the Construction Manager, Construction Manager's consultants, Separate Contractors described in Article 6, if any, and the subcontractors, sub-subcontractors, agents and employees of any of them, by appropriate written agreements, similar waivers each in favor of other parties enumerated herein. The policies shall provide such waivers of subrogation by endorsement or otherwise. A waiver of subrogation shall be effective as to a person or entity even though that person or entity would otherwise have a duty of indemnification, contractual or otherwise, did not pay the insurance premium directly or indirectly, and whether or not the person or entity had an insurable interest in the property damaged.

11.4 PERFORMANCE BOND AND PAYMENT BOND

- **11.4.1** The Owner shall have the right to require the Contractor to furnish bonds covering faithful performance of the Contract and payment of obligations arising thereunder as stipulated in bidding requirements or specifically required in the Contract Documents on the date of execution of the Contract.
- **11.4.2** Upon the request of any person or entity appearing to be a potential beneficiary of bonds covering payment of obligations arising under the Contract, the Contractor shall promptly furnish a copy of the bonds or shall authorize a copy to be furnished.

ARTICLE 12 UNCOVERING AND CORRECTION OF WORK 12.1 UNCOVERING OF WORK

- **12.1.1** If a portion of the Work is covered contrary to the Construction Manager's request or to requirements specifically expressed in the Contract Documents, it must, if requested in writing by the Construction Manager, be uncovered for the Construction Manager examination and be replaced at the Contractor's expense without change in the Contract Time.
- **12.1.2** If a portion of the Work has been covered that the Construction Manager has not specifically requested to examine prior to it's being covered, the Construction Manager may request to see such Work and it shall be uncovered by the Contractor. If such Work is in accordance with the Contract Documents, costs of uncovering and replacement shall, by appropriate Change Order, be at the Owner's expense. If such Work is not in accordance with the Contract Documents, such costs and the cost of correction shall be at the Contractor's expense

unless the condition was caused by the Owner or a Separate Contractor in which event the Owner shall be responsible for payment of such costs.

12.2 CORRECTION OF WORK

12.2.1 BEFORE OR AFTER SUBSTANTIAL COMPLETION

The Contractor shall promptly correct Work rejected by the Construction Manager or failing to conform to the requirements of the Contract Documents, whether discovered before or after Substantial Completion and whether or not fabricated, installed or completed. Costs of correcting such rejected Work, including additional testing and inspections, the cost of uncovering and replacement, and compensation for the Construction Manager's services and expenses made necessary thereby, shall be at the Contractor's expense. Contractor shall replace, repair, or restore to Owner's satisfaction any other parts of the Work and any other real or personal property which is damaged or destroyed as a result of defective work or the correction of defective work.

12.2.2 AFTER SUBSTANTIAL COMPLETION

- 12.2.2.1 In addition to the Contractor's obligations under Section 3.5, if, within one year after the date of Substantial Completion of the Work or designated portion thereof or after the date for commencement of warranties established under Section 9.9.1, or by terms of an applicable special warranty required by the Contract Documents, any of the Work is found to be not in accordance with the requirements of the Contract Documents, the Contractor shall correct it promptly after receipt of written notice from the Owner to do so unless the Owner has previously given the Contractor a written acceptance of such condition. The Owner shall give such notice promptly after discovery of the condition. If the Contractor fails to correct nonconforming Work within a reasonable time after receipt of notice from the Owner or Construction Manager, the Owner may correct it in accordance with Section 2.4.
- **12.2.2.2** The one-year period for correction of Work shall be extended with respect to portions of Work first performed after Substantial Completion by the period of time between Substantial Completion and the actual completion of that portion of the Work.
- **12.2.2.3** The one-year period for correction of Work shall not be extended by corrective Work performed by the Contractor pursuant to this Section 12.2.
- **12.2.3** The Contractor shall remove from the site portions of the Work that are not in accordance with the requirements of the Contract Documents and are neither corrected by the Contractor nor accepted by the Owner.
- **12.2.4** The Contractor shall bear the cost of correcting destroyed or damaged construction, whether completed or partially completed, of the Owner or Separate Contractors caused by the Contractor's correction or removal of Work that is not in accordance with the requirements of the Contract Documents.
- **12.2.5** Nothing contained in this Section 12.2 shall be construed to establish a period of limitation with respect to other obligations the Contractor has under the Contract Documents. Establishment of the one-year period for correction of Work as described in Section 12.2.2

North County Admin Roof Upgrade Issue for Bid – **Dec 2023**

relates only to the specific obligation of the Contractor to correct the Work, and has no relationship to the time within which the obligation to comply with the Contract Documents may be sought to be enforced, nor to the time within which proceedings may be commenced to establish the Contractor's liability with respect to the Contractor's obligations other than specifically to correct the Work.

12.3 ACCEPTANCE OF NONCONFORMING WORK

If the Owner prefers to accept Work that is not in accordance with the requirements of the Contract Documents, the Owner may do so instead of requiring its removal and correction, in which case the Contract Sum will be reduced as appropriate and equitable. Such adjustment shall be effected whether or not final payment has been made.

ARTICLE 13 MISCELLANEOUS PROVISIONS 13.1 GOVERNING LAW

The Contract shall be governed by the law of the place where the Project is located except that, if the parties have selected arbitration as the method of binding dispute resolution, the Federal Arbitration Act shall govern Section 15.4.

13.2 SUCCESSORS AND ASSIGNS

13.2.1 The Owner and Contractor respectively bind themselves, their partners, successors, assigns and legal representatives to covenants, agreements and obligations contained in the Contract Documents. Except as provided in Section 13.2.2, neither party to the Contract shall assign the Contract as a whole without written consent of the other. If either party attempts to make such an assignment without such consent, that party shall nevertheless remain legally responsible for all obligations under the Contract.

13.2.2 The Owner may, upon advance written notice to Contractor, assign this Agreement to an affiliated entity (as defined in California Corporations Code Section 150) or merger partners of Owner, and such assignment will not require Contractor's consent. Also, the Owner may, without consent of the Contractor, assign the Contract to a lender providing construction financing for the Project. The Contractor shall execute all consents reasonably required to facilitate any such assignment.

13.3 WRITTEN NOTICE

Written notice shall be deemed to have been duly served if delivered in person to the individual, to a member of the firm or entity, or to an officer of the corporation for which it was intended; or if delivered at, or sent by registered or certified mail or by courier service providing proof of delivery to, the last business address known to the party giving notice.

13.4 RIGHTS AND REMEDIES

3.4.1 Duties and obligations imposed by the Contract Documents and rights and remedies available thereunder shall be in addition to and not a limitation of duties, obligations, rights and remedies otherwise imposed or available by law.

13.4.2 No action or failure to act by the Owner, Construction Manager or Contractor shall constitute a waiver of a right or duty afforded them under the Contract, nor shall such action or failure to act constitute approval of or acquiescence in a breach there under, except as may be specifically agreed in writing.

13.5 TESTS AND INSPECTIONS

- 13.5.1 Tests, inspections and approvals of portions of the Work shall be made as required by the Contract Documents and by applicable laws, statutes, ordinances, codes, rules and regulations or lawful orders of public authorities. Unless otherwise provided, the Contractor shall make arrangements for such tests, inspections and approvals with an independent testing laboratory or entity acceptable to the Owner, or with the appropriate public authority, and shall bear all related costs of tests, inspections and approvals. The Contractor shall give the Construction Manager timely notice of when and where tests and inspections are to be made so that the Construction Manager may be present for such procedures. The Owner shall bear costs of (1) tests, inspections or approvals that do not become requirements until after bids are received or negotiations concluded, and (2) tests, inspections or approvals where building codes or applicable laws or regulations prohibit the Owner from delegating their cost to the Contractor.
- 13.5.2 If the Construction Manager, Owner or public authorities having jurisdiction determine that portions of the Work require additional testing, inspection or approval not included under Section 13.5.1, the Construction Manager will, upon written authorization from the Owner, instruct the Contractor to make arrangements for such additional testing, inspection or approval by an entity acceptable to the Owner, and the Contractor shall give timely notice to the Construction Manager of when and where tests and inspections are to be made so that the Construction Manager may be present for such procedures. Such costs, except as provided in Section 13.5.3, shall be at the Owner's expense.
- **13.5.3** If such procedures for testing, inspection or approval under Sections 13.5.1 and 13.5.2 reveal failure of the portions of the Work to comply with requirements established by the Contract Documents, all costs made necessary by such failure including those of repeated procedures and compensation for the Construction Manager's services and expenses shall be at the Contractor's expense.
- **13.5.4** Required certificates of testing, inspection or approval shall, unless otherwise required by the Contract Documents, be secured by the Contractor and promptly delivered to the Construction Manager.
- **13.5.5** If the Construction Manager is to observe tests, inspections or approvals required by the Contract Documents, the Construction Manager will do so promptly and, where practicable, at the normal place of testing.
- **13.5.6** Tests or inspections conducted pursuant to the Contract Documents shall be made promptly to avoid unreasonable delay in the Work.

13.6 INTEREST

Payments due and unpaid under the Contract Documents shall bear interest from the date payment is due at such rate as the parties may agree upon in writing or, in the absence thereof, at the legal rate prevailing from time to time at the place where the Project is located.

13.7 TIME LIMITS ON CLAIMS

The Owner and Contractor shall commence all claims and causes of action, whether in contract, tort, breach of warranty or otherwise, against the other arising out of or related to the Contract in accordance with the requirements of the final dispute resolution method selected in the Agreement within the time period specified by applicable law, but in any case not more than 10 years after the date of Substantial Completion of the Work. The Owner and Contractor waive all claims and causes of action not commenced in accordance with this Section 13.7.

13.7 MEDICARE AUDIT

Pursuant to Section 952 of P.L. 96-499, which governs access to books and records of contractors and subcontractors to Medicare providers where the cost or value of the Work under the contract exceeds \$10,000 over a 12-month period, each Contractor and its Subcontractors agree to permit representatives of the Secretary of the Department of Health and Human Services and the Comptroller General, in accordance with criteria and procedures contained in applicable Federal regulations, to have access to their books, documents and records as necessary to verify the nature and extent of the cost of the Work for a period of 4 years from Final Completion.

ARTICLE 14 TERMINATION OR SUSPENSION OF THE CONTRACT 14.1 TERMINATION BY THE CONTRACTOR

14.1.1 The Contractor may terminate the Contract upon 7 days prior notice to Owner if the Work is stopped for a period of 60 consecutive days through no act or fault of the Contractor or a Subcontractor, Sub-subcontractor or their agents or employees or any other persons or entities performing portions of the Work under direct or indirect contract with the Contractor, for any of the following reasons. If underlying cause of the Work suspension is removed prior to the end of the 7 day notice period, then the Contract is not terminated.

- 1. Issuance of an order of a court or other public authority having jurisdiction that requires all Work to be stopped;
- 2. An act of government, such as a declaration of national emergency that requires all Work to be stopped;
- 3. Because the Construction Manager not issued a Certificate for Payment and has not notified the Contractor of the reason for withholding certification as provided in Section 9.4.1, or because the Owner has not made payment on a Certificate for Payment within the time stated in the Contract Documents, and Contractor has duly suspended the Work per the terms of Section 9.7; or
- **4.** The Owner has failed to furnish to the Contractor promptly, upon the Contractor's request, reasonable evidence as required by Section 2.2.1.

14.1.2 The Contractor may terminate the Contract if, through no act or fault of the Contractor or a Subcontractor, Sub-subcontractor or their agents or employees or any other persons or entities performing portions of the Work under direct or indirect contract with the Contractor, repeated suspensions, delays or interruptions of the entire Work by the Owner as described in Section

- 14.3 constitute in the aggregate more than 100 percent of the total number of days scheduled for completion, or 120 days in any 365-day period, whichever is less.
- **14.1.3** If Contractor terminates the Contract under Sections 14.1.1 or 14.1.2, then the Contractor may recover from the Owner payment for Work executed and reasonable costs incurred due to termination, but no more than the Contract Sum in total.
- **14.1.4** If the Work is stopped for a period of 60 consecutive days through no act or fault of the Contractor or a Subcontractor or their agents or employees or any other persons performing portions of the Work under contract with the Contractor because the Owner has repeatedly failed to fulfill the Owner's obligations under the Contract Documents with respect to matters important to the progress of the Work, the Contractor may, upon seven additional days' written notice to the Owner, Construction, terminate the Contract and recover from the Owner as provided in Section 14.1.3.

14.2 TERMINATION BY THE OWNER FOR CAUSE

- **14.2.1** The Owner may terminate the Contract if the Contractor
 - 1. Repeatedly refuses or fails to supply enough properly skilled workers or proper materials;
 - **2.** Fails to make payment to Subcontractors for materials or labor in accordance with the respective agreements between the Contractor and the Subcontractors;
 - **3.** Repeatedly disregards applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of a public authority; or
 - **4.** Otherwise, is guilty of substantial breach of a provision of the Contract Documents.
- **14.2.2** When any of the above reasons exist, the Owner may without prejudice to any other rights or remedies of the Owner and after giving the Contractor and the Contractor's surety, if any, seven days' written notice, terminate employment of the Contractor and may subject to any prior rights of the surety:
 - 1. Exclude the Contractor from the site and take possession of all materials, equipment, tools, and construction equipment and machinery thereon owned by the Contractor;
 - 2. Accept assignment of subcontracts pursuant to Section 5.4; and
 - 3. Finish the Work by whatever reasonable method the Owner may deem expedient. Upon written request of the Contractor, the Owner shall furnish to the Contractor a detailed accounting of the costs incurred by the Owner in finishing the Work.
- **14.2.3** When the Owner terminates the Contract for one of the reasons stated in Section 14.2.1, the Contractor shall not be entitled to receive further payment until the Work is finished.
- **14.2.4** If the unpaid balance of the Contract Sum exceeds costs of finishing the Work, including compensation for the Construction Manager's services and expenses made necessary thereby, and other damages incurred by the Owner and not expressly waived, such excess shall be paid to the Contractor. If such costs and damages exceed the unpaid balance, the Contractor shall promptly pay the difference to the Owner upon demand. The obligations in this paragraph shall survive termination of the Contract.

14.2.5 If a court of competent jurisdiction determines that termination of Contractor was wrongful or otherwise improper, the termination will be deemed a termination for convenience under Section 14.4.

14.3 SUSPENSION BY THE OWNER FOR CONVENIENCE

- **14.3.1** The Owner may, without cause, order the Contractor in writing to suspend, delay or interrupt the Work in whole or in part for such period of time as the Owner may determine. Such suspension will be a Compensable Delay; however, Owner's suspension pursuant to another provision of the Contract is not a suspension under this Section 14.3.
- **14.3.2** No adjustment of the Contract Sum or Contract Time for Owner's suspension under Section 14.3.1 shall be made to the extent
 - 1. That performance is, was or would have been so suspended, delayed or interrupted by another cause for which the Contractor is responsible; or
 - **2.** That an adjustment of the Contract Sum or Contract Time is made or denied under another provision of the Contract.

14.4 TERMINATION BY THE OWNER FOR CONVENIENCE

- **14.4.1** The Owner may, at any time, upon written notice terminate the Contract for the Owner's convenience and without cause.
- **14.4.2** Upon receipt of written notice from the Owner of such termination for the Owner's convenience, the Contractor shall
 - 1. Cease operations as directed by the Owner in the notice;
 - **2.** Take actions necessary, or that the Owner may direct, for the protection and preservation of the Work; and
 - **3.** Except for Work directed to be performed prior to the effective date of termination stated in the notice, terminate all subcontracts and purchase orders that Owner is not taking by assignment, and enter into no further subcontracts and purchase orders.
- **14.4.3** In case of such termination for the Owner's convenience, the Contractor shall be entitled to receive payment for Work executed and reasonable costs incurred due to termination, but no more than the Contract Sum in total.

ARTICLE 15 CLAIMS AND DISPUTES 15.1 CLAIMS 15.1.1 DEFINITION

A Claim is a demand or assertion by one of the parties seeking, as a matter of right, payment of money, or other relief with respect to the terms of the Contract. The term "Claim" also includes other disputes and matters in question between the Owner and Contractor arising out of or relating to the Contract. The responsibility to substantiate Claims shall rest with the party making the Claim.

15.1.2 NOTICE OF CLAIMS

Claims by either the Owner or Contractor must be initiated by written notice to the other party with a copy sent to the Construction Manager. Claims by either party must be initiated within 7 days after the later of the occurrence of the event giving rise to such Claim or when the claimant first recognizes (or reasonably should have recognized) the condition giving rise to the Claim. Any claim by Contractor for (a) an adjustment in the Contract Sum for changes in the Work or (b) for an increase in the Contract Time shall be waived if not provided to Construction Manager and Owner in writing within the time specified in this Section 15.1.2. Contractor acknowledges that Owner's receipt of any notices for Claims within the time period specified in this Section 15.1.2 is of the essence of this Contract.

15.1.3 CONTINUING CONTRACT PERFORMANCE

Pending final resolution of a Claim, except as otherwise agreed in writing or as provided in Section 9.7 and Article 14, the Contractor shall proceed diligently with performance of the Contract and the Owner shall continue to make payments of undisputed amounts in accordance with the Contract Documents.

15.1.4 CLAIMS FOR ADDITIONAL COST

If the Contractor wishes to make a Claim for an increase in the Contract Sum, written notice as provided herein shall be given before proceeding to execute the Work. Prior notice is not required for Claims relating to an emergency endangering life or property arising under Section 10.4.

15.1.5 CLAIMS FOR ADDITIONAL TIME

15.1.5.1 If the Contractor wishes to make a Claim for an increase in the Contract Time, written notice as provided herein shall be given. The Contractor's Claim shall include an estimate of probable effect of delay on progress of the Work. In the case of a continuing delay, only one Claim is necessary.

15.1.5.2 If adverse weather conditions are the basis for a Claim for additional time, such Claim shall be documented by data substantiating that weather conditions were abnormal for the period of time, could not have been reasonably anticipated, had an adverse effect on the scheduled construction and exceeded the number of weather delays built into the schedule and shown therein.

15.1.5.3 CONTRACTOR CLAIM DOCUMENTATION REQUIREMENTS

A Claim must include the following:

- 1. A detailed factual narrative of events fully describing the nature and circumstances giving rise to the Claim, including but not limited to, necessary dates, locations, and items of work affected.
- **2.** A statement demonstrating that a Change Order Request was timely submitted as required by

Section 7.5.3 of the General Conditions.

3. A detailed justification for any remedy or relief sought by the Claim including an explanation of the contractual provisions allowing the Claim, and a detailed cost breakdown of the amounts claimed, including the items specified in Section 7.3.7 of the General Conditions. The cost breakdown must be provided even if the costs claimed have not been incurred when the Claim is submitted.

15.1.6 CLAIMS FOR CONSEQUENTIAL DAMAGES

The Contractor and Owner waive Claims against each other for consequential damages arising out of or relating to this Contract. This mutual waiver includes

- 1. Damages incurred by the Owner for rental expenses, for losses of use, income, profit, financing, business and reputation, and for loss of management or employee productivity or of the services of such persons; and
- 2. Damages incurred by the Contractor for principal office expenses including the compensation of personnel stationed there, for losses of financing, business and reputation, and for loss of profit except profit included in the Contract Sum on account of completed Work.

This mutual waiver is applicable, without limitation, to all consequential damages due to either party's termination in accordance with Article 14. Nothing contained in this Section 15.1.6 shall be deemed to preclude an award of liquidated damages, when applicable, in accordance with the requirements of the Contract Documents.

15.2 INITIAL DECISION

- **15.2.1** Claims by the Contractor (for itself or on a pass-through basis) shall be referred to the Initial Decision Maker for initial decision. An initial decision shall be required as a condition precedent to litigation of any Claim by the Contractor arising prior to the date final payment is due, unless 30 days have passed after the Claim has been referred to the Initial Decision Maker with no decision having been rendered.
- **15.2.2** The Initial Decision Maker will review Claims and within ten days of the receipt of a Claim take one or more of the following actions: (1) request additional supporting data from the claimant or a response with supporting data from the other party, (2) reject the Claim in whole or in part, (3) approve the Claim, or (4) suggest a compromise.
- **15.2.3** In evaluating Claims, the Initial Decision Maker may, but shall not be obligated to, consult with or seek information from either party or from persons with special knowledge or expertise who may assist the Initial Decision Maker in rendering a decision, at the Owner's expense.
- **15.2.4** If the Initial Decision Maker requests a party to provide a response to a Claim or to furnish additional supporting data, such party shall respond, within ten days after receipt of such request, and shall either (1) provide a response on the requested supporting data, (2) advise the Initial Decision Maker when the response or supporting data will be furnished or (3) advise the Initial Decision Maker that no supporting data will be furnished. Upon receipt of the response or

North County Admin Roof Upgrade Issue for Bid – **Dec 2023**

supporting data, if any, the Initial Decision Maker will either reject or approve the Claim in whole or in part.

15.2.5 The Initial Decision Maker will render an initial decision approving or rejecting the Claim or suggesting a compromise. This initial decision shall (1) be in writing; (2) state the reasons therefor; and (3) notify the parties, and the Construction Manager of any change in the Contract Sum or Contract Time or both. The initial decision shall stand for the duration of the Project, but may be overturned by a legal proceeding after Final Completion or termination of this Agreement, in which the court need not give any deference to the Initial Decision Maker's decision.

15.2.6 Not used.

- **15.2.7** In the event of a Claim against the Contractor, the Owner may, but is not obligated to, notify the surety, if any, of the nature and amount of the Claim. If the Claim relates to a possibility of a Contractor's default, the Owner may, but is not obligated to, notify the surety and request the surety's assistance in resolving the controversy.
- **15.2.8** Without limiting Section 9.6.8, if a Claim relates to or is the subject of a mechanic's lien, the party asserting such Claim may proceed in accordance with applicable law to comply with the lien notice or filing deadlines.

15.3 LITIGATION

- **15.3.1** If the parties are unable to resolve a dispute under Section 15.2, then after Final Completion or termination of this Agreement either party may pursue any legal or equitable relief afforded them under the Contract.
- **15.3.1** Both parties agree that any claim or enforcement of a judgment or alternative dispute award will be filed with the appropriate court in Glenn County.

ARTICLE 16 STATUTORY AND OTHER REQUIREMENTS

16.1 Not Used

16.2 NONDISCRIMINATION

- **16.2.1** For purposes of this Article 16.2, the term Subcontractor shall not include suppliers, manufacturers, or distributors.
- **16.2.2** Contractor shall comply and shall ensure that all Subcontractors comply with Section 12900 through 12996, of the State of California Government Code.
- **16.2.3** Contractor agrees as follows during the performance of the Work:
 - 1. Contractor shall not willfully discriminate against any employee or applicant for employment because of race, color, religion, sex, age, ancestry, national origin, sexual orientation, handicap, veteran's status, medical condition (as defined in

Section 12926 of the State of California Government Code), marital status, or citizenship (within the limits imposed by law) because of habit, local custom, or otherwise. All applicants for employment and employees are to be treated without regard to their race, color, religion, sex, age, ancestry, national origin, sexual orientation, handicap, veteran's status, medical condition (as defined in Section 12926 of the State of California Government Code), marital status, or citizenship (within the limits imposed by law). Such equal treatment shall apply, but not be limited to, employment, upgrading, demotion, or transfer; recruitment or recruitment advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship.

- 2. Contractor and all Subcontractors will permit access to their records of employment, employment advertisements, application forms, and other pertinent data and records by Owner or any appropriate agency of the State of California designated by Owner for the purposes of investigation to ascertain compliance with this Article 16.2. The outcome of the investigation may result in the following:
 - **A.** A finding of willful violation of the provisions of this Contract or of the Fair Employment Practices Act may be regarded by Owner as (1) a basis for determining that Contractor is not a "responsible bidder" as to future contracts for which such Contractor may submit bids or (2) a basis for refusing to accept or consider the bids of Contractor for future contracts.
 - **B.** Owner may deem a finding of willful violation of the Fair Employment Practices Act to have occurred upon receipt of written notice from the Fair Employment Practices Commission that it has (1) investigated and determined that Contractor has violated the Fair Employment Practices Act and (2) issued an order under the State of California Government Code Section 12970 or obtained an injunction under Government Code Section 12973.
 - C. Upon receipt of such written notice from the Fair Employment Practices Commission, Owner may notify Contractor that, unless it demonstrates to the satisfaction of Owner within a stated period that the violation has been corrected, Contractor's bids on future projects will not be considered.
 - **D.** Contractor agrees that, should Owner determine that Contractor has not complied with this Article 16.2, Contractor shall forfeit to Owner, as a penalty, for each day or portion thereof, for each person who was denied employment as a result of such non-compliance, the penalties provided in Article 16.3 below for violation of prevailing wage rates. Such penalty amounts may be recovered from Contractor; and Owner may deduct any such penalty amounts from the Contract Sum.
 - **E.** Nothing contained in this Article 16.2 shall be construed in any manner so as to prevent Owner from pursuing any other remedies that may be available at law.

- **F.** Contractor shall meet the following standards for compliance and provide Owner with satisfactory evidence of such compliance upon Owner's request, which shall be evaluated in each case by Owner:
 - i) Contractor shall notify its Superintendent and other supervisory personnel of the nondiscrimination requirements of the Contract Documents and their responsibilities thereto.
 - ii) Contractor shall notify all sources of employee referrals (including unions, employment agencies, and the State of California Department of Employment) of the nondiscrimination requirements of the Contract Documents by sending to such sources and by posting the Notice of Equal Employment Opportunity (EEO).
 - iii) Contractor or its representative shall, through all unions with whom it may have agreements, develop agreements that (1) define responsibilities for nondiscrimination in hiring, referrals, upgrading, and training and (2) implement an affirmative nondiscrimination program, in terms of the unions' specific areas of skill and geography, such that qualified minority women, non-minority women, and minority men shall be available and given an equal opportunity for employment.
 - **iv)** Contractor shall notify Owner of opposition to the nondiscrimination requirements of the Contract Documents by individuals, firms, or organizations during the term of the Contract.
- **G.** Contractor shall include the provisions of the foregoing Articles 16.2.3.1 through 16.2.3.6 in all subcontracts with Subcontractors, so that such provisions will be binding upon each such Subcontractor.

16.3 PREVAILING WAGE RATES

- **16.3.1** For purposes of this Article 16.3, the term Subcontractor shall not include suppliers, manufacturers, or distributors.
- **16.3.2** Contractor shall comply and shall ensure that all Subcontractors comply with Sections 1770, 1771, 1772, 1773, 1774, and 1775 of the State of California Labor Code. Compliance with these sections is required by this Contract.
- 16.3.3 The State of California Department of Industrial Relations has ascertained the general prevailing per diem wage rates in the locality in which the Work is to be performed for each craft, classification, or type of worker required to perform the Work. A copy of the general prevailing per diem wage rates will be on file at Owner's principal facility office and will be made available to any interested party upon request. Contractor shall post a copy of the general prevailing per diem wage rates at the job site. By this reference, such schedule is made part of the Contract Documents. Contractor shall pay not less than the prevailing wage rates, as specified in the schedule and any amendments thereto, to all workers employed by Contractor in

the execution of the Work. Contractor shall cause all subcontracts to include the provision that all Subcontractors shall pay not less than the prevailing rates to all workers employed by such Subcontractors in the execution of the Work. Contractor shall forfeit to Owner, as a penalty, not more than \$50 for each calendar day or portion thereof for each worker that is paid less than the prevailing rates as determined by the Director of Industrial Relations for the work or craft in which the worker is employed for any portion of the Work done by Contractor or any Subcontractor. The amount of this penalty shall be determined pursuant to applicable law. Such forfeiture amounts may be deducted from the Contract Sum or sought directly from the surety under its Performance Bond if there are insufficient funds remaining in the Contract Sum. Contractor shall also pay to any worker who was paid less than the prevailing wage rate for the work or craft for which the worker was employed for any portion of the Work, for each day, or portion thereof, for which the worker was paid less than the specified prevailing per diem wage rate, an amount equal to the difference between the specified prevailing per diem wage rate and the amount which was paid to the worker. Review of any civil wage and penalty assessment shall be made pursuant to section 17420 of the California Labor Code.

16.4 PAYROLL RECORDS

16.4.1 For purposes of this Article 16.4, the term Subcontractor shall not include suppliers, manufacturers, or distributors.

16.4.2 Contractor and all Subcontractors shall keep an accurate payroll record, showing the name, address, social security number, job classification, straight time and overtime hours worked each day and week, and the actual per diem wages paid to each journey worker, apprentice, worker, or other employee employed in connection with the Work. All payroll records shall be certified as being true and correct by Contractor or Subcontractors keeping such records; and the payroll records shall be available for inspection at all reasonable hours at the principal office of Contractor on the following basis:

- 1. A certified copy of an employee's payroll record shall be made available for inspection or furnished to such employee or the employee's authorized representative on request.
- 2. A certified copy of all payroll records shall be made available for inspection upon request to Owner, the State of California Division of Labor Standards Enforcement, and the Division of Apprenticeship Standards of the State of California Division of Industrial Relations.
- 3. A certified copy of all payroll records shall be made available upon request by the public for inspection or copies thereof made; provided, however, that the request by the public shall be made to either Owner, the Division of Apprenticeship Standards, or the Division of Labor Standards Enforcement. The public shall not be given access to such records at the principal offices of Contractor or Subcontractors. Any copy of the records made available for inspection as copies and furnished upon request to the public or any public agency by Owner shall be marked or obliterated in such a manner as to prevent disclosure of an individual's name, address, and social

security number. The name and address of Contractor awarded the Contract or performing the Contract shall not be marked or obliterated.

16.4.3 Contractor shall file a certified copy of the payroll records with the entity that requested the records within 10 days after receipt of a written request. Contractor shall inform Owner of the location of such payroll records for the Project, including the street address, city, and county; and Contractor shall, within 5 working days, provide notice of change of location of such records. In the event of noncompliance with the requirements of this Article 16.4 or with the State of California Labor Code Section 1776, Contractor shall have 10 days in which to comply following receipt of notice specifying in what respects Contractor must comply. Should noncompliance still be evident after the 10 day period, Contractor shall forfeit to Owner, as a penalty, \$25 for each day, or portion thereof, for each worker, until strict compliance is accomplished. Such forfeiture amounts may be deducted from the Contract Sum.

16.5 APPRENTICES

16.5.1 For purposes of this Article 16.5, the term Subcontractor shall not include suppliers, manufacturers, and distributors.

16.5.2 Only apprentices, as defined in the State of California Labor Code Section 3077, who are in training under apprenticeship standards and written apprentice agreements under Chapter 4, Division 3, of the State of California Labor Code, are eligible to be employed by Contractor and Subcontractors as apprentices. The employment and training of each apprentice shall be in accordance with the provisions of the apprenticeship standards and written apprentice agreements under which the apprentice is training.

16.5.3 Every apprentice shall be paid the standard wage to apprentices, under the regulations of the craft or trade at which the apprentice is employed, and shall be employed only at the Work in the craft or trade to which the apprentice is indentured.

16.5.4 When Contractor or Subcontractors employ workers in any apprenticeship craft or trade on the Work, Contractor or Subcontractors shall 1) send contract award information to the applicable joint apprenticeship committee that can supply apprentices to the site of the public work and 2) apply to the joint apprenticeship committee, which administers the apprenticeship standards of the craft or trade in the area of the Project site, for a certificate approving Contractor or Subcontractors under the apprenticeship standards for the employment and training of apprentices in the area of the Project site. The committee will issue a certificate fixing the number of apprentices or the ratio of apprentices to journeypersons who shall be employed in the craft or trade on the Work. The ratio will not exceed that stipulated in the apprenticeship standards under which the joint apprenticeship committee operates; but in no case shall the ratio be less than 1 hour of apprentice work for every 5 hours of journeyperson work, except as permitted by law. Contractor or Subcontractors shall, upon the issuance of the approval certificate in each such craft or trade, employ the number of apprentices or the ratio of apprentices to journeypersons fixed in the certificate issued by the joint apprenticeship committee or present an exemption certificate issued by the Division of Apprenticeship Standards.

North County Admin Roof Upgrade Issue for Bid – **Dec 2023**

- **16.5.5** "Apprenticeship craft or trade," as used in this Article 16.5, shall mean a craft or trade determined as an apprenticeship occupation in accordance with rules and regulations prescribed by the Apprenticeship Council.
- 16.5.6 If Contractor or Subcontractors employ journey-workers or apprentices in any apprenticeship craft or trade in the area of the Project site, and there exists a fund for assisting to allay the cost of the apprenticeship program in the trade or craft, to which fund or funds other Contractors in the area of the Project site are contributing, Contractor and Subcontractors shall contribute to the fund or funds in each craft or trade in which they employ journey-workers or apprentices on the Work in the same amount or upon the same basis and in the same manner done by the other contractors. Contractor may include the amount of such contributions in computing its bid for the Contract; but if Contractor fails to do so, it shall not be entitled to any additional compensation therefor from Owner.
- **16.5.7** In the event Contractor willfully fails to comply with this Article 16.5, it will be considered in violation of the requirements of the Contract.
- **16.5.8** Nothing contained herein shall be considered or interpreted as prohibiting or preventing the hiring by Contractor or Subcontractors of journey-worker trainees who may receive on-the-job training to enable them to achieve journey-worker status in any craft or trade under standards other than those set forth for apprentices.

16.6 WORK DAY

16.6.1 Contractor shall not permit any worker to labor more than 8 hours during any 1 day or more than 40 hours during any 1 calendar week, except as permitted by law and in such cases only upon such conditions as are provided by law. Contractor shall forfeit to Owner, as a penalty, \$25 for each worker employed in the execution of this Contract by Contractor, or any Subcontractor, for each day during which such worker is required or permitted to work more than 8 hours in any 1 day and 40 hours in any 1 calendar week in violation of the terms of this Article 16.6 or in violation of the provisions of any law of the State of California. Such forfeiture amounts may be deducted from the Contract Sum. Contractor and each Subcontractor shall keep, or cause to be kept, an accurate record showing the actual hours worked each day and each calendar week by each worker employed on the Project, which record shall be kept open at all reasonable hours to the inspection of Owner, its officers and agents, and to the inspection of the appropriate enforcement agency of the State of California.

END OF SECTION

SECTION 00 73 00 SUPPLEMENTARY CONDITIONS

1. MODIFICATION OF GENERAL CONDITIONS, ARTICLE 11 - INSURANCE AND BONDS

A. Contractor's Liability Insurance

The Contractor shall procure, maintain, and keep in force at all times during the term of the Contract, at the Contractor's sole expense, the following insurance:

General Liability Insurance including, but not limited to, protection for claims of bodily injury and property damage liability, personal and advertising injury liability, and products and completed operations liability. Coverage shall be at least as broad as "Insurance Services Office Commercial General Liability Coverage Form CG 0001" (occurrence). The limits of liability shall be not less than:

Each Occurrence One Million Dollars (\$1,000,000) General Aggregate Two Million Dollars (\$2,000,000)

The policy shall cover contractual liability applicable to the Contractor's assumed liability under this Contract. The policy shall provide coverage for claims arising out of subsidence. The Products and Completed Operations coverage shall be maintained for at least two years after completion of the Contract.

<u>Automobile Liability Insurance</u> providing protection against claims of bodily injury and property damage arising out of ownership, operation, maintenance, or use of owned, hired, and non-owned automobiles. Coverage shall be at least as broad as "Insurance Services Office Business Auto Coverage Form CA 0001," symbol 1 (any auto). The limits of liability shall not be less than:

Bodily Injury and Property Damage

Combined Single Limit One Million Dollars (\$1,000,000)

The Workers' Compensation policy required hereunder shall be endorsed to state that the Workers' Compensation carrier waives its right of subrogation against the County, its officers, officials, employees, agents or volunteers.

In the event the Contractor is self-insured, the Contractor shall furnish a Certificate of Permission to Self-Insure by the Department of Industrial Relations Administration of Self-Insurance, Sacramento. No Workers' Compensation insurance is required for contractors that do not have employees.

Excess or Umbrella Liability Insurance. If the Special Provisions require limits of general liability insurance of more than one million dollars (\$1,000,000) per occurrence, the Contractor shall carry excess or umbrella liability insurance providing excess coverage at least as broad as the underlying coverage for general, automobile and employer's liability with a limit equal to the amount stated in the Special Provisions per occurrence and aggregate.

<u>Contractor's Equipment Insurance</u>. The Contractor, and each of its Subcontractors, shall separately insure its own equipment for loss and damage. The Contractor's Property and Inland Marine policies shall include, or be endorsed to include, a waiver of subrogation against the County, its officers, officials, employees, agents, and volunteers which might arise by reason of damage to the Contractor's property or equipment (owned, leased or borrowed) in connection with work performed under this Contract by the Contractor.

Other Provisions

- 1. The Contractor's General Liability, Automobile Liability, and any Excess or Umbrella Liability, shall contain the following provisions:
 - a. The County, its officers, officials, employees, agents, and volunteers shall be covered as additional insureds as respects liability arising out of the activities performed by or on behalf of the Contractor, products and completed operations of the Contractor, premises owned, occupied, or used by the Contractor, or automobiles owned, leased, hired, or borrowed by the Contractor. The policy shall contain no special limitations on the scope of coverage afforded to the County, its officers, officials, employees, agents, or volunteers.
 - b. For any claims related to this Contract, the Contractor's insurance coverage shall be primary insurance as respects the County, its officers, officials, employees, agents, or volunteers. Any insurance or self-insurance maintained by the County, its officers, officials, employees, agents, or volunteers shall be not be excess of the Contractor's insurance and shall not contribute with it.
 - c. Any failure to comply with reporting or other provisions of the policies on the part of the Contractor, including breaches of warranties, shall not affect coverage provided to the County, its officers, officials, employees, agents, or volunteers.
- 2. The Contractor's General Liability and any Excess or Umbrella Liability insurance policies shall contain an endorsement stating that any aggregate limits shall apply separately to the work.
- 3. The Contractor's insurance shall apply separately to each insured against whom claim is made or suit is brought, except with respect to the limits of the insurer's liability.
- 4. Each insurance policy shall state that coverage shall not be suspended, voided, cancelled by the Contractor or the County, reduced in scope of coverage or in limits, non-renewed, or materially changed unless the insurer(s) provide thirty (30) days written notice by certified mail to the County prior to such change. Ten (10) days prior written notice by certified mail shall be given to the County in the event of cancellation due to nonpayment of premium.
- 5. All of the Contractor's insurance coverage, except as noted below, shall be placed with insurance companies with a current A.M. Best rating of at least A-:VII. Exceptions:
 - a. Underwriters at Lloyd's of London, which are not rated by A.M. Best.
 - b. Workers' Compensation which is provided through a State Compensation Insurance Fund or a qualified self-insurer for Workers' Compensation under California law.
 - c. For liability insurance required under Environmental Liability insurance, insurance requirements shall be placed with insurance companies with a current A.M. Best rating of at least B+:VII.
- 6. The Contractor shall sign and file with the County the following certification prior to commencing performance of the work of the Contract:

"I am aware of the provisions of Section 3700 of the Labor Code which require every employer to be insured against liability for worker's compensation or to undertake self-insurance in accordance with the provisions of the Code, and I will comply with such provisions before commencing the performance of the work of this Contract."

Said certification is included in the Contract and the signing and returning the Contract shall constitute signing and filing of the said certification.

- 7. The County, at its discretion, may require new types of insurance coverage or increase the limits of insurance coverage required hereunder at any time during the term of the Contract by giving thirty (30) days written notice to the Contractor. Contractor shall immediately procure such insurance or increase the limits of coverage and provide certificates of insurance, including copies of all required endorsements, to the County within thirty (30) days of receipt of the County's request.
- 8. The required insurance coverage shall be subject to the approval of the County, but any acceptance of insurance certificates by the County shall in no way limit or relieve the Contractor of its duties and responsibilities in this Contract.
- 9. If the Contractor fails to procure or maintain insurance as required by this Chapter and any Special Provisions, or fails to furnish the County with proof of such insurance, the County, at its discretion, may procure any or all such insurance. Premiums for such insurance procured by the County shall be deducted and retained from any sums due the Contractor under the Contract. Failure of the County to obtain such insurance shall in no way relieve the Contractor from any of the Contractor's responsibilities under the Contract. Any failure of the Contractor to maintain any item of the required insurance is sufficient cause for termination of the Contract.
- 10. The making of progress payments to the Contractor shall not be construed as relieving the Contractor of responsibility for loss or damage, or destruction occurring prior to final acceptance by the County.
- 11. The County is authorized to execute amendments and waivers, with or without conditions, to the insurance requirements of the Contract. The County will provide such amendments or waivers in writing to the Contractor.

The failure of the County to enforce in a timely manner any of the provisions of this Section shall not act as a waiver to enforcement of any of these provisions at any time during the term of the Contract.

Notification of Accident or Occurrence

The Contractor shall report by telephone to the County within twenty-four (24) hours and also report in writing to the County within fifteen (15) days after the Contractor or any subcontractors or agents have knowledge of any accident or occurrence involving death of or injury to any person or persons, or damage in excess of ten thousand dollars (\$10,000) to the work, property of the County or others, arising out of any work done by or on behalf of the Contractor as part of the Contract. Such report shall contain:

- 1. The date and time of the occurrence.
- 2. The names and addresses of all persons involved, and a description of the accident or occurrence and the nature and extent of injury or damage.

II. Small Construction Projects.

For projects less than \$125,000, the County may, at their discretion, decrease the required limits of coverage for General Liability and/or Automobile Liability insurance appropriately with regard to the lesser risk involved. Unless stated otherwise Contractor's Equipment Insurance is not required

B. Professional Liability Insurance

If Contractor is providing design services beyond preparation of shop drawings for this Project, then Contractor shall carry professional liability insurance coverage with limits of [\$1,000,000] per claim and annual aggregate. The services provided for the Project will not be excluded from coverage under such policy and the retroactive date on such policy will be prior to the first date upon which Contractor provided services for this Project. If, prior to four years following Final Completion, the retroactive date changes in subsequent policy years or if the policy is replaced, CM/GC will furnish evidence that it has purchased the supplemental "Extended Reporting Period Endorsement" under the previous policy extending for at least three years the period during which a claim may first be made.

C. Bid, Performance and Payment Bonds, Paragraph 11.4

Contractor shall furnish bonds covering the well and truly to be made (Bid Bond), faithful performance of the Contract (Performance Bond) and payment of obligations arising thereunder (Payment Bond) on the forms contained in Exhibits 1 and 2.

The Bid Bond shall be in the amount of ten (10) percent of the total amount of the bid.

The Payment Bond and Performance Bond shall each be in the amount of the Contract Sum.

The Payment Bond and Performance Bond shall be in effect on the date the Contract is signed by Owner.

Contractor shall promptly furnish such additional security as may be required by Owner to protect its interests and those interests of persons or firms supplying labor or materials to the Work. Contractor shall furnish supplemental Payment and Performance Bonds each in the amount of the current Contract Sum at the request of Owner.

Surety companies used by Contractor shall be, on the date the Contract is signed by Owner, an admitted surety insurer.

The premiums for the Payment Bond and Performance Bond shall be paid by Contractor.

2. CODE OF CONDUCT

- A. Any deliveries to the project site by common carrier must be marked to the Contractor only with the official project title clearly identified. A representative of the contractor must be onsite and equipped to receive the delivery. Owner will not receive any shipment on behalf of the Contractor.
- **B.** You are reminded to comply with CALOSHA requirements for job safety. Designate a site safety representative for the work place as required by OSHA and conduct a regular program of safety.

- C. You will be responsible for your employees conduct while on the job site, i.e. whistling, profanity, garbage, dress code, etc. You are required to inform your employees working at the construction site that Owner is strongly opposed to sexual harassment and that such behavior is prohibited both by law. It is the intention of the Owner to take whatever action may be needed to prevent, correct, and, if necessary, discipline behavior which violates this policy.
- **D.** For the health and safety of staff, visitors, volunteers, and inmates, smoking is not permitted inside any facility. Smoking is also not permitted within 20 feet of any entrance at any building entrance.
- **E.** You are required to establish a policy of non-discriminatory practice in all personnel actions.
- **F.** Identification badges are required inside and/or next to the existing jail.
- G. Owner strives to maintain worksites free from the illegal use, possession, or distribution of alcohol or of controlled substances, as defined in schedules I through V of the Controlled Substances Act, 21 United States Code section 812, and by regulation at 21 Code of Federal Regulations section 1308. Unlawful manufacture, distribution, dispensing, possession, use, or sale of alcohol or of controlled substances on Owner property is prohibited.
- **H.** Owner prohibits and has zero tolerance for workplace violence. Any disruptive behavior, act of intimidation, threat of violence or act of violence committed against any person and or property is prohibited.

3.

- **A.** Standard of Care. Contractor will perform its services using that skill and care used by other competent licensed design professionals skilled in designing scopes of Work similar in size, scope, and complexity in jail facilities in California.
- **B.** Notwithstanding anything to the contrary in the Contract Documents, Contractor shall not be entitled to any adjustment in the Contract Sum or Contract Time due to errors, omissions or inconsistencies in the Contract Documents that are prepared by Contractor or are reviewed by Contractor in its capacity as a licensed design professional.
- **C.** Ownership and Use of Documents
 - (i) Definition. Provided Owner is in compliance with all payment provisions under the Agreement (except as may be subject to a good-faith dispute), Owner shall own all title and copyright privileges to all drawings, plans, specifications, calculations, physical or electronic models, other electronic data, and other documents, materials or information prepared by Contractor or its subcontractors or suppliers in connection with the Project (and any portable electronic media upon which they were prepared or stored) (the "Project Documents"), subject only to the use provisions set forth below.
 - (ii) Possession. The originals of all Project Documents shall be held by Contractor or its subcontractors for the benefit of Owner. At Owner's request and without charge,

North County Admin Roof Upgrade Issue for Bid – **Dec 2023**

- copies of any or all Project Documents shall be immediately delivered to Owner in their original form, or in clear, reproducible form, regardless of whether the Project is completed, suspended or terminated.
- (iii) Owner's Use. Owner shall have the right, regardless of whether the Project is completed, suspended or terminated, in whole or in part, to use and reuse the Project Documents for any purpose and without royalty or charge. If Owner uses or reuses the Project Documents to have work performed on its behalf for which Contractor is not retained, Owner shall indemnify, defend, and hold Contractor free and harmless from any resulting liability to third parties for personal injury, death, or property damage.
- (iv) Limited Use by Constructors. Contractor and its subcontractors and suppliers are granted a limited license to use and reproduce applicable portions of the Project Documents appropriate to, and for use in, the execution of their Work under the Contract Documents. The drawings, specifications and other Project Documents furnished to Contractor or any subcontractor or supplier are for use solely with respect to the Project.

END OF SECTION

Attachment 1



































